School of Molecular and Life Sciences

Sustainable Latent Fingermark Detection Protocols for Remote-Location and Resource-Limited Jurisdictions

Jemmy Terence Bouzin 0000-0001-6027-2997

This thesis is presented for the Degree of Doctor of Philosophy of Curtin University

October 2023

Declaration

To the best of my knowledge and belief, this thesis contains no material previously published by any other person except where due acknowledgement has been made.

This thesis contains no material which has been accepted for any other degree or diploma in any university.

Human Ethics

The research presented and reported in this thesis was conducted in accordance with the National Health and Medical Research Council National Statement on Ethical Conduct in Human Research (2007) – updated March 2018. The proposed study received human research ethics approval from the Curtin University Human Research Ethics Committee (EC00262), Approval Number HRE2020-0162.

Signature:

Date: 18/10/2023

Acknowledgement of Country

We acknowledge that Curtin University works across hundreds of traditional lands and custodial groups in Australia, and with First Nations people around the globe. We wish to pay our deepest respects to their ancestors and members of their communities, past, present, and to their emerging leaders. Our passion and commitment to work with all Australians and peoples from across the world, including our First Nations peoples are at the core of the work we do, reflective of our institutions' values and commitment to our role as leaders in the Reconciliation space in Australia.

Abstract

Over time, the growth of forensic science has become a worldwide practice that aims to promote justice, peace, and prosperity. In particular, latent fingermark detection plays an important role in resource-limited Global South jurisdictions where advanced techniques like DNA profiling are not readily available. Nonetheless, most research and development in fingermark detection has been carried out in the Global North, leading to a lack of methods specifically tailored to the challenges of the Global South. This thesis introduces the concept of frugal forensics, which presents the first systematic framework for sustainable forensic science provision, with a focus on latent fingermark detection in resource-limited jurisdictions. Implementing sustainable methods is key to addressing the gap in forensic science services prevalent in the Global South and serves as a catalyst for international efforts to reduce inequalities between regions, in line with the United Nations Sustainable Development Goals.

The practical implementation of frugal forensics was demonstrated through its application to three different latent fingermark techniques. As a result, two sustainable formulations tailored for the Seychelles Police Force (SPF) were developed, namely SPF 1,2-indanedione zinc (IND/Zn) and Wet SPF powder suspension. The assessments were conducted following the International Fingerprint Research Group's guidelines, which provide a framework for evaluating new or revised fingermark detection methods. It also ensures compliance with quality assurance requirements and supports the implementation of detection techniques in operational laboratories.

A cost-effective and sustainable version of the IND/Zn formulation was developed by modifying existing formulations used by police in Australia National Centre for Forensic Studies (NCFS), Germany Bundeskriminalamt (BKA), and the UK Home Office Centre for Applied Science and Technology (CAST). The main challenge was to find a cost-effective and supply chain-friendly alternative to the expensive solvent HFE- 7100 used in the IND/Zn formulation. Based on a frugal analysis, the BKA petroleum ether-based formulation was identified as the most cost-effective and low-risk supply chain option. Solstice[®] Performance Fluid (PF) as an alternative carrier solvent was also evaluated. Although it was found to have potential as a substitute, some aspects of Solstice PF fall short as a carrier solvent for a sustainable reagent. Spectroscopy studies revealed that reducing excess 1,2-indanedione in the formulation enhances reagent effectiveness. The SPF formulation was modified to contain 594 mg/L of 1,2-indanedione, reducing the cost of the formulation by 40%. This formulation was assessed through laboratory and incidental fingermark trials across five substrates, two ageing periods of one and 4 weeks, and a range of ten donors producing 1200 fingermark halves and 100 incidental items. The findings demonstrated comparable performance between SPF and NCFS formulations. Both formulations exhibited high sensitivity and effectively developed fingermarks that were 1 week and 4 weeks old.

The recently published Wet UCIO (Unitat Central d'Inspeccions Oculars) carbonbased powder suspension was modified by replacing the surfactant solution with an in-house sodium dodecyl sulfate (SDS) surfactant solution to develop the Wet SPF powder suspension. This formulation containing 15% SDS in a 5% ethanol/water mixture was compared to the commercial Wetwop[™] black powder suspension on eight different tapes with two aging periods using ten donors. The Wet SPF performed similarly to the commercial Wetwop[™] with slightly better sensitivity. Using an inexpensive and easily accessible SDS salt allows for better transportation, storage, and quality control compared to commercial surfactant solutions.

Acid Nile blue as an emerging lipid-sensitive technique with promising frugal forensic attributes was also investigated. Like Oil red O, acid Nile blue was found to successfully develop charged fingermarks on wet porous substrates, with its effectiveness decreasing as the fingermark ages. It was also observed that Nile blue targets the 'fragile fraction' of the sebaceous component and should be applied to fingermarks less than 4 weeks old. While acid Nile blue has limited sensitivity, its affordability, water-based nature, and low toxicity make it a useful tool for training and outreach programs. In addition, acid Nile blue was compared with Basic Yellow 40 (BY40) as a post-cyanoacrylate stain. The study revealed that acid Nile blue was an effective method for staining preferentially lipid-rich fingermarks, unlike BY40, which produced no preferential development. The result suggests that acid Nile blue has a different staining mechanism than BY40, penetrating through the polymer structure to interact with the fingermark residue. These findings can aid in developing improved post-cyanoacrylate stains by providing insight into the post-cyanoacrylate stains by providing insight into the post-cyanoacrylate stains.

Finally, a comprehensive and objective assessment tool was developed to evaluate the sustainability of methods in the frugal forensics framework. Drawing from the fingermark grading scale, this tool objectively evaluates all six attributes of frugal forensics, promoting transparent method selection and adhering to quality assurance requirements. The tool was successfully applied to evaluate the sustainability and implementation recommendations of the three latent fingermark detection methods. Moreover, it can also serve as a valuable planning tool by identifying areas for improvement.

Acknowledgements

First and foremost, I would like to express my utmost gratitude to God, the Almighty, for bestowing upon me numerous blessings, knowledge, wisdom, and opportunities, eventually leading to the completion of this thesis.

The accomplishment of this feat would not have been possible without the support and encouragement of a multitude of people around me, and I would like to take this opportunity to acknowledge and thank them.

In particular, I would like to express my heartfelt gratitude and thanks to my supervisory team for their invaluable guidance and motivation without which this thesis would not have been a reality. To my primary supervisor, Professor Simon Lewis, for his warm reception and support throughout my research journey. Your tutelage during my undergraduate studies shaped my interest in forensic science, but I could not have foreseen that 12 years later I would be pursuing a PhD under your supervision, to challenge a global dominant narrative in this field - It has been an absolute privilege. I would also like to express my profound gratitude to my brilliant co-supervisor, Dr Georgina Sauzier, whose quick-witted and priceless guidance has been instrumental in completing this thesis.

I would like to extend my thanks and appreciation to Dr Steve Bleay for his collaboration on this project, Dr Thomas Becker for his assistance with the CRM studies, and Mrs. Angel Lebon for guiding and providing me with the opportunity to commence this research project.

I am also grateful to the members of the Forensic Research Group and extend special thanks to Dr Amanda Frick and Aaron Horrocks for their collaboration on the 1,2indanedone study and Emeritus Professor Wilhelm van Bronswijk for his advice and words of wisdom. The camaraderie and support of the friendly and welcoming group have been essential to my experience throughout my PhD. To the amazing technical staff at Curtin University; A big thank you to Peter Chapman for providing exceptional technical support, to Tomoko for ensuring that everything is in place for the teaching labs, to Marcus and Grant for always being willing to assist me.

I would like to extend my thanks to all fingermark donors, for taking the time to provide samples. To my friends Jean-Paul, Lester, Archicandra, Raymond, Sohib for your support, endless chat and a game of soccer or pool to take my mind off research.

I would like to acknowledge the invaluable support of the Seychelles Police Force (SPF) and the Agency for National Human Resources Development (ANHRD) during my Research Program and the Seychelles Government for awarding me the Post-Graduate Research Scholarship that has funded this research.

Last but not least, my heartfelt gratitude to the special people in my life, my dear dad †Maxime Poris and sister Veronica, my beloved wife Ronita and my precious sons Teron and Liam, whose unwavering love, patience, and support have been a source of strength and motivation throughout this journey.

Contribution to Others

The following have contributed to the thesis, with contributions being listed in the form of CRediT (Contributor Roles Taxonomy) statements for each chapter – see https://www.elsevier.com/authors/policies-and-guidelines/credit-author-statement

Chapters 1 – 8:

Professor Simon W. Lewis (Principal Supervisor): Project Administration, Supervision, Conceptualisation, Writing – Review & Editing.

Dr Georgina Sauzier: Supervision, Conceptualisation, Writing – review & Editing.

Chapter 1:

Portions of this chapter have been published in *Forensic Science International: Synergy and Forensic Chemistry* see Appendix E for details.

Aaron J. Horrocks, Stephen M. Bleay: Conceptualization, Investigation, Writing – review & editing.

Thais Lópes, Anna L. Heavey, Jessie Parrish, Georgina Sauzier, and Simon W. Lewis: Conceptualization, Investigation, Writing – original draft, Writing – review & editing.

Chapter 3:

Portions of this chapter have been published in *Forensic Chemistry and Forensic Science International* see Appendix E for details.

Aaron J. Horrocks, Amanda A. Frick: Conceptualization, Investigation, Writing – review & editing.

Georgina Sauzier, Stephen M. Bleay, Simon W. Lewis: Conceptualization, Writing – review & editing.

Chapter 4:

Portions of this chapter have been published in *Forensic Science International* see Appendix E for details.

Aaron J. Horrocks, Jason Merendino, Stephen M. Bleay: Conceptualization, Investigation, Writing – review & editing. Georgina Sauzier, Simon W. Lewis: Conceptualization, Supervision, Writing – review & editing.

Chapter 7:

Portions of this chapter have been published in *Forensic Science International: Synergy* see Appendix E for details.

Thais Lópes, Anna L. Heavey, Jessie Parrish, Georgina Sauzier, and Simon W. Lewis: Conceptualization, Investigation, Writing – original draft, Writing – review & editing.

Chapter 8:

Portions of this chapter have been published in *Forensic Science International: Synergy and Forensic Chemistry* see Appendix E for details.

Aaron J. Horrocks, Stephen M. Bleay: Conceptualization, Investigation, Writing – review & editing.

Thais Lópes, Anna L. Heavey, Jessie Parrish, Georgina Sauzier, and Simon W. Lewis: Conceptualization, Investigation, Writing – original draft, Writing – review & editing.

Publications

This dissertation contains work which has been submitted for publication in the following peer-reviewed journals:

Jemmy T. Bouzin, Jason Merendino, Stephen M. Bleay, Georgina Sauzier, and Simon W. Lewis. New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc. *Forensic Science International: Reports* 2020 2 100145.

DOI: https://doi.org/10.1016/j.fsir.2020.100145

Jemmy T. Bouzin, Georgina Sauzier, and Simon W. Lewis. Forensic science in Seychelles: An example of a micro-jurisdiction forensic delivery system. *Forensic Science International: Synergy* 2021 3 100139. DOI: <u>https://doi.org/10.1016/j.fsisyn.2021.100139</u>

Jemmy T. Bouzin, Aaron J. Horrocks, Georgina Sauzier, Stephen M. Bleay, and Simon W. Lewis. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry* 2022 30 100439. DOI: https://doi.org/10.1016/j.forc.2022.100439

Jemmy T. Bouzin, Amanda A. Frick, Georgina Sauzier, and Simon W. Lewis. Preliminary evaluation of Solstice[®] PF as a replacement carrier solvent for Australian fingermark detection. *Forensic Science International* 2022 340 111465. DOI: https://doi.org/10.1016/j.forsciint.2022.111465

Jemmy T. Bouzin, Thais Lópes, Anna L. Heavey, Jessie Parrish, Georgina Sauzier, and Simon W. Lewis. Mind the gap: The challenges of sustainable forensic science service provision. *Forensic Science International Synergy* 2023 6 100318. DOI: <u>https://doi.org/10.1016/j.fsisyn.2023.100318</u> In addition, the following peer reviewed article was published during the course of this dissertation:

Kenny J. Travouillon, Christine Cooper, Jemmy T. Bouzin, Linette S. Umbrello, and
Simon W. Lewis. All-a-glow: spectral characteristics confirm widespread
fluorescence for mammals. *Royal Society Open Science 2023 10 (10) 230325*.
DOI: <u>https://doi.org/10.1098/rsos.230325</u>

Conference Presentations

Selected aspects of the work contained within this thesis have been presented, or have been accepted for presentation, at the following conferences:

Oral Presentations

Preliminary evaluation of Solstice[®] PF as a replacement carrier solvent for Australian fingermark detection, International Association of Forensic Sciences - 23rd Triennial Meeting, Sydney, September 2023 (Amanda A. Frick).

Frugal Forensics, International Fingerprint Research Group, Delft, Amsterdam, June 2023 (Simon W. Lewis).

Forensics in Paradise, Science Teachers Association of Western Australia - Future Science Conference, Perth, December 2022.

1,2-indanedione reagent: Investigation into operational use in a small, remote jurisdiction, Australian New Zealand Forensic Science Society - 25th International Symposium, Brisbane, September 2022.

New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc, Virtual 10th Annual Forensic Science Symposium - hosted by Florida International University, (online), June 2021.

New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc, Curtin University School of Molecular and Life Sciences Symposium, Perth, December 2020.

Invited Presentations

Panellist for Round Table 1 - Challenges for the future, Forensic Document Examiners Live International Knowledge Exchange on Documents meeting, (online), March 2023. Forensic science in Seychelles: A micro-jurisdiction forensic delivery system, PathWest - Forensic Biology Seminar Series, December 2021.

Improving latent fingermarks protocols, Inaugural Western Australian Forensic Science Forum, Perth, October 2021.

Forensic science in Seychelles: A micro-jurisdiction forensic delivery system, Guest Lecture to First Year Forensic Students, Curtin University, Perth 2020-23.

A unique perspective of a small-Island-State forensic delivery system, Australian New Zealand Forensic Science Society, Western Australia Branch meeting, Perth, August 2019.

Poster Presentations

Developing sustainable formulations enhances global access to latent fingermark protocols, International Association of Forensic Sciences - 23rd Triennial Meeting, Sydney, September 2023 (Simon W. Lewis).

1st Place Poster Prize - Fingerprint Evidence & Biometrics

Detection of historic fingermarks on paper documents using 1,2-indanedione/zinc, Australian New Zealand Forensic Science Society - 25th International Symposium, Brisbane, September 2022.

Forensic Science in Seychelles: A unique perspective of a small-island-state forensic delivery system, Australian New Zealand Forensic Science Society - 25th International Symposium, Brisbane, September 2022.

New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc, Royal Society of Chemistry Twitter Conference, (online), March 2021.

> 2nd Place Poster Prize

Luminescence detection of latent fingermarks on paper surfaces after incidental handling, Royal Society of Chemistry Twitter Conference, (online), March 2020

Investigation into improved protocols for latent fingermark detection technique for jurisdictions with limited resources and tropical climates, Curtin University School of Molecular and Life Sciences Symposium, Perth, December 2019.

Jemmy T Bouzin, presenting author of the above-listed orals and posters, except were stated otherwise.

Table of Contents

1.4	1.4.1.5 1,2-indaneutone	
	.1.4 Lipid-sensitive reagents	19
:	1.4.1.5 Physical developer (PD)	20
:	1.4.1.6 Oil red O (ORO)	20
:	1.4.1.7 Nile red	22
:	1.4.1.8 Nile blue	22
1.4.2	Non-porous-surface fingermark detection methods	24
1.4	.2.1 Fingerprint powder	24
1.4	.2.2 Powder suspension	25
1.4	.2.3 Cyanoacrylate	27
1.4	.2.4 Post-cyanoacrylate stains	29
1.5 Lat	ent fingermark detection service provision	
1.5.1	Seychelles latent fingermark detection service provision	
1.5.2	Challenges in operationalising latent fingermark detection protocols	31
1.5.3	Sustainable forensic science provision; frugal forensics	
16 Ain	and Objectives	37
210 7 111		
Chapter 2.	Sample collection, experimental methods, and instrumentation	40
2.1 Intr	roduction	41
2.2 Evn	parimental considerations	
2.2 EXP		11
2 2 1	IEPG guidelines	41
2.2.1	IFRG guidelines	41 41
2.2.1 2.2.2	IFRG guidelines Frugal forensics approach	41 41 42
2.2.1 2.2.2 2.2.3	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact	41 41 42 42
2.2.1 2.2.2 2.2.3 2.3 Fing	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology	41 41 42 42 42
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information	41 41 42 42 42 42
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures	
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions	41 42 42 42 42 42 42 42 43 45
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions pstrates selection and environmental considerations	41 42 42 42 42 42 42 43 45 45
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions pstrates selection and environmental considerations Substrates surface selection	41 41 42 42 42 42 42 43 45 45 45
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions pstrates selection and environmental considerations Substrates surface selection Environmental considerations	41 41 42 42 42 42 42 43 45 45 45
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Fing	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions pstrates selection and environmental considerations Substrates surface selection Environmental considerations germark development	41 42 42 42 42 42 42 42 45 45 45 45 45
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Fing 2.5.1	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions pstrates selection and environmental considerations Substrates surface selection Environmental considerations germark development 1,2-indanedione zinc	41 42 42 42 42 42 42 42 45 45 45 45 45 45 45 46
2.2.1 2.2.2 2.2.3 2.3 Finy 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Finy 2.5.1 2.5.2	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions postrates selection and environmental considerations Substrates surface selection Environmental considerations germark development 1,2-indanedione zinc Powder suspension	41 42 42 42 42 42 42 42 43 45 45 45 45 45 46 46 48
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Fing 2.5.1 2.5.2 2.5.3	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions patrates selection and environmental considerations Substrates surface selection Environmental considerations germark development 1,2-indanedione zinc Powder suspension Oil red O	41 42 42 42 42 42 42 42 42 43 45 45 45 45 45 46 46 48 48
2.2.1 2.2.2 2.2.3 2.3 Fing 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Fing 2.5.1 2.5.2 2.5.3 2.5.4	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions Substrates selection and environmental considerations Substrates surface selection Environmental considerations germark development 1,2-indanedione zinc Powder suspension Oil red O Cyanoacrylate fuming	41 42 42 42 42 42 42 42 42 42 43 45 45 45 45 46 46 48 48 49 49 49
2.2.1 2.2.2 2.2.3 2.3 Finy 2.3.1 2.3.2 2.3.3 2.4 Sub 2.4.1 2.4.2 2.5 Finy 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4 2.5.5	IFRG guidelines Frugal forensics approach COVID-19 pandemic impact germark methodology Donor information Fingermark collection procedures Fingermark ageing and storage conditions Substrates selection and environmental considerations Substrates surface selection Environmental considerations germark development 1,2-indanedione zinc Powder suspension Oil red O Cyanoacrylate fuming Acid Nile Blue	41 42 42 42 42 42 42 42 42 43 45 45 45 45 45 45 46 48 49 49 49 49 50

2	.5.6	Basic Yellow 40	51
2.6	Instr	umentation	51
2	.6.1	Photography of developed fingermarks	
2	.6.2	Fluorescence spectroscopy	
2	.6.3	UV-vis spectroscopy	54
2	.6.4	ATR-FTIR spectroscopy	55
2	.6.5	Confocal Raman microscopy	55
2.7	Fing	ermark grading	56
20	Ctat		F7
2.8	Stati	stical analysis	57
2.9	Hum	an ethics approval	59
Chapte	er 3.	A sustainable 1,2-indanedione zinc formulation	60
2.4		· ··	64
3.1	Intro	duction	61
3.2	Ехре	rimental	70
3	.2.1	Preliminary consideration	70
3	.2.2	Experimental methods	70
	3.2.2	.1 Substrate	70
	3.2.2	.2 Fingermark collection	70
	3.2.2		71
_	3.2.2	Photography and grading of developed fingermarks	
3	.2.3	Head-to-head comparative assessment of existing IND/Zn formulations	
	3.2.3	1.1 Investigation into co-solvents	
	3.2.3	1.2 Investigation into 1,2-indanedione concentration	
3	.2.4	Investigation into Carrier solvent – Solstice® PF as an alternative	73
	3.2.4	.1 Overall performance of Solstice [®] PF	
	3.2.4	2 Effect of co-solvents on ink diffusion	73
	2.2.4	A Percent stability	75
3	2.5	Statistical analysis	75
J	.2.5		
3.3	Resu	Its and discussion	76
3	.3.1	Head-to-head comparative assessment of existing IND/Zn formulations	76
	3.3.1	.1 Overall performance and sensitivity	76
	3.3.1		77
	3.3.1		80
_	3.3.1	4 Frugal forensics analysis of existing operational IND/Zn protocols	80
3	.3.2	Carrier solvent – Petroleum ether	

	3.3.2	2.1	Co-solvents	81
	3.3.2	2.2	1,2-indanedione concentration	83
3	.3.3	Carr	ier solvent - Solstice [®] PF as an alternative	87
	3.3.3	3.1	Overall performance of Solstice [®] PF	87
	3.3.3	3.2	Compatibility with inks	90
	3.3.3	3.3	Effect of co-solvents on ink diffusion	92
	3.3.3	3.4	Reagent stability	94
	3.3.3	8.5	Frugal forensics analysis of Solstice® PF as a carrier solvent for fingermark	
	deve	lopm	ent	95
3.4	Cond	clusior	<i>IS</i>	96
Chapte	er 4.	SPF	1,2-indanedione zinc comparative assessment	99
4.1	Intro	oductio	on	.100
4.2	Ехре	rimen	tal	. 102
4	.2.1	Ехре	rimental consideration	. 102
4	.2.2	Ехре	rimental methods	. 103
	4.2.2	2.1	Substrate	. 103
	4.2.2	2.2	Fingermark collection	. 103
	4.2.2	2.3	Fingermark development	. 103
	4.2.2	2.4	Photography and grading of developed fingermarks	. 103
4	.2.3	Preli	minary investigations	.104
	4.2.3	3.1	Inks compatibility	. 104
	4.2.3	3.2	Processing conditions	. 104
	4.2.3	3.3	Simple comparative assessment	.104
4	.2.4	Labo	pratory trials	. 105
	4.2.4	1.1	Comparative assessment	. 105
	4.2.4	1.2	Incidental fingermarks	. 105
	4.2.4	1.3	Environmental considerations	.106
	4.2.4	1.4	Reagent stability	.106
	4	.2.4.5	UV-visible spectroscopy	. 106
4	.2.5	Stati	stical analysis	. 107
4.3	Resu	ılts an	d discussion	.108
4	.3.1	Preli	minary investigations	.108
	4.3.1	l.1	Inks compatibility	. 108
	4.3.1	L.2	Ambient temperature processing conditions	. 109
	4.3.1	L.3	Simple comparative assessment	. 109
4	.3.2	Labo	pratory trials	. 110

	4.3.2.	1 Comparative assessment	
	4.3.2.	2 Incidental fingermarks	115
	4.3.2.	3 Environmental considerations	
	4.3.2.	4 Reagent stability	
	4.3	3.2.5 UV-vis spectroscopy	
4	.3.3	Statistical Analysis	
4.4	Concl	usions	122
Chapte	r 5.	Modification of WET UCIO powder suspension for fingermark detection on	pressure
sensitiv	/e adhe	sive tape	123
5.1	Introd	duction	
5.2	Exper	imental	
5	.2.1	Experimental methods	
	5.2.1.	1 Powder suspension preparation	
	5.2.1.	2 Substrates	
	5.2.1.	3 ATR-FTIR Tape Analysis	
	5.2.1.	4 Fingermark collection, photography, and grading	
5	.2.2	Stage A – Formulation modification	
5	.2.3	Stage B – Supplementary studies	
5	.2.4	Stage C - Comparative assessment to Wetwop [™]	
5	.2.5	Statistical analysis	
5.3	Resul	ts and discussion	
5	.3.1	Tape Characterisation	
	5.3.1.	1 PSA tape backing	
	5.3.1.	2 PSA tape adhesive	139
5	.3.2	Formulation modification	
5	.3.3	Supplementary studies	
5	.3.4	Comparative assessment to Wetwop [™]	
	5.3.4.	1 Overall performance	
	5.3.4.	2 Sensitivity	
5	.3.5	Statistical analysis	
5.4	Concl	usion	155
Chapte	r 6.	Investigation into acid Nile blue as an emerging lipid-sensitive and post-	
cyanoa	crylate	detection method	157
61	Introd	duction	158

6.2	?	Exper	riment	al	. 161
	6.2	.1	Exper	imental methods	. 161
		6.2.1.	1	Reagent preparation and sample development	. 161
		6.2.1.	2	Substrates	. 161
		6.2.1.	3	Fingermark collection	. 162
		6.2.1.	.4	Sample visualisation and evaluation	. 162
	6.2	.2	As a l	ipid-sensitive reagent on wet porous substrates	. 162
		6.2.2.	1	Preliminary Study	. 162
		6.2.2.	2	Comparison to Oil red O	. 163
	6.2	.3	As a p	oost-cyanoacrylate stain	. 163
		6.2.3.	1	Comparison to Basic Yellow 40	. 163
		6.2.3.	2	Confocal Raman microscopy investigation into post-cyanoacrylate staining	
		mech	anism		. 163
6.3	3	Resul	ts and	discussion	. 164
	6.3	.1	As a l	ipid-sensitive reagent on wet porous substrates	. 164
		6.3.1.	.1	Preliminary investigations	. 164
		6.3.1.	2	Comparison to Oil red O	. 165
	6.3	.2	Suital	bility as a post-cyanoacrylate stain and exploring the staining mechanism	. 169
		6.3.2.	.1	Preliminary investigation	. 169
		6.3.2.	2	Comparison to Basic Yellow 40	. 171
		6.3	3.2.3	Sequencing to enhance detection of overlapping fingermarks.	. 173
		6.3.2.	.4	Confocal Raman microscopy investigation into post-cyanoacrylate staining	
		mech	anism		. 176
6.4	1	Concl	usion.		. 180
Chapt	er	7.	Fruga	I forensics sustainability assessment tool	.181
7.1		Introd	ductio	n	. 182
7 2	,	Fyner	iment		184
7.2		слрег	ment		104
7.3	}	Resul	ts and	Discussion	. 184
7.4	ļ,	Concl	usion.		. 188
		_			
Chapt	er	8.	Concl	usion and future work	.190
8.1		Concl	usion.		. 191
8.2	?	Alterr	native	carrier solvent for amino acid sensitive reagent	. 192
g a	2	Powd	lpr cur	nension and acid Nile blue improvements	102
0.5	, ,	1.0000	er sus	pension and acia whe blac improvements	. 193

	8.4	IFRG	G guidelines requirements	194
	8.5	Susta	tainable forensic science service provision	194
Ch	apter	9.	References	196
Aŗ	ppend	ices		227
	Appe	endix A	A-1: Statistical analysis test assumptions	228
	Appe	endix A	A-2: Statistical analysis - Chapter 3 R codes and results	230
	Appe	endix A	A-3: Statistical analysis - Chapter 4 R codes and results	233
	Арре	endix A	A-4: Statistical analysis - Chapter 5 R codes and results	237
	Арре	endix E	B-1: Air-condition office datalogger temperature and relative humidity records	242
	Appe	endix E	B-2: Non-air-condition Level 5 space datalogger temperature and relative humidity	
	recor	rds		243
	Арре	endix (C-1: 3M Announcement letter (PFAS Discontinuation)	244
	Арре	endix [D-1: Human ethics participant information form (English)	245
	Арре	endix [D-2: Human ethics consent form (English)	247
	Appe	endix L	D-3: Human ethics participant information form (Creole)	248
	Арре	endix [D-4: Human ethics consent form (Creole)	250
	Арре	endix E	E-1: Contribution statement - New light on old fingermarks paper	251
	Арре	endix E	E-2: Contribution statement – Forensic science in Seychelles paper	252
	Appe	endix E	E-3: Contribution statement – IND/Zn Comparative assessment paper	253
	Appe	endix E	E-4: Contribution statement – Solstice® PF paper	254
	Appe	ndix E	E-5: Contribution statement – Mind the gap paper	255

List of Tables

Table 1.1 : The United Nations Sustainable Development Goals. ⁹
Table 1.2: Summary of the main constituents of eccrine and sebaceous skin
secretions. ^{25, 28, 41}
Table 2.1 Summary of some of the chemical detection capabilities of the Seychelles
Police46
Table 2.2 : Photographic conditions optimised to each detection method
Table 2.2 : Fingermark grading scale and grading classification used in this study. ^{19, 219}
Table 2.3: UC (University of Canberra) comparative scale to assess the relative
performance of two detection methods for split fingermarks57
Table 3.1: Mass in grams and molar ratio of active material present in one litre of
NCFS, BKA and CAST IND/Zn working solution62
Table 3.2: Comparison of physical properties of carrier solvents used in amino acid
sensitive fingermark detection reagent. ^{238, 240, 242, 243} 66
Table 3.3: Approximate cost of one litre of 1,2-indanedione zinc working solution in
USD. Chemical sourced from two regular Seychelles Police suppliers based in the UK
and USA69
Table 3.4: Details of substrates used in the head-to-head comparison of NCFS, BKA
and CAST IND-Zn formulations70
Table 3.5: Pen used in ink diffusion investigations
Table 4.1 : Details of substrates used in the SPF IND/Zn formulation laboratory trials.
Table 5.1: Composition and approximate c.m.c of in-house surfactant solution used
in this study. The c.m.c of SDS was taken to be 8.2 mM. ²⁹⁷ 130
Table 5.2: Details of tapes used in this study. 130
Table 5.3: Details of the number of donors, tapes and powder suspension
formulations used in each experiment132
Table 5.4 Classification of adhesive and backing composition based on IR analysis of
the main polymer type135

Table 5.5: PSA tapes characteristic IR peak assignment for PSA tapes backing and
adhesive. ³⁰²⁻³¹¹
Table 5.6: Percentage of fingermarks graded 3&4 on each of the eight types of
adhesive tape152
Table 6.1: Detail of substrates used in the comparative assessment to ORO.
Table 6.2: Details of substrates used in the post-cyanoacrylate study161
Table 7.1 : Frugal forensics sustainability comparative score assessment tool183
Table 7.2: Frugal forensics method recommendation score assessment tool183
Table 7.3: Comparative score and recommendation results for SPF IND/Zn with CAST
formulation, Wet SPF powder suspension with Wetwop ^{M} formulation and post-
cyanoacrylate acid Nile blue method with BY40185

List of Figures

Figure 1.1: Schematic illustrating the influence of the "triangle of interaction" on the
initial and aged composition9
Figure 1.2: An example of a flow chart procedure for the development of latent
fingermarks. ^{16, 17} 16
Figure 1.3 : The reaction of 1,2-indanedione with amino acids to form Joullies's pink.
Figure 1.4: Chemical structure of Oil red O21
Figure 1.5: Hydrolysis of Nile blue to Nile red in aqueous solution23
Figure 1.6: Schematic illustration of powder suspension development. Fingermark
constituents disrupt surfactant micelles exposing suspended powder particulates,
causing discriminate deposition on fingermark and not the substrate26
Figure 1.7: Polymerisation reaction of ethyl cyanoacrylate27
Figure 1.8: The geographical location of Seychelles
Figure 1.9: Illustrating the gap in operationalisation of latent fingermark detection
technique in remote and small jurisdictions32
Figure 1.10: Number of publications on "forensic science" in a simple Bibliometric
search on Scopus for the (a) Total papers 1972–2022 (b)Top 10 countries and (c) Top
1034
Figure 1.11: The concept of sustainable forensic science provision (a) Definition (b)
The three principles and six attributes that underpin it
Figure 1.12: Illustrating the relationship between frugal forensics and UN sustainable
goals through SDG 1636
Figure 2.1: Schematic illustrating split fingermark and depletion series methodology
for fingermark collection and the reverse of development process A and B to
minimise donor biasing on one side of the finger44
Figure 2.2: Schematic of photographic setup for photoluminescence fingermarks
examination

Figure 2.3: Experimental setup using the fibre optic probe for fluorescence measurements of amino acid spots treated with IND/Zn formulation containing different 1,2-indanedione concentrations......54 Figure 2.4: Illustrating transformation of fingermark grades frequency results (left) into vectors (right), the basic data structure in R code to compute statistical analysis. Figure 3.1: Percentage concentration v/v of carrier solvents and co-solvents in NCFS, Figure 3.2: The reaction mechanism of IND/Zn with amino acid and zinc ions as a Lewis acid catalyst.¹⁰⁴......63 **Figure 3.3**: The molecular structure of HFE-7100 (C₄F₉OCH₃) consisting of two isomers CF₃CF₂CF₂OCH₃ and (CF₃)₂CFCF₂OCH₃.....65 Figure 3.4: The molecular structure of Solstice® PF (trans-1-chloro-3,3,3-Figure 3.5: Percentage of fingermarks graded (3&4), (1&2), and (0) out of 450 fingermarks developed in head-to-head comparison between NCFS, BKA and CAST 1,2-indanedione zinc formulation......76 Figure 3.6: Comparison of percentage fingermark grades assigned to 1st, 5th and 10th depletion series in the head-to-head comparison between NCFS, BKA and CAST 1,2indanedione zinc formulation......77 Figure 3.7: Comparison of percentage fingermark grades assigned to fingermarks developed on five substrates in the head-to-head comparison between NCFS, BKA and CAST 1,2-indanedione zinc formulation.78 Figure 3.8: Fingermarks treated with (a) NCFS formulation, (b) BKA, (c) CAST on notepad paper showing difference in luminescence and background development. Figure 3.9: Fingermarks on copy paper treated with NCFS formulation (left) and the more polar modified NCFS formulation (right) showing the marginally greater background development......82 Figure 3.10: Fingermarks on copy paper treated with modified CAST formulation containing ethanol (left) and CAST formulation containing methanol (right)...........82

Figure 3.11: Difference in luminescence of fingermarks on copy paper with modified BKA formulation containing 227 mg/L of 1,2-indanedione (left) and BKA formulation Figure 3.12: Amino acid spots on copy paper treated with BKA IND/Zn formulations containing different 1,2-indanedione concentrations, (a) emission spectrum excited at 505 nm illustrating the difference in luminescence intensity (b) difference in colour Figure 3.13: Amino acid spots on cardboard treated with BKA IND/Zn formulations containing different 1,2-indanedione concentrations, (a) emission spectrum excited at 505 nm illustrating the difference in luminescence intensity (b) difference in colour Figure 3.14: Latent fingermarks on copy paper treated with HFE-IND/Zn (left) and Sol-Figure 3.15: Comparison of percentage fingermark grades assigned to fingermarks treated with HFE-IND/Zn and Sol-IND/Zn on white copy paper, white envelope and Figure 3.16: Fingermark halves developed on (a) white envelope treated with HFE-IND/Zn (left) and Sol-IND/Zn (right) and (b) cardboard treated with Sol-IND/Zn (left) and HFE-IND/Zn (right)......88 Figure 3.17: Comparison of percentage fingermark grades assigned to a) fingermarks aged 1, 7 and 30 days b) fingermarks aged for each donor developed on white copy paper and treated with HFE-IND/Zn and Sol-IND/Zn......90 Figure 3.18: Comparison of the effect of Sol-IND/Zn and HFE-IND/Zn reagents on inks, Figure 3.19: Inks spotted with neat solvent and photographed under ambient light Figure 3.20: Sol-IND/Zn after 2.5 months in refrigerated storage, showing white precipitate......94 Figure 4.1: SPF 1,2-indanedione working solution less than 1 day old (left) and more Figure 4.2: Comparison of the effect of SPF IND/Zn and NCFS IND/Zn reagents on inks, **Figure 4.3**: Comparison of fingermarks treated with SPF IND/Zn and processed with heat press at 160 °C (left) and ambient temperature of 25 °C (right) on copy paper.

Figure 4.4: Comparison of quartered fingermarks on blue notepad paper, developed with SPF IND/Zn (top left), NCFS IND/Zn (top right), and BKA IND/ZN (bottom left) Figure 4.5: Percentage of fingermarks graded (3&4), (1&2), and (0) out of 1200 fingermarks halves developed with SPF and NCFS IND/Zn formulations on five Figure 4.6: Comparison of percentage fingermarks grades across five paper Figure 4.7: Percentage of fingermarks graded (3&4), (1&2), and (0) developed for 1 week and 4 weeks samples, treated with SPF and NCFS IND/Zn formulation.113 Figure 4.8: Examples of fingermark developed with NCFS IND/Zn (left half) and SPF IND/Zn (right half) on (a) COS Premium white copy paper, (b) OfficeMax recycled white copy paper, (c) JBurrows blue notepad, (d) PPS cardboard, (e) The West Figure 4.9: Comparison of total number of useful incidental fingermarks detected with SPF and NCFS IND/Zn treatment on five different substrates......115 Figure 4.10: Examples of incidental fingermarks developed with NCFS IND/Zn (left) and SPF IND/Zn (right half) on (a) white plain envelope, (b) JBurrows blue notepad. Fingermarks were deposited by the same donor on each type of substrate......116 Figure 4.11: Comparison of fingermarks on white copy paper stored under non-airconditioned and air-conditioned environments and treated with SPF IND/Zn......118 Figure 4.12: Comparison of fingermarks on cardboard stored under non-airconditioned and air-conditioned environments and treated with SPF IND/Zn......118 Figure 4.13: Comparison of fingermark ridge details on recycled copy paper stored under an air-conditioned environment (left), non-air-conditioned environment Figure 4.14: Percentage of useful fingermarks detected on white copy paper and

Figure 4.15: Absorption spectra for SPF IND/Zn working solutions of less than 7 days
old, 6, 12 and 18 months since preparation, showing the decrease in 1,2-indanedione
peak absorbance121
Figure 5.1: The molecular structure of materials used in the adhesive of PSA tapes, a)
acrylate monomer, b) natural rubber polyisoprene, c) synthetic rubber poly-cis-1,4-
butadiene, d) synthetic rubber polystyrene-butadiene
Figure 5.2: The molecular structure of materials used in the backing of PSA tapes, a)
polypropylene, b) polyethylene, c) polyvinyl chloride126
Figure 5.3: The molecular structure of diisononyl phthalate, a common plasticiser
used in adhesive tape to impart flexibility. ²⁸⁵ 126
Figure 5.4: The products and cost of Wet UCIO and two commercial powder
suspension formulations Wet Powder black Wetwop™ black
Figure 5.5: The molecular structure of sodium dodecyl sulfate (SDS)128
Figure 5.6: Schematic of latent fingermark deposition on tape samples
Figure 5.7: ATR-FTIR spectra of polypropylene backing of packaging tape #1-3.
Spectra have been offset for clarity137
Figure 5.8: ATR-FTIR spectra of (a) polyethylene backing of duct tape #4 and (b)
cellulose acetate backing of masking tape #5. Spectra have been offset for clarity.
Figure 5.9: ATR-FTIR spectra of polyvinylchloride backing of insulating tapes #6-#12.
Spectra have been offset for clarity138
Figure 5.10: ATR-FTIR spectra of acrylic adhesive packaging tape #1 and #2139
Figure 5.11: ATR-FTIR spectra of synthetic rubber adhesive for packaging tapes #2,
duct tape #4, masking tape #5 and insulating tape #6-#12. Spectra have been offset
for clarity140
Figure 5.12: Overlay of the fingerprint region of both backing and adhesive IR
spectrum of rubber adhesive tapes showing significant contribution of the backing in
the adhesive spectrum of tape#6 to #12141
Figure 5.13: Diisononyl phthalate IR spectrum ³¹² offset over rubber adhesive tape's
backing and adhesive spectrum of tape #9. Showing the contribution of diisononyl
phthalate in the backing and adhesive tape spectra142

Figure 5.14: Comparison of infrared spectra of different insulating tape synthetic rubber adhesive. The main differences are highlighted (blue shaded regions). 143 **Figure 5.15**: Sodium dodecyl sulfate solutions (a) insoluble solution of 27 % w/v in deionised, (b) clear solution of 15 % w/v in 5 % ethanol deionised water mixture.

Figure 5.16: Comparison of (a) the number and grades of fingermarks developed (b) the degree of powder deposition on fingermarks and background after treatment with modified WET UCIO powder suspensions prepared with 1 %, 5 %, 10 % and 27 Figure 5.17: Comparable performance and sensitivity of Wet SPF formulation (left) and 27 % SDS-modified WET UCIO formulation (right) in developing fingermarks on rubber-based adhesive tape.146 Figure 5.18: Comparison of (a) percentage of fingermarks graded (3&4), (1&2), and (0) out of 150 fingermarks (b) the degree of powder deposition on fingermarks and background after treatment with WET UCIO and Wet SPF powder suspensions on packaging tape, duct tape and insulating tape......147 Figure 5.19: Comparison of fingermarks ridge details and contrast developed on four different tapes after treatment with Wet SPF powder suspensions prepared with different additives; 5 % ethanol, 5 % PEG-400, and 10 % PEG-400 v/v......148 Figure 5.20: Fingermark ridge details and contrast after treatment with Wet SPF, WET UCIO and Wet Powder on (a) Click white insulating tape #10 under white light, (b) Figure 5.21: Percentage of fingermarks graded (3&4), (1&2), and (0) out of 320 fingermarks developed with Wet UCIO and Wet SPF on eight different types of tape.

(right) on eight pressure sensitives tapes #2 Scotch 3M clear packaging tape, #3 Scotch 3M brown packaging tape, #4 Paint Partner silver duct tape, #5 PPS white masking tape, #6 Nitto yellow insulating tape, #10 Click white insulating tape, #11 Figure 5.24: Percentage of fingermarks grades out of 300 fingermark halves developed with Wet UCIO and Wet SPF on three different types of tape grouped into depletion series 1-5 and 6-10.154 Figure 6.1: Reaction pathway proposed for the spontaneous hydrolysis of Nile Blue to Nile red in aqueous solution.¹⁴⁶.....158 Figure 6.2: The molecular structure of Basic Yellow 40 (CAS Number 29556-33-0) Figure 6.3: Comparison of fingerprints developed by filtered acid Nile blue solution (left half) and unfiltered acid Nile blue solution (right half): (a) under white light with no observable yellow stains and (b) at 450 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter.....164 Figure 6.4: Percentage of fingermarks graded 3&4 out of 72 fingermark halves Figure 6.5: Comparison of fingerprints aged 1 day, 10 days and 30 days developed by ORO (left half) under white light and acid Nile blue (right half) at 505 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter......167 Figure 6.6: Number and grades of fingermarks developed by ORO and acid Nile blue on A4 copy paper, recycled A4 copy paper, kraft brown paper and newspaper after 1 week of submersion in tap water.168 Figure 6.7: Comparison of fingerprints on different substrates submerged in tap water for 1 week and developed by Oil red O (left half) under white light and acid Nile blue (right half) at 505 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter: (a) white copy paper (b) 50 % recycled copy paper (c) brown kraft paper and (d) newspaper.....169 Figure 6.8: (a) Improvised cyanoacrylate fuming chamber (b) quality of cyanoacrylate fingermarks developed after 2 hours of fuming on borosilicate microscope slides. Figure 6.9: Comparison of cyanoacrylate fumed fingerprints stained by acid Nile blue

Figure 6.10: Percentage of fingermarks grades developed out of 30 fingermark halves
with acid Nile blue and Basic Yellow 40 on borosilicate glass, aluminium sheet, and
polypropylene plastic lid171
Figure 6.11: Comparison of fingermarks developed with acid Nile blue (left half) and
Basic Yellow 40 (right half) illustrating donor effect: (a) on borosilicate glass for donor
1 (b) on aluminium sheet for donor 1 (c) borosilicate glass for donor 3 (d) aluminium
sheet for donor 3 (e) on polypropylene superimposed of donor 1 (top) and donor 3
(bottom)172
Figure 6.12: Overlapping fingermark collection (a) natural over charged fingermarks
(b) charged over natural fingermarks174
Figure 6.13: Overlapped fingermarks developed with cyanoacrylate and stained using
the sequence BY40 $ ightarrow$ acid Nile blue, (top) image of results at different development
stages and (bottom) assessment result using UC scale175
Figure 6.14: Overlapped fingermarks developed with cyanoacrylate and stained using
the sequence acid Nile blue $ ightarrow$ BY40, (top) image of results at different development
stages and (bottom) assessment result using UC scale175
Figure 6.15: Raman spectra of acid Nile blue and Basic Yellow 40 liquid smear on glass
slide, acquired using WITec alpha 300SAR confocal Raman microscope equipped with
a 532 nm NdYAG laser177
Figure 6.16: Raman spectra of unstained cyanoacrylate fingermark (red),
cyanoacrylate fingermark stained with acid Nile blue (blue) and cyanoacrylate
fingermark stained with Basic Yellow 40 (yellow) on a glass slide, acquired using
WITec alpha 300SAR confocal Raman microscope equipped with a 532 nm NdYAG
laser177
Figure 6.17: Visual image of the cross-section of post-cyanoacrylate stain fingermark
on glass slide. The fingermark and polymer structure not apparent. The glass side was
mounted edge-up
Figure 6.18: Visual and Raman image overlay of the cross-section of post-
cyanoacrylate stain fingermark on glass slide showing fluorescent and non-
fluorescent regions

List of Abbreviations

ATR	Attenuated total reflection
AUD	Australian Dollars
AWRE	Atomic Weapons Research Establishment
ВКА	Bundeskriminalamt
BY40	Basic Yellow 40
ca.	Circa
cmc	Critical micelle concentration
CAST	Centre for Applied Science and Technology
DFO	1,8-diazafluoren-9-one
DNA	Deoxyribonucleic acid
EU	European Union
FTIR	Fourier transform infrared
FVM	Fingermark Visualisation Manual
GC-MS	Gas-chromatography mass spectrometry
GDP	Gross Domestic Product
GWP	Global Warming Potential
HDPE	High-density polyethylene
HFE-7100	Methyl nonafluorobutyl ether
HFEs	Hydrofluoroethers

HSP	Hansen solubility parameters
IFRG	International Fingerprint Research Group
IND/Zn	1,2-indanedione zinc
IR	Infrared
JP	Joullie's pink
LDPE	Low-density polyethylene
LLDPE	Linear low-density polyethylene
MALD-MSI	Matrix-assisted laser desorption/ionisation mass spectrometry imaging
NCFS	National Centre for Forensic Studies
NHMRC	National Health and Medical Research Council
NSW	New South Wales
ORO	Oil red O
PAACSS	Performance, Accessibility, Availability, Cost, Simplicity, Safety
PD	Physical developer
PEG	Polyethylene glycol
PFAS	Per- and polyfluoroalkyl substances
PSA	Pressure sensitive adhesive
PVC	Polyvinyl chloride
RH	Relative humidity
SDGs	Sustainable Development Goals

- SDS Sodium dodecyl sulfate
- SPF Seychelles Police Force
- UC University of Canberra
- UK United Kingdom
- USA United States of America
- USD United States dollars
- UV Ultraviolet
- WA Western Australia
Portions of this chapter have been published in the following articles:

Jemmy T. Bouzin, Georgina Sauzier, and Simon W. Lewis. Forensic science in Seychelles: An example of a micro-jurisdiction forensic delivery system. *Forensic Science International: Synergy* **2021** *3* 100139.

DOI: https://doi.org/10.1016/j.fsisyn.2021.100139

Jemmy T. Bouzin, Aaron J. Horrocks, Georgina Sauzier, Stephen M. Bleay, and Simon W. Lewis. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry* **2022** *30* 100439. DOI: https://doi.org/10.1016/j.forc.2022.100439

Jemmy T. Bouzin, Thais Lópes, Anna L. Heavey, Jessie Parrish, Georgina Sauzier, and Simon W. Lewis. Mind the gap: The challenges of sustainable forensic science service provision. *Forensic Science International Synergy* **2023** *6* 100318. DOI: <u>https://doi.org/10.1016/j.fsisyn.2023.100318</u>

1.1 Global importance of fingermark detection

Forensic science has become a global practice,^{1, 2} serving as a means to promote justice, peace and prosperity,³⁻⁷ aligning with the United Nations Sustainable Development Goals (SDGs)⁸ - especially SDG16 (Table 1.1). These 17 goals, universally adopted by United Nations Member States in 2015⁹ aim to protect the planet and reduce disparities between the Global North and South. The terms 'Global North' and 'Global South' are not geographic labels but relate to socio-economic development and global influence.¹⁰ The Global North is usually equated with technically developed and well-resourced countries, whereas the Global South often denotes politically or culturally marginalised regions with comparatively limited income. In nations of the Global South, where resources are limited and sophisticated techniques such as DNA profiling are not readily available, latent fingermark detection serves as a fundamental service provision¹¹ that plays an instrumental role in advancing the global practice of forensic science.

For over a century,¹² fingermarks have been a crucial forensic tool in criminal investigations, utilised across various jurisdictions,¹³ regardless of their size or technological advancement. Despite their widespread and longstanding use, the detection of fingermarks remains a challenging task, particularly for developing regions in the Global South. This is attributed to the inherent fingermark variability coupled with the Global North hegemony in the research and development of fingermark enhancement methods.

A fingermark is a mirror image of the ridge pattern of the finger, formed by the deposition of residue from the fingertip when in contact with a surface.^{14, 15} Due to their highly polymorphic nature, fingermarks can be used to identify and link people, places, and things involved in criminal activity. Fingermarks found at scenes are often latent (invisible) to the naked eye, requiring chemical or physical treatment to visualise. The detection process involves exploiting the fingermark and substrate's physical and chemical properties to produce appropriate contrast.¹⁶⁻¹⁸ However, the detection effectiveness can be affected by the fingermark composition, the nature of the substrate and environmental conditions. This is because fingermark composition

is highly variable,¹⁹⁻²³ and substrates and conditions where they may be found are diverse. Consequently, various detection techniques have been developed through research, but no global optimum exists.^{24, 25} This is underpinned by guidelines of the International Fingerprint Research Group (IFRG), which recommends evaluation of new or modified enhancement reagents using locally sourced test substrates and visualisation equipment.²⁴

Against this backdrop, it is important to note that research to continuously develop more effective procedures for improving forensic services has primarily been carried out in countries with ample resources. This is evident from the dominance of the Global North in fingermark detection research output,^{26, 27} leading to a lack of techniques that have been developed and validated for jurisdictions in the Global South. Particularly, those that address the challenges such as supply chain, budgetrestrictions, and limited resources. Therefore, there is a need to develop alternative fingermark detection capabilities adapted to the vulnerabilities of the Global South so that they can be implemented effectively. This applies to fingermark detection provision and other forensic science services alike.

This thesis presents a framework for sustainable forensic science provision, to address the gap in fingermark service provision outlined above. To introduce the concept, this chapter begins with an overview of the chemistry of latent fingermarks and the factors that affect its composition, followed by the various fingermark detection techniques available, and the challenges in implementation in resourcelimited jurisdictions. Finally, the framework and principles of sustainable forensic science is discussed, followed by the aims and objectives. This project centres on sustainable latent fingermark detection methods, focusing on the challenges faced by Seychelles, a small and remote jurisdiction with limited resources.

Table 1.1 :	The United	Nations Su	stainable	Development	Goals. ⁹
--------------------	------------	------------	-----------	-------------	---------------------

Goal Number	Goal Name	Goal Description
SDG 1	No Poverty	End poverty in all its forms everywhere
SDG 2	Zero Hunger	End hunger, achieve food security and improved nutrition and promote sustainable agriculture
SDG 3	Good Health and Well-being	Ensure healthy lives and promote well-being for all at all ages
SDG 4	Quality Education	Ensure inclusive and equitable quality education and promote lifelong learning opportunities for all
SDG 5	Gender Equality	Achieve gender equality and empower all women and girls
SDG 6	Clean Water and Sanitation	Ensure availability and sustainable management of water and sanitation for all
SDG 7	Affordable and Clean Energy	Ensure access to affordable, reliable, sustainable and modern energy for all
SDG 8	Decent Work and Economic Growth	Promote sustained, inclusive and sustainable economic growth, full and productive employment and decent work for all
SDG 9	Industry, Innovation and Infrastructure	Build resilient infrastructure, promote inclusive and sustainable industrialisation and foster innovation
SDG 10	Reduced Inequalities	Reduce inequality within and among countries
SDG 11	Sustainable Cities and Communities	Make cities and human settlements inclusive, safe, resilient and sustainable
SDG 12	Responsible Consumption and Production	Ensure sustainable consumption and production patterns
SDG 13	Climate Action	Take urgent action to combat climate change and its impacts
SDG 14	Life Below Water	Conserve and sustainably use the oceans, seas and marine resources for sustainable development
SDG 15	Life on Land	Protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertification, and halt and reverse land degradation and halt biodiversity loss
SDG 16	Peace, Justice and Strong Institutions	Promote peaceful and inclusive societies for sustainable development, provide access to justice for all and build effective, accountable and inclusive institutions at all levels
SDG 17	Partnerships for the Goals	Strengthen the means of implementation and revitalise the global partnership for sustainable development

1.2 Fingermark composition

The residue left behind by a fingermark is a complex mixture of natural secretions, contaminants, and substances present on the skin of the finger.^{14, 28, 29} These secretions originate from three types of sweat glands: eccrine, apocrine, and sebaceous.^{30, 31} Sweat glands are essential for regulating electrolytes and body temperature, and removal of metabolic products.³¹⁻³⁴ In the context of latent fingermark detection, the composition originating from sweat glands is broadly classified as either eccrine or sebaceous. The eccrine secretion comprises inorganic salts and water-soluble organic compounds, while the sebaceous secretion consists of lipids and fats (summarised in Table 1.2).^{28, 29, 35, 36} Due to the hydrophilic nature of eccrine and the hydrophobic nature of sebaceous secretions, they are respectively referred to as the water-soluble deposit and water-insoluble deposit of latent fingermark residue.^{16, 17}

The composition of the residue varies from person to person based on donor traits such as biological sex, age, ethnicity, psychological state, health, medication, and diet.^{15, 32, 37, 38} Personal habits and activities, such as smoking or use of hair and cosmetic products, may introduce endogenous and exogenous contaminants.^{35, 36} Additionally, the composition of the initial fingermark residue deposited on a surface is influenced by deposition conditions, such as contact pressure, angle, and time duration, as well as the characteristics of the substrate.^{16, 36} These factors can cause high variability in the composition of fingermark residue.

Knowledge of the composition and influencing factors of fingermarks is key to understanding its effect on detection performance and sampling protocol in a research context. For example, fingermark test methodology primarily uses 'natural' fingermarks collected from donors conducting their normal daily routine instead of sebaceous 'charged' fingermarks, where donors wipe their fingers across the face or forehead prior to the deposition of marks.^{19, 24} Studies show that charged samples are lipid-rich,^{39, 40} potentially biasing results since it may be an unrealistic representation of marks encountered in operational work. Hence, the use of charged fingermarks is limited to the early phase trial, particularly when establishing interactions between the fingermark and the technique.^{19, 24}

Secretions	Organic constituents	Inorganic constituents
Eccrine	Amino acids	Water (>70%)
	Proteins	Chloride
	Urea	Metal ions (Na+,K+, Ca ²⁺)
	Uric acid	Sulphate
	Lactic acid	Phosphate
	Sugars	Bicarbonate
	Creatinine	Ammonia
	Choline	
Sebaceous	Glycerides	
	Fatty acids	
	Wax esters	
	Squalene	
	Sterol	
	Sterol esters	

 Table 1.2: Summary of the main constituents of eccrine and sebaceous skin secretions.

1.2.1 Eccrine components

The primary function of eccrine glands is to dissipate body heat through sweat and excrete excess water. Eccrine glands are the only type of secretory glands present on the palms and fingertips, making them a major contributor to the composition of fingermark residue. Eccrine residue primarily consists of amino acids, lactic acid, proteins, urea, uric acid, sugars, and inorganic salts.^{35, 36}

Amino acids have been studied as sweat and fingermark constituents, as they are a primary target in fingermark detection on porous surfaces that have not been wet.^{31, 42-44} So far, up to 20 amino acids have been quantified, with serine being the most abundant, followed by glycine and alanine.^{39, 45} Croxton *et al.*³⁹ reported the total amino acid content to range from 20.7 ng to 345.1 ng per fingermark and did not find significant difference between charged and natural fingermarks. A recent study by Helmond *et al.*²³ profiling the chemical composition of fingermarks from 463 donors using mass spectrometry, reported large inter-variability in the amino acid profile and a content range from 100 ng to 10 μ g. The amount and profile of amino acids depend on several factors, including sex, age, diet, physical exercise, and general health.^{28, 35, 36}

Inorganic compounds including salts and trace amounts of metals such as sodium, potassium, and calcium have been detected in eccrine residue.²⁸ Sodium chloride and potassium chloride salts are the most abundant. Studies suggest that the amount of chloride ions in fingermarks decreases as the donor's age increases .^{36, 46, 47} Other than chloride ions targeted by silver nitrate, most of the relatively less reactive inorganic compounds are not subject of fingermark treatment.²⁸

The average water content of eccrine sweat has long been reported in the literature to be 98 % by weight.^{31, 34, 48} This has been perpetuated in a similar reported amount in fingermark residue.⁴⁹ According to Kent's⁵⁰ theoretical argument, the suggested amount is unrealistic as it fails to consider the presence of sebaceous secretions, which are primarily composed of lipids. Kent instead suggested the water content would be close to 20 % or less. Keisar *et al.*⁴¹ verified this claim by utilising quartz crystal microbalance and temperature-programmed desorption-mass spectrometry to measure the complete water loss through evaporation. Their findings revealed a varying water content of 20-70 %.

1.2.2 Sebaceous components

Sebaceous glands are found in varying abundance all over the body except on the palms of the hands and soles of the feet.³¹ Sebum, a secretory product of these glands, forms a major component of fingermark composition due to contact of the fingertips with other parts of the body (e.g., hair and face).⁵¹ Sebum and compounds from the epidermis contribute to the lipid composition of fingermark residue consisting of triglycerides, fatty acids, wax esters, cholesterol, and squalene.^{15, 39}

The lipid composition has been studied both as skin surface and fingermark residue, and a good correlation between the two compositions has been found.^{36, 39, 52-54} Free fatty acids, which form 15-25 % of sebum, are major lipid compounds identified in fingermark residue with palmitic acid identified as the most abundant.⁵³ The fatty acids found in sebum are quite diverse, from volatile short-carbon chain fatty acids (<C10) partly responsible for body odour,⁵⁵ to more stable long-carbon chain fatty acids (C30) targeted in the treatment of older fingermarks. Wax esters produced

exclusively by the sebaceous glands make up 20-25 % of the fingermark's residue, primarily comprising fatty acids and fatty alcohols. Squalene, a cholesterol precursor produced by all tissues, accounts for 10-15 % of fingermark residue and is usually not detected in older fingermarks due to rapid decomposition over time.^{52, 56} Sterol and sterol esters originate mainly from the epidermis and form 1-5 %, with cholesterol and cholesterol esters being the most abundant. Triglycerides are estimated to make up 30-40 % of fingermark residue and are the least studied due to most analytical methods involving hydrolysis which break them down into glycerol and fatty acids.³⁵

Whilst many lipid compounds are common to fingermark residue, the relative composition varies between individuals, based on donor traits and deposition characteristics, including the amount of contact of the finger with other body parts. Several studies have highlighted the inherent variability in the lipid compositions of fingermark residue.^{18, 35, 36} Frick *et al.*⁵⁷ demonstrated the high intra-donor variability in the lipid components of fingermark residue of 116 donors using gas chromatography - mass spectrometry (GC-MS). Croxton *et al.*³⁹ reported significant quantitative differences in amounts of fatty acids and squalene between natural and charged fingermarks from the same donor. These studies illustrate fingermarks' inherent variability and the importance of appropriate sampling protocol in a research context.

1.2.3 Other contaminants

In addition to secretory glands and epidermis, fingermark composition is also affected by contaminants.⁵⁸⁻⁶² There are two primary sources of contaminants: contact of the fingertip with exogenous substances and excreted metabolites on the skin surface.

Everyday activities can transfer materials onto fingertips, mixing with the fingermark residue. Fingermark residues have been found to contain cosmetics and other consumer products like hair products, sunscreen, hand sanitisers and lotions.^{59, 60, 62-64} Compounds with additional forensic value, such as gunshot residue, illicit drugs and explosives, have also been identified.^{61, 65} Other studies have successfully

identified metabolites in sweat and fingermark residue related to smoking habits or drug consumption, which can provide additional donor information.^{66, 67} The presence of these contaminants can affect the chemistry of the residue, potentially interfering with detection processes and affecting degradation rates. For example, handling oil-based substances such as food, lotion, and hair products can increase the lipid fraction in a fingermark residue,⁶³ impacting the ageing and detection of latent fingermarks.⁶⁸

1.3 Factors affecting fingermark composition post-deposition

Following deposition, the initial composition of the latent fingermark residue undergoes further changes, the nature of which depends on the interactions with the substrate and environmental conditions with time. This process is illustrated by the triangle of interaction¹⁹ in Figure 1.1, which depicts the relationship between the three elements over time. The resulting aged composition of the fingermark residue is a composite of the triangle of interaction from deposition to the detection time, and impacting the ability of a fingermark to persist, and ultimately its detection.³⁵ Therefore, understanding how aged fingermarks evolve is crucial for the successful recovery of latent fingermarks. Several studies^{52, 69-72} have explored the evolution of fingermark composition over time to develop effective detection techniques, while others⁷³⁻⁷⁷ have investigated ageing markers to develop methods for determining fingermark age.



Figure 1.1: Schematic illustrating the influence of the "triangle of interaction" on the initial and aged composition.

1.3.1 Effect of time

Fingermark residue undergoes ageing between deposition and detection through various chemical, physical, and biological processes. These processes include degradation, drying, evaporation, metabolism, migration, oxidation, and polymerisation.^{35, 36, 78, 79} These changes are influenced by factors such as the substrate deposited on and the environmental conditions,^{35, 73} ultimately affecting the mechanism and rate of the fingermark ageing process.

Over time, the mass of fingermark residue decreases mainly through evaporation of volatile components.^{21, 72} The loss of moisture through this process has been shown to cause physical changes in the thickness of the fingermarks due to cumulation of residual organic and inorganic components.^{71, 80} While some studies have investigated the inorganic salt compounds, such as the concentration of chloride ions over time,⁴⁷ most of the focus has been on the organic compounds, which are the most targeted by fingermark detection techniques.

Amino acids have been suggested to be stable over time, particularly on paper substrates due to hydrogen bonding with cellulose.⁸¹ This has been practically demonstrated with successful detection of old fingermarks using amino acid sensitive reagents on 32 and 80 year old documents. ^{82, 83} However, it has been generally reported that with increasing age of fingermarks, there is a significant decrease in the quality, indicating that amino acids binding with cellulose are not completely stable.⁶⁹ The progressive diffusion over time of the amino acids in the paper matrix has been suggested as a primary mechanism for the decrease in quality of developed fingermarks.^{35, 83}

Lipids undergo degradation and decomposition over time, with unsaturated compounds such as triglycerides and squalene found to be less stable compared to saturated lipids such as wax esters, which are less affected.^{52, 72, 76, 84} Pleik *et al.*^{78, 85} suggested that ozonolysis is a major lipid degradation pathway in fingermark residues in ambient air, primarily through oxidation of unsaturated triglycerides to form ozonides (1,2,4-trioxolanes), which are proposed as a possible aged marker. More

recently, Frick *et al.*⁸⁶ demonstrated that diunsaturated triglycerides in fingermarks undergo progressive oxidation to mono- and diozonides of fingermarks when exposed to ambient atmospheric conditions, using untargeted ultra performance liquid chromatography-ion mobility spectrometry-quadrupole time-of-flight mass spectrometry and five donors. The study found that the number of unsaturated triglycerides decreased in samples left for 7 and 28 days, while mono- and diozonides of these lipids were identified as the main components, indicating that ozonolysis is a major pathway for fingermark degradation over time.

The physicochemical changes of latent fingermark residue over time have been explored in relation to fingermark age determination. Various indicators such as fingermark quality developed with powdering techniques,^{74, 87} changes in the ridge width thickness,^{71, 75} variations in lipids^{52, 88} and amino acid concentration ^{77, 89} have been studied. However, no reliable method for determining fingermark age has been established, primarily due to the intrinsically variable nature of fingermark composition. In a recent study, Boseley *et al.*²¹ investigated the changes within fingermark residue of 12 donors from deposition for the first 7-13 hours under ambient conditions using synchrotron-sourced attenuated total reflection-Fourier transform infrared (ATR-FTIR) microspectroscopy. The study revealed high variation in the initial composition and morphology of fingermarks, with substantial loss of water (14-20 µg) within the first 8 hours and relatively higher variation in lipid concentration due to degradation and redistribution.²¹ The study highlighted the unpredictability of fingermark age estimation, particularly when the chemistry of the initial composition is unknown.

In essence, the detection of latent fingermarks in operational cases is generally carried out days, months, or even years after deposition, hence the detectability of fingermarks depends on the stability over time of the target compounds and sensitivity of the development technique. Between deposition and processing, fingermarks undergo ageing through various processes; however, these mechanisms and the rate of the ageing processes are also influenced by the nature of the substrate and environmental conditions as discussed below.

1.3.2 Substrates

The nature of the substrate plays a crucial role in influencing the fingermark deposit and in determining the appropriate enhancement method.^{90, 91} For purposes of latent fingermark detection, substrates are grouped into porous, non-porous and semi-porous. The porosity dictates the capacity of the substrate to absorb fingermarks and, consequently, their persistence.

Porous substrates like paper or cardboard tend to absorb fingermark deposits to varying degrees, with different fractions of the residue penetrating to different depths.⁹¹ Eccrine materials, which include amino acids, are more easily absorbed and retained due to hydrogen bonding with cellulose in the substrate matrix.⁸¹ In contrast, lipid material is less likely to penetrate due to factors such as lower wettability.^{15, 68} Almog *et al*.⁹¹ demonstrated the absorption of fingermarks using amino acid sensitive reagents and fluorescence microscopy to study the depth of penetration of fingermarks cross-sections on paper. They found that the depth of penetration of fingermarks on paper correlates with the substrate's porosity, with better quality developed marks produced at depths between 40-60 microns.

When fingermarks are left on non-porous substrates like glass and plastics, they are not absorbed into the material and remain exposed on the surface.⁶⁸ This can cause changes to the chemical profile of the mark, such as the loss of water and other volatile substances. For instance, it was observed that the volatile compound squalene decreased more quickly on glass surfaces than on paper over a 30 day period.⁷⁶

The ability of different substrates to absorb and retain fingermark residue is crucial in determining their persistence and recoverability.¹⁶ Fingermarks on non-porous substrates are more prone to physical damage, degradation, and evaporation, while porous substrates tend to minimise exposure to these factors, resulting in increased persistence. This is demonstrated by studies showing that fingermarks on paper substrate can be developed using amino acid sensitive reagents, even after several decades.^{82, 83} In addition, the chemistry and reactivity of a substrate with fingermark

residue can also impact recoverability. Fingermarks on metal surfaces, for instance, can cause corrosion,^{92, 93} altering the properties of the fingermark deposit and affecting its recoverability. Within the same context, new substrates, such as biodegradable plastics, have prompted investigation^{94, 95} of their impact on fingermark enhancement methods.

1.3.3 Environmental conditions

Various environmental factors, including temperature, light exposure and humidity, can impact the composition and rate of change to the fingermark residue as well as the substrate deposited on.^{52, 74} Studies have shown that exposure to high temperatures can lead to accelerated water loss, volatile materials and chemical degradation of fingermark constituents. Richmond-Aylor *et al.*⁹⁶ explored the amino acid degradation products using pyrolysed GC-MS on extracted fingermark residue. Decomposition products such as 2,5-furandione from aspartic acid and 3,6dimethylpiperazine-2,5-dione from alanine were identified. Some of these compounds were found to fluoresce, providing an additional enhancement method for fingermarks subjected to extreme temperatures.^{97, 98} Similar results of thermal degradation have been observed with urea.^{98, 99} Elevated temperatures also affect the lipid components, with Wolstenholme *et al.*¹⁰⁰ revealing the degradation of oleic acid at 60 °C using matrix-assisted laser desorption/ionisation mass spectrometry imaging (MALDI-MSI). In a more recent study, Kim *et al.*¹⁰¹ explored the degradation of squalene and cholesterol over a short exposure time of 8 hours using GC-MS and reported significant loss of both compounds at 100 °C.

Light exposure has also been found to increase the oxidation rate of the lipid material. Archer *et al.*⁵² found that squalene and oleic acid degradation were more rapid for fingermarks stored in light compared to dark conditions. Squalene was found to be undetectable after 9 days in light conditions, while it lasted for 33 days in the dark. Mountfort *et al.*⁵⁶ demonstrated the rapid oxidation of squalene in the presence of a photoxidiser and identified squalene monoxide and squalene epoxide as oxidation products in fingermark residue within 1 day of light exposure. Several

other studies have reported the rapid degradation of squalene under light exposure with UV radiation light having a greater impact.^{84, 102}

Although the impact of humidity on detection methods has been studied, little research has been conducted on its effect on fingermark composition. Paine *et al.*¹⁰³ found that the quality of cyanoacrylate fume prints varies with humidity. Relative humidity (RH) of 80 % was found to develop the best quality fingermarks. This was attributed to water absorption by sodium chloride salt crystals, influencing the initiation of polymerisation and the type of microstructures formed. In contrast, a previous study using silver nitrate⁴⁷ found that higher humidity had a detrimental effect on developed fingermarks, suggesting that chlorides diffuse faster under high RH, making silver nitrate less effective on older fingermarks. The variation between jurisdictions in the effectiveness of amino acid sensitive reagent 1,2-indanedione in its early development has also been attributed to differences in humidity, with the addition of zinc chloride to the formulation minimising this effect.¹⁰⁴ These studies show that humidity plays a crucial role in the composition of eccrine material. Water submersion has a major influence on the persistence of eccrine material, with most of the aqueous residues washed away and the lipid constituents least affected. This is supported by successfully detecting fingermarks on wetted substrates using lipid targeting detection methods such as physical developer (PD) and Oil red O (ORO).¹⁰⁵⁻ 107

It is important to note that although environmental factors can be considered in isolation, from an operational perspective, a combination of several environmental factors are encountered, which may have a greater impact than a single factor.¹⁶ Additionally, the climatic difference between geographical locations needs to be considered when developing or introducing operational detection methods.

1.4 Fingermark detection methods

The basis of the detection process is to produce appropriate contrast between the fingermark and the substrate to enable the discrimination required to capture and record an image. This can be achieved through optical, chemical and physical

processes that exploit certain properties of the fingermark and substrate.^{16, 17} Optical processes exploit the reflectivity, refractive index, emissivity, colour or fluorescence properties of fingermarks and the deposited surface when illuminated or irradiated to enhance contrast.¹⁷ The physical properties of fingermarks, such as their adhesive nature or induced electrical charge, may also be exploited for visualisation. For example, fingermark powder development is a method that takes advantage of the preferential adhesion of powder particulate to the fingermark constituents rather than the substrate.¹⁶ Chemical processes target the chemical constituents of fingermark deposits to form coloured or fluorescent products that improve contrast with the substrate.^{16, 17}

Optical processes are applied to all types of fingermarks and are generally nondestructive, while physical and chemical processes are more selective and target specific constituents of the residue.⁶⁸ Considering the array of conditions under which a fingermark could be encountered, it is important for practitioners to understand which constituents are being targeted by each process to enhance fingermarks under various conditions effectively.

Enhancement treatments can be applied sequentially to maximise recoverability, with optical processes applied first and in between other detection methods. Physical and chemical processes are applied in order of least to most destructive, generally in the order of liquid-free, organic solvent and water-based processes.¹⁶ Figure 1.2 shows an example of a flow chart practitioners use to select enhancement treatment orders. Practitioners must also consider factors like time, substrate, and environment, which can alter fingermark composition and impact detection effectiveness.

While the most effective detection sequence may be known, resource limitations can impact practitioner's choices. These limitations are jurisdiction-specific and include restricted budget, availability and accessibility to reagents, personnel and equipment, and supply chain disruptions. This section outlines common detection methods and discusses some of the challenges associated with their operational use in resource-limited jurisdictions.

15



Figure 1.2: An example of a flow chart procedure for the development of latent fingermarks.^{16, 17}

1.4.1 Porous-surface fingermark detection methods

Porous surfaces such as paper and cardboard are frequently encountered in criminal investigations. As discussed earlier, on these surfaces, water-soluble deposits like amino acids, urea, and chlorides get absorbed quickly and are contained within the substrate. In contrast, the water-insoluble deposits remain on top of the surface, with only a small amount persisting over time. This means the fingermark pattern based on the water-soluble deposit is well preserved, and an image can be developed with amino acid sensitive reagents. However, environmental factors like water can wash away amino acids while diffusion of urea and chlorides due to humidity can affect water-soluble deposits.³⁶ In such cases, lipid-sensitive detection methods are used.

1.4.1.1 Amino acid sensitive reagents

Various amino acid sensitive reagents have been established over the years. Research in this area has mostly centred around the synthesis and optimisation of ninhydrin analogues,^{25, 108-110} with an emphasis on increasing the sensitivity and the use of luminescence detection.^{14, 29} Nonetheless in some instances the accessibility and cost of the development reagents limit its use in some jurisdictions.

1.4.1.2 Ninhydrin and 1,8-diazafluoren-9-one (DFO)

Ninhydrin (2,2-dihydroxy-1,3-indanedione) was discovered in 1910 by Ruhemann.¹¹¹ It was the first amino acid sensitive reagent to be used for latent fingermark detection. Ninhydrin reacts with amino acids to produce a purple product known as "Ruhemann's purple" through the generally accepted mechanism of Strecker degradation.²⁵ Ruhemann's purple does not have photoluminescent properties unless further treated with metal salts¹⁶ which makes the method cumbersome. Photoluminescence is desirable as it offers better sensitivity and contrast, particularly on coloured or patterned surfaces. This in part led to further investigation into ninhydrin analogues¹¹²⁻¹¹⁴ that can produce both coloured and photoluminescent products in reaction with amino acids. Two reagents that became of particular interest were 1,8-diazafluoren-9-one (DFO) and 1,2-indanedione. DFO was first synthesised by Druey and Schmidt in 1950²⁵ and introduced as a fluorescent detection technique for latent fingermarks by Pounds and Grigg in 1990.¹¹² Although DFO is not a direct analogue of ninhydrin, it follows a similar reaction mechanism to produce a red product, which is fluorescent under excitation in the green region of the spectrum (430 to 580 nm).⁶⁸ Unlike ninhydrin, the reaction does not require secondary treatment with metal salts. The advantage of a one-step reaction and the noticeable increase in sensitivity compared to ninhydrin, lead to DFO gaining widespread use in the UK.¹⁶

1.4.1.3 1,2-indanedione

1,2-indanedione is a more recent amino acid sensitive detection method for porous surfaces. It was first proposed as a fingermark reagent in 1997 after initial observation in the 1990s at the University of Pennsylvania.^{113, 114} The compound was shown to be sensitive to amino acids and produce a highly fluorescent, bright pink complex known as Joullie's pink (JP) (Figure 1.3).¹⁰⁸ The reaction mechanism has been characterised to involve Strecker degradation¹¹⁵ similar to ninhydrin and DFO. Joullie's pink has a maximum absorption at 550 nm and a luminescence emission at 555 nm when excited with a green light source (500–530 nm).



1,2-indanedione

Joullie's pink

Figure 1.3: The reaction of 1,2-indanedione with amino acids to form Joullies's pink.

Following the promising initial study of 1,2-indanedione as an alternative to ninhydrin and DFO, researchers worldwide conducted comparative assessment. Variable results were obtained, with research groups in Australia¹¹⁶ and Israel¹¹⁷ reporting superiority of 1,2-indanedione over DFO, whilst researchers in Canada¹¹⁸ and the UK^{119, 120} found the reverse. The contradicting results were highlighted in a

2004 survey conducted by Wallace-Kunkel *et al.*¹²¹ in Australia, New Zealand, USA, UK, and Europe on detecting and enhancing latent fingermarks on porous surfaces. The authors reported a lack of widespread use of 1,2-indanedione, with only 28 % of respondents having used it in casework. This is likely due to various formulations and processing conditions adopted by different laboratories worldwide, as well as differences in the chemistry of the local substrate or environmental conditions.

A modified 1,2-indanedione formulation was developed for Australian conditions containing premixed zinc chloride, and was found to be more resilient to humidity fluctuations.^{115, 122} Spindler *et al.*¹⁰⁴ characterised the role of zinc(II) in the 1,2-indanedione zinc (IND/Zn) formulation as a Lewis acid catalyst, stabilising a key intermediate during a rate-limiting hydrolysis step in forming JP. IND/Zn is recognised as the most sensitive amino acid fingermark detection method on porous surfaces following several comparative studies¹²²⁻¹²⁵ to ninhydrin and DFO. This has led to its widespread use in various jurisdictions.^{124, 126, 127} More recently, pseudo-operational trials^{125, 127, 128} conducted in the UK demonstrated the superiority of IND/Zn to DFO, and it is expected to be fully implemented in casework by the UK Home Office Centre for Applied Science and Technology (CAST) Fingermark Visualisation Manual. However, implementation in micro-jurisdictions such as Seychelles remains a concern due to cost and supply chain disruption of the standard carrier solvent HFE-7100 (methyl nonafluorobutyl ether) used in the formulation.

1.4.1.4 Lipid-sensitive reagents

When items have been wetted, commonly used amino acid sensitive reagents are ineffective since amino acids dissolve in water. In such cases, lipid-sensitive reagents such as PD, ORO or Nile red are used to target the water-insoluble components. The water-insoluble deposit can be grouped into 'fragile' and 'robust' fractions.^{105, 107, 129, 130} The robust fraction consists of proteins and lipo-proteins that can form hydrogen bonds with the cellulose content of paper, allowing them to remain on the surface for longer periods of time. The fragile fraction includes unsaturated fatty acids, lipids, and triglycerides, which quickly undergo chemical changes when exposed to air.⁷⁸ It is suggested that the lipid reagents target different fractions of the water-insoluble

deposit, given variation in the quality of developed marks between fresh and older fingermarks. However, some of these methods are complex and use toxic reagents, limiting their use in some jurisdictions due to cost and access to trained personnel and equipment. A prime example is the use of PD.

1.4.1.5 Physical developer (PD)

The PD technique was developed as a fingermark detection method in early 1970 at the Atomic Weapons Research Establishment (AWRE) in the UK.⁶⁸ Its development was based on an adaptation of the silver nitrate solution development process that was commonly used in the photographic industry. The PD solution contains silver ions, ferrous and ferric salt, citric acid and a surfactant in a delicate formulation.¹⁷ It works by selectively depositing silver particles on the ridges of a fingermark, resulting in a light grey image. Due to alkaline binders and fillers present in some paper substrates, an acid prewash with maleic acid forms part of the treatment process to avoid background development.^{16, 106} PD has been reported to be an effective method in developing fingermarks exposed to extreme conditions such as, immersed in water,¹³¹ temperatures excess of 100 °C¹³² and up to 90 year old fingermarks⁸² have been successfully developed. Nonetheless, the complexity of the method (e.g. requiring several different stock solutions and baths, along with sensitivity to light and contamination) and its high cost limit its use in some jurisdictions.^{133, 134}

1.4.1.6 Oil red O (ORO)

Oil red O (1-([4-(xylylazo)xylyl]azo)-2-naphthol) is another alternative method for detecting fingermarks on wet porous surfaces. It is a lysochrome used as a biological stain for lipids since the 1920s.¹³⁵ In the early 2000s, ORO was used as a reagent to visualise lip marks¹³⁶ and in 2004 Beaudoin *et al.*¹²⁹ introduced it as a latent fingermark detection technique. ORO is a diazo dye (R–N=N–R') with non-ionising structural conformation shown in Figure 1.4. Hence it can dissolve in the lipid components of fingermark residue, staining the ridges red and the background pink.



Figure 1.4: Chemical structure of Oil red O.

The ORO process developed by Beaudoin *et al.*¹²⁹ consists of three steps (staining, neutralisation and drying) and two stock solutions; the stain solution dissolved in methanol and a buffer solution. The reagent and treatment processes are more economical and less tedious than PD. However, the ORO process was found to be more time-consuming taking up to 90 minutes to develop fingermarks.⁶⁸ Frick *et al.*¹³⁷ proposed a non-toxic modified ORO formulation by dissolving the stain in propylene glycol instead of methanol. The amount of ORO in the formulation was reduced, making it more straightforward and economical to prepare. The development process was also modified, involving a 15-minute immersion in the reagent, resulting in an efficient process in contrast to the original method. The authors reported comparative performance between the original and modified ORO detection techniques on various paper substrates, including under wet conditions.

Several studies have conducted comparative assessments of ORO and PD. Studies using lipid-rich fingermark deposits have reported the superiority of ORO,^{105, 138, 139} whereas PD was found to be more effective with natural fingermarks.^{107, 140} In essence, ORO is generally effective in developing a charged fingermark that is less than a few weeks old, although the detection of a 20 year old fingermark has been reported.¹⁴¹. PD is a superior method for detecting older and uncharged fingermarks, that have been exposed to water. The variation in effectiveness has been attributed to ORO targeting the fragile fraction, which is more susceptible to environmental factors. Whereas PD targets the stable robust fraction, hence the ability to detect older fingermarks and on items under prolonged exposure to water. Nonetheless, ORO (particularly the modified formulation) offers an alternative for jurisdictions where PD may not be feasible, subject to recognition of its limitations.

1.4.1.7 Nile red

Nile Red (9-diethylamino-5H-benzo[a]-phenoxazine-5-one) is a benzophenoxazine lipophilic stain that can stain fingermarks for visualisation.¹⁴² It has photoluminescent properties and was initially introduced as a fluorescent stain for intracellular lipid staining in the 1980s.¹⁴³ More recently, Braasch *et al.*¹⁴⁴ proposed it for detecting latent fingermarks on wetted paper. Nile red preferentially dissolves in neutral lipids of the fingermarks and can be visualised under luminescent mode. Due to poor solubility and quenching of fluorescence in aqueous environment, methanol is used to dissolve Nile red.¹⁴²

A recent study¹⁴⁴ compared Nile red to PD and ORO on wetted papers in a pseudooperational study. The study found that Nile red was effective on fingermarks less than 1 month old, but like ORO, it was outperformed by PD on older marks. However, Nile red has the advantage of a simple and easy preparation process and develops luminescent fingermarks with better contrast on dark and patterned surfaces. Unfortunately, the high cost of Nile red (ca. \$760 USD per gram)¹⁴⁵ and the use of toxic reagent methanol limits its operational use. Frick *et al.*¹⁴⁶ developed an alternative method to overcome these issues using the basic phenoxazine dye Nile blue in aqueous solution.

1.4.1.8 Nile blue

Nile blue A or Nile blue (benzo[a]phenoxazin-7-ium,5-amino-9-(diethylamino)-, sulfate) is another benzophenoxazine dye that is commonly used as a histological stain.¹⁴⁷⁻¹⁴⁹ It was initially trialled as a fluorescent latent fingermark detection technique in the 1980s.¹⁵⁰ Other tested forensic applications include a post-cyanoacrylate stain¹⁵¹ and lip print detection,¹⁵² where it is dissolved in an alcoholic solution.

Frick *et al.*^{57, 146} developed an aqueous-based formulation to eliminate the use of organic solvents. Nile blue is a cationic dye, readily dissolving in aqueous solution but exhibiting low levels of luminescence compared to Nile red.¹⁴² In aqueous solution, a small amount of Nile blue undergoes spontaneous hydrolysis to Nile red (Figure 1.5).

The presence of the two stains results in a dual-purpose fingermark detection reagent. Nile red dissolves preferentially in neutral lipids, producing photoluminescence, while Nile blue interacts with the acidic components, resulting in visible dark blue development and suppression of background fluorescence.



Figure 1.5: Hydrolysis of Nile blue to Nile red in aqueous solution.

Studies by Frick *et al.*¹⁴⁶ demonstrated aqueous Nile blue's effectiveness in developing lipid-rich fingermarks on a range of wet porous and non-porous surfaces, including some adhesive surfaces. The advantage of this formulation is that it is water-based, making it a safe and organic solvent-free procedure. Moreover, Nile blue dye is inexpensive (ca. \$3.20 USD per gram)¹⁵³ compared to Nile red. Its low toxicity, cost, and ease of use make it a promising option for jurisdictions with limited resources or for non-scientifically trained personnel. It also underlines an avenue to adapt to challenges encountered in resource-limited jurisdictions through the use of alternatives chemicals and modified reagents.

Further studies by Crocker¹⁵⁴ have shown that modifying the aqueous Nile blue formulation with sulfuric acid can increase the technique's sensitivity. This is because the presence of acid increases the hydrolysis of Nile blue into Nile red, resulting in increased Nile red concentration as demonstrated using luminescence spectroscopy. Similarly, the preliminary study reported that acidified Nile blue produces fingermarks with greater luminescence than aqueous Nile blue on wet porous substrates. Other studies have also investigated the potential of Nile blue as a postcyanoacrylate stain, which is discussed in section 1.4.2.4.

1.4.2 Non-porous-surface fingermark detection methods

When a fingermark is deposited on non-porous substrates like glass or plastics, it remains on the surface and is vulnerable to damage from environmental factors.⁹⁰ Therefore, solvent-based detection methods are not generally preferred as they are likely to wash away the fingermark residue. Instead, it is recommended to use dry powders or cyanoacrylate fuming methods.^{16, 17} However, on challenging surfaces and where the fingermark is partially fixed, such as on adhesive tape, using a liquid powder suspension is more effective.

1.4.2.1 Fingerprint powder

The powdering technique is one of the oldest reported latent fingermark detection methods on smooth non-porous surfaces, and it remains a common method used at crime scenes.¹⁵ Various commercial powders, such as magnetic, granular (carbon black and titanium dioxide white), metal flake (aluminium and brass alloy), and fluorescent powders, are available for different applications. However, some of these powders were developed in the 1920s¹⁵⁵ and appeared not to have been developed within a standardised approach, underlining the need for a revisit under the more current assessment framework.²⁴ This is especially important when considering the method's sustainability.

The development of latent fingermarks with powder occurs based on the preferential adhesion to the fingermark constituents in contrast to the substrate background.¹⁵ Several mechanisms contribute to the overall adherence, including particle shape, size, surface chemistry and electrostatic charge.¹⁵⁶ The presence of grease and liquid in fingermark deposits has also been identified to increase adhesion through increased contact surface area and the liquid surface tension capillary action.⁶⁸ This explains why the powdering technique is not suitable for old, dried fingermarks on porous substrates. Additionally, surfaces with high adhesion to fingermark powders,

such as adhesive tape, result in poor contrast between the fingermark and substrate, presenting a challenge.

1.4.2.2 Powder suspension

Powder suspension is an effective and recommended method for developing latent fingermarks on the sticky side of non-porous tapes.¹⁶ It consists of a powder suspended by a surfactant in water to form a paste that is then lightly painted over the adhesive tape using a brush, followed by washing with tap water. An early process was adopted from Japan and commercialised in the 1990s as "Sticky-side Powder" by the Lightning Powder Company.¹⁵⁷ This formulation, containing grey and black powder in Kodak Photoflo surfactant water mixture, was compared to gentian violet, the primary method used on adhesive tapes in 1996.¹⁵⁸ The sticky-side powder method was found to outperform gentian violet and led to investigations into alternative formulations; using black,¹⁵⁹ grey^{160, 161} and white^{162, 163} commercial fingerprint powders combined with different surfactants.

In the UK, several powder suspension formulations were compared against Sticky-Side Powder to determine their effectiveness on various surfaces. Results showed that a black formulation containing precipitated magnetic iron oxide was more effective in treating light-coloured non-adhesive surfaces.⁶⁸ Further research revealed that its effectiveness is highly influenced by the type of iron (II/III) oxide powder used.¹⁶⁴⁻¹⁶⁶ On dark non-porous adhesive surfaces, a white powder suspension formulation based on titanium dioxide was found to be more effective,⁶⁸ which was supported by other independent studies.^{163, 167}

Meanwhile surfactants such as Liquinox¹⁶², Kodak Photoflo^{160, 161} and Triton[™] X-100¹⁶ were found to be the most effective for the various formulations. However, some surfactants have faced supply chain issues, such as Kodak Photoflo being discontinued due to commercial viability and Triton X-100 being subjected to environmental regulations.¹⁶⁸

The exact mechanism of powder suspension is unknown, but it has been suggested that surfactant micelles encapsulate the powder particles, and certain components

of the fingermark residue disrupt the micelles, exposing the powder and causing selective deposition on the fingermark,¹⁶⁶ illustrated in Figure 1.6. Based on the observation that powder suspension develops fingermarks on wetted surfaces^{169, 170} and does not appear to be selective towards sebaceous fingermarks, it has been suggested that the eccrine component encapsulated by the water-insoluble constituents is responsible for the destabilisation of the surfactant micelles.¹⁶



Figure 1.6: Schematic illustration of powder suspension development. Fingermark constituents disrupt surfactant micelles exposing suspended powder particulates, causing discriminate deposition on fingermark and not the substrate.

Various pre-mixed commercial proprietary black and white powder suspension formulations have become available in recent years,¹⁷¹⁻¹⁷⁴ with carbon-based powder suspension generally recommended as the most effective on non-porous lightcoloured adhesive surfaces. However, their operational use may be limited in small laboratories with restricted budgets due to the high cost of commercial powder suspension. For example, the black powder Wetwop[™] costs \$340 USD per litre.¹⁷¹ Another consideration when using a pre-mixed product is the general lack of control over formulation quality, and shelf-life¹⁶⁸. This has led to interest in developing cheaper in-house alternatives.

A recent example is the development of WET UCIO powder suspension formulation,¹⁷⁵ consisting of carbon black Sirchie powder mixed with a commercial Gran Velada surfactant solution. This formulation has been reported to be more effective than the commercial Wetwop[™] on natural fingermarks deposited on eight

types of adhesive tapes.¹⁷⁵ The WET UCIO powder is estimated to cost c.a \$64 USD per litre, which is at least five times cheaper than Wetwop[™]. The development of more economical reagents increases the accessibility of the method to the Global South.

1.4.2.3 Cyanoacrylate

Cyanoacrylate fuming also known as superglue fuming, is one of the most common methods used on non-porous surfaces such as glass and plastics.¹⁵ This technique was discovered independently in Japan, America and the UK in the late 1970s.^{176, 177} The glue is based on colourless cyanoacrylate esters, specifically methyl-2-cyanoacrylate or ethyl-2-cyanoacrylate, which are monomeric liquids that polymerise into a white solid in the presence of weak bases. Although the exact mechanism for polymerisation growth on fingermarks is not well understood, it has been suggested that anionic polymerisation can be initiated by water and other weak bases in the residue.¹⁷⁸ Polymerisation occurs due to the strong inductive effects of the cyano group (– CN) and the ester group (– COOR), which makes the alkyl cyanoacrylate susceptible to nucleophilic attack. Once nucleophile initiation occurs, it will react with another alkyl cyanoacrylate polymer chain. An example of ethyl cyanoacrylate polymerisation reaction is shown in Figure 1.7.

Initiation:



Polymerisation:





Polycyanocrylate has been observed to occur preferentially on greasy and moist fingermark deposits. Lewis et al.¹⁷⁹ found that moist fingermarks prior to development produce high quality cyanoacrylate fingermarks. However, sebaceous fingermarks produce better quality fingermarks than eccrine when allowed to age. This was suggested to be due to sebaceous material retaining moisture rather than its constituents contributing to the polymerisation reaction. Differences in the morphology of the polymer structure have been observed under scanning electron microscopy.⁶⁸ Sebaceous fingermarks exhibit clumps of noodle-like polymer structure compared to longer microstructure with eccrine marks. Further studies have shown that high relative humidity increases polymerisation, which is thought to be due to rehydration by sodium chloride salts in the fingermark deposit facilitating the initiation process. Paine *et al.*¹⁰³ reported that the highest quality developed fingermarks were obtained at 80 % RH, which aligns with the theory that water is an initiator given that sodium chloride is saturated at 75 % RH,¹⁸⁰ the point at which it draws in water from immediate surroundings. The morphology of polymer structure above 70 % RH is noodle-like compared to flat film-like microstructure at 60 % RH.¹⁰³ Farrugia *et al.*¹⁸¹ suggested the difference in morphology can be attributed to the abundance of water at high RH leading to a soft anion initiation causing one directional polymer growth.¹⁸² At lower RH, initiation by a hard anion results in many active centres of polymer growth.¹⁸³ The noodle-like morphology is favoured as it provides better visualisation due to increasing light scattering interactions with staining dyes.

Initially, the processing method of cyanoacrylate involved a simple improvised fuming cabinet comprising a sealed chamber and drops of cyanoacrylate at the bottom.¹⁸⁴ In an attempt to decrease the processing time and enhance the quality of developed fingermarks, the use of heat¹⁸⁵ to increase the rate of evaporation of the glue and other accelerating agents such as sodium hydroxide¹⁸⁴ and sodium carbonate¹⁸⁶ were investigated. As the literature shed light on the polymerisation mechanism and optimum conditions for fingermark development, dedicated controlled humidity superglue cabinets were developed. However, due to the high cost of commercial cabinets, many police agencies still use improvised fuming

chambers, such as an aquarium and a hotplate as a heat source.⁶⁸ Other adaptations include using a small dish of water to increase humidity or small fans for even development in large chambers.¹⁷ These adaptations demonstrate how jurisdictions with limited resources can access latent fingermark detection methods. However, safety precautions must be taken to avoid health risks, associated with the suggested carcinogenic hazards with heating cyanoacrylate above 200 °C. This underscores the importance of safety whilst adapting fingermark enhancement techniques.

1.4.2.4 Post-cyanoacrylate stains

Developed cyanoacrylate fingermark is a white deposit, and photography can be challenging on light-coloured or multicoloured surfaces. Post-treatment methods such as powdering and staining are employed to enhance visualisation.

Enhancement using staining techniques can be achieved using coloured and fluorescent dyes, with the latter recommended due to higher sensitivity.¹⁷ Rhodamine 6G and Basic Yellow 40 (BY40) are two of the most commonly used stains. Rhodamine 6G was suggested by Menzel *et al.*¹⁵⁰ in the 1980s, and it is one of the most widely used post-cyanoacrylate stains. However, the most effective formulations use methanol, which is toxic, and there is also an ongoing debate on the carcinogenic effect of rhodamine 6G.⁶⁸ As an alternative BY40 was proposed,⁶⁸ which has lower toxicity since it uses ethanol. Nonetheless, the use of alcohol is detrimental to surfaces such as varnishes and some ink-printed surfaces. A waterbased formulation of BY40 has been explored but exhibited lower effectiveness. This has led to the investigation of other dyes that can be used with a water-based formulation, such as basic red 14 and Nile blue. Nile blue has additional benefits being an inexpensive dye that is water soluble. Although preliminary study by Chesher *et al.*¹⁵¹ reported weak development with Nile blue as a post-treatment stain for cyanoacrylate, there has been limited research on its potential in the literature. More recently a series of preliminary studies^{154, 187, 188} at Curtin University have shown promise for Nile blue as a safe and cost-effective post-cyanoacrylate enhancement technique.

Furthermore, the mechanism of post-cyanoacrylate staining is not well understood. One of the few suggestions is that the cyanide (CN⁻) anions of polycyanoacrylate may form weak van der Waals interactions with basic dye cations.⁶⁸ A fundamental understanding of the staining process can direct the use of new dyes and sequencing processes.

1.5 Latent fingermark detection service provision

The detection of latent fingermarks is an important tool in both humanitarian and criminal investigations worldwide^{26, 189, 190} Therefore, it is essential for jurisdictions to have access to effective latent fingermark detection techniques and the ability to ensure the continuity of service. The previous section highlighted some of the obstacles that resource-limited jurisdictions face in implementing fingermark enhancement methods, as well as potential solutions. The next section examines the specific case of Seychelles, a remote, small jurisdiction, before expanding the focus to a global perspective. Amid the discussion, the concept of sustainable forensic science service provision is presented as a viable solution.

1.5.1 Seychelles latent fingermark detection service provision

Seychelles is an archipelago of 115 islands in the West Indian Ocean covering a total land mass of 444 square kilometres, spread over a vast sea area of 1.4 million square kilometres (Figure 1.8). Because it is located near the equator, Seychelles has a tropical climate with temperatures ranging from 21 °C to 23 °C and relative humidity between 70% and 80%.¹⁹¹ It is the smallest African state with a population of approximately 100,000 and a fragile economy primarily based on tourism, contributing 70 % of the gross domestic product (GDP).¹⁹² Relative to population size, the Seychelles Forensic Section has modest resources and trained local staff within a unified forensic science service, which operates on a hybrid of local service provision and overseas outsourcing under the auspices of the Seychelles Police. The potential socio-economic impact of crime on a small island community and fragile tourism

economy has led the Government to invest in strengthening forensic science service provision,¹⁹³ including latent fingermark detection and identification capabilities.



Figure 1.8: The geographical location of Seychelles.

Established in 1968, the Seychelles latent fingermark section was the first local forensic science service provision.¹⁹⁴ Its administration and delivery system are primarily based on the UK model, as Seychelles was a former British colony. This is evident with most of the personnel trained in the UK and using the CAST Fingermark Visualisation Manual¹⁶ as the basis for applying fingermark detection methods despite major differences in climatic conditions. In 2013, the Seychelles Police Force (SPF) enhanced its fingermark identification capabilities by acquiring an automated fingerprint identification system under the framework of the Interpol capacity building for Eastern African countries funded by the European Union (EU).¹⁹⁵ However, as a prerequisite, the system's effectiveness depends on an enhanced latent fingermark detection capability.

1.5.2 Challenges in operationalising latent fingermark detection protocols

When implementing new or modified latent fingermark detection protocols in Seychelles, there are two primary considerations due to its locality: i) the limited budgetary and technical resources and ii) supply chain interruptions. As a small jurisdiction, Seychelles has limited resources, making it essential to adopt an economical, safe, and simple latent fingermark detection method to optimise resource management. Secondly, Seychelles is geographically remote as shown in Figure 1.8, and sits at the end of the commercial supply chain. As a result this increases the risk of supply chain disruption.¹⁹⁶ Therefore ensuring a readily available and easily accessible detection method is crucial to service continuity.

Over the last two decades, a significant amount of research has been into developing novel and improving existing latent fingermark detection methods.^{26, 125, 197, 198} Overwhelmingly, this research has been carried out in the northern hemisphere and highly developed nations with abundant resources and non-tropical climates. The primary focus of these studies has been to improve performance of detecting quality and quantity of fingermarks, but not necessarily to address challenges faced by jurisdictions with limited resources and supply chain issues. Therefore, the operationalisation of these methods can be problematic in a small jurisdiction as illustrated in Figure 1.9. To overcome these challenges and provide sustainable fingermark service provision, smaller jurisdictions require cost-effective and resilient detection methods.



Figure 1.9: Illustrating the gap in operationalisation of latent fingermark detection technique in remote and small jurisdictions.

As an example, the carrier solvent for the amino acid sensitive reagents has been standardised over the last two decades to HFE-7100 (methyl nonafluorobutyl ether).^{120, 199} Despite its ideal properties as a carrier solvent, HFE-7100 is expensive

(ca. \$240 USD per litre) and prone to restricted supply^{168, 200} due to environmental regulations. The supply risk is exacerbated as it is a single-source proprietary solvent from 3M[™] Novec^{™201} and under demand from multiple industries. Manufacturing plant issues and the COVID-19 pandemic have increased the lead time for this product in large jurisdictions like Australia and the United States by up to 9 months over the last 2 years,^{202, 203} underscoring the global extent of supply chain disruption. It follows from the perspective of a restricted-budget laboratory in a remote location that the reliance on HFE-7100 poses a substantial risk to service continuity.

This signifies a gap in developing and validating latent fingermark detection techniques for operational use. The question that arises is how small jurisdictions such as Seychelles can adapt to access and develop fit-for-purpose fingermark detection capabilities and maintain continuity of service. In other words, how to establish a sustainable provision for latent fingermark detection.

1.5.3 Sustainable forensic science provision; frugal forensics

The challenges that Seychelles faces in the field of latent fingermark detection reflect the broader issue of limited forensic science resources in Global South jurisdictions. A Scopus bibliometric search of articles using "forensic science" as a search term reveals that the space is dominated by literature emanating from the Global North (Figure 1.10). A similar conclusion may be drawn from a more detailed recent study²⁷ examining research trends in forensic fingermark detection and interpretation.

The research dominance, resourcing, and advanced capabilities of the Global North may convey the view that forensic science provision is globally well-developed. However, the stark reality is that most Global South nations face substantial obstacles in forensic science service provision compared to the Global North. For example, the literature reveals that challenges of the Global North revolve around high-level issues such as forensic backlogs,^{204, 205} human bias or error influencing decision-making,²⁰⁶⁻²⁰⁸ and the need for evaluative and activity-level reporting.^{209, 210} By contrast, the Global South faces fundamental resourcing issues such as a lack of fingerprint or DNA databases,^{211, 212} inadequate training,^{213, 214} and supply chain

problems.^{196, 215} Hence it is not surprising that the limited forensic research literature from the Global South primarily focuses on the challenges around developing and strengthening fundamental forensic science capability and service provision. Against this backdrop, it is important to note that latent fingermark service provision is common across all jurisdictions and is generally a primary forensic science service in the Global South.¹¹



Figure 1.10: Number of publications on "forensic science" in a simple Bibliometric search on Scopus for the (a) Total papers 1972–2022 (b) Top 10 countries and (c) Top 10.

This thesis proposes an approach to how Seychelles and, to some extent, broader Global South jurisdictions can adapt to the challenges of their locality and develop sustainable fingermark detection provisions. This can be achieved by recognising vulnerabilities within the process and building resilient capacity to deliver innovative, simple, and economical services meeting fit-for-purpose quality benchmarks. For example, in the case of Seychelles investigating the use of more economical chemical analogues of fingermark reagents, or sourcing of more accessible or locally available alternatives would provide a more resilient fingermark detection method. The investigation should be carried out in a standardised framework to ensure the quality of the results. This approach follows a concept referred to in this thesis as frugal forensics, defined as the development of resilient and economical forensic science provision that meets the needs of society without compromising quality and safety. This concept is derived from a combination of frugal innovation²¹⁶ and the Brundtland Commission's definition of sustainable development.²¹⁷ Frugal forensics is based on three core principles; Resilience, Economics and Quality, and six attributes; Performance, Accessibility, Availability, Cost, Simplicity and Safety (PAACSS) (Figure 1.11), which underpin the sustainable provision of forensic science services. At the heart of this approach is shifting the focus from 'pure' performance to a holistic consideration of jurisdictional vulnerabilities while ensuring that risks or limitations are recognised and documented to ensure transparent, high-quality service provision.



Figure 1.11: The concept of sustainable forensic science provision (a) Definition (b) The three principles and six attributes that underpin it.

Although this approach focuses on fingermark detection, the considerations raised in relation to economics, resilience and quality also apply to the sustainability of the broader forensic science service provision. While it may not have been explicitly defined previously, sustainable forensic science provision is not a new concept, as by nature and necessity, it is embraced in Global South countries due to their unique needs. However, its development thus far has been piecemeal and ad hoc. The challenge is for it to be systematically integrated and aligned to international standards as far as can be practically achieved. For latent fingermark detection, the IFRG guidelines²⁴ provide an objective evaluation and reporting framework for
performance assessment of new or modified methods. The guidelines specifically recommend testing methods in the environment where they are to be operationally used, aligning with the concept of frugal forensics as 'meeting the needs' of the locality. The recommendation for local validation extends this approach a step further by evaluating not only the method's performance but also its sustainability in the light of local challenges.

Embracing this concept has the potential to contribute towards the United Nations Sustainable Development Goals (SDGs).⁹ These goals consisting of 17 intertwined targets, aiming to build a set of shared global targets for a just and equitable world. The 'SDG 16 –Peace and Justice' goal implies that forensic science service provision is essential in achieving sustainable development. Developing a sustainable latent fingermark detection protocol under the banner of frugal forensics (Figure 1.12) increases accessibility to forensic science service provision in jurisdictions where such provision would have been limited or otherwise absent. At the core of the SDGs is the drive to reduce inequalities between nations and to transfer resources, knowledge, technology, and capability to the Global South. The collaboration of Curtin University and the Seychelles Police Forensic Section through this thesis is a prime example.



Figure 1.12: Illustrating the relationship between frugal forensics and UN sustainable goals through SDG 16.²¹⁸

1.6 Aim and Objectives

This study focuses on enhancing the accessibility of latent fingermark detection techniques and enabling their use in remote and resource-limited jurisdictions. The research is conducted within the frugal forensics framework, which recognises the constraints of these jurisdictions to develop adaptable and fit-for-purpose detection techniques. This concept can be used to develop sustainable latent fingermark protocols, increasing access to fingermark detection capabilities in jurisdictions where they were previously unavailable or restricted.

The study aims to achieve four main objectives:

- Develop a sustainable IND/Zn formulation to facilitate its use in the Global South jurisdictions as a recognised most sensitive amino acid sensitive reagent for detection of fingermarks on paper substrates (Chapters 3 and 4).
- Modify a recently published carbon-based powder suspension formulation for non-porous surfaces to address the accessibility issue of commercial surfactant solutions (Chapter 5).
- Investigate Nile blue as an emerging, low-cost, and easily accessible method that can be used as a lipid-sensitive reagent and post-cyanoacrylate-fuming stain (Chapter 6).
- Develop a comprehensive evaluation tool to assess the sustainability of latent fingermark detection methods (Chapter 7).

This thesis adheres to the IFRG guidelines, which provide best practice guidelines for evaluating new or modified fingermark detection methods. These guidelines are essential due to the intrinsic high variation of fingermarks, and it's important to account for multiple donors, ageing conditions, and result assessment. Additionally, it fulfils quality assurance requirements to facilitate validation and its adoption of detection methods in operational laboratories. Chapter 2 outlines the experimental considerations in line with the IFRG guidelines, followed by an overview of the experimental methods and instrumentation used throughout this study. Chapters 3 and 4 investigated how to enhance the accessibility of existing latent fingermark detection methods for the Global South jurisdictions. Chapter 3 begins with a preliminary head-to-head comparison of three active IND/Zn formulations to identify the most suitable formulation that can achieve sustainability with minimum modifications. Subsequently the different chemical components of the formulation were assessed in the context of a frugal forensics approach. This provided direction in terms of methodology and modification of the formulation to come to a final formulation that was assessed through laboratory trials in Chapter 4, including the use of incidental fingermarks to provide a more realistic casework sample. This will provide a formulation that is more accessible to other jurisdictions and enhance the recoverability of fingermarks.

Chapter 5 investigates the use of sodium dodecyl sulfate salts to prepare an in-house surfactant solution to replace the commercial Gran Velada solution used in a recently published carbon-based WET UCIO powder suspension formulation. The study is divided into two parts; the first involves characterising pressure-sensitive tapes using ATR-FTIR analysis. The second part consists of modifying the WET UCIO formulation and assessing it in comparison to the original and commercial powder suspension formulations. This will improve the accessibility of the low-cost WET UCIO formulation to other jurisdictions.

Chapter 6 investigates the benzophenoxazine dye Nile blue as an emerging lipidsensitive latent fingermark detection technique. Firstly, its effectiveness as a lipidsensitive method was compared to ORO on wet porous surfaces. Secondly, its potential as a post-cyanoacrylate stain was investigated by comparing it with BY40. Additionally, the staining mechanism was explored using confocal Raman microscopy. This gives an insight into the use of acid Nile blue as a frugal forensics method and a fundamental understanding of its staining mechanisms that can assist in developing post-cyanoacrylate dyes.

Chapter 7 addresses how the methods investigated in Chapters 3-6 can be objectively evaluated within the frugal forensics framework. This involves developing an assessment tool for a holistic evaluation of the frugal forensics attributes in consideration of the jurisdiction-specific challenges. This tool can assist with decision-making and identify areas that need modification to achieve sustainability. Moreover, it provides a transparent and objective evaluation for end-users of forensic information and aligns with ensuring quality assurance in fingermark service provision.

Chapter 2. Sample collection, experimental methods, and instrumentation

2.1 Introduction

This chapter outlines the experimental methods used in the study. It begins with an overview of the methodology and experimental considerations, followed by discussion of the fingermark collection procedures and an outline of the fingermark development methods. Finally, it provides details of the instrumentation, fingermark grading system and statistical analysis used in this study.

2.2 Experimental considerations

2.2.1 IFRG guidelines

Fingerprints are intrinsically highly variable, and the methods used to enhance them vary greatly, making it necessary to have a standardised approach to compare new or modified enhancement techniques. A standardised approach was highlighted by Kent in 2010²¹⁹ and Sears *et al.* in 2012¹⁹ due to inconsistencies in the methodology and reporting of fingermark detection research. In response, the International Fingerprint Research Group (IFRG) published guidelines²⁴ in 2014, providing a standardised research framework for assessing detection techniques from the initial concept to the final casework implementation. This thesis adheres to IFRG guidelines with deviations clearly stated and the reason for the deviation.

The framework divides the research process into four phases: initial proof-of-concept studies (phase 1), optimisation and comparative assessment with an established method (phase 2), validation using large numbers and more realistic samples (phase 3) and operational evaluation (phase 4). As a method progresses through different phases, the research methodology changes to manage the number of samples while obtaining meaningful results. For example, phase 1 uses a smaller sample size, while phases 3 and 4 have more rigorous evaluations using realistic samples. The different experiment in this study covers phase 1 to phase 3 and broadly consist of two strands. Firstly, method development and preliminary comparative study to improve relevant parameters in line with the frugal forensics approach. Secondly, if the modified method is deemed viable, it was then compared to existing techniques through

laboratory trials on incidental fingermarks or simulated casework substrates to assess its operational suitability.

2.2.2 Frugal forensics approach

The method progression also depends on the resources at the disposal of researchers and the operational laboratories.²⁴ The aim of this thesis is to address the challenges posed by the limited availability and accessibility of resources in some jurisdictions, which hinder the use of effective latent fingermark detection methods. Therefore, using minimal equipment for operational simplicity, cost-effectiveness, and health and safety are underlying considerations.

2.2.3 COVID-19 pandemic impact

Due to part of the project coinciding with the COVID-19 pandemic, consideration had to be taken given the limited availability and access to donors. The number of donors was kept at a minimum but within the recommendation of the IFRG guidelines. Additionally, laboratory trials intended to be conducted in Seychelles were scaled back to Western Australia (WA) due to travel restrictions in Western Australia and the Seychelles that lasted until mid-2022.

2.3 Fingermark methodology

The standardised methodology limits fingermark variability by addressing key variables such as the number of donors, collection and storage procedures, substrates and environmental conditions, development and visualisation conditions, and analysis and reporting. As a result, experiments are designed to reduce the number of uncontrollable variables as well as account for the intrinsic variability of fingermarks.^{19, 24, 219}

2.3.1 Donor information

To ensure accurate results fingermark reagent testing is carried out on a range of donors across both biological sexes and covering a wide range of ages. A total of seventeen donors participated in this study, nine females and eight males aged 22-

60 years. The IFRG recommends using 5-15 donors for phase 2 and \geq 20 donors for phase 3. Due to COVID-19, the study used five donors for the initial study and ten for laboratory trials. However, the number of samples for the laboratory trials was increased, as studies have shown that meaningful results can be obtained either in the form of a large donor pool or a large number of marks per donor.²²⁰

2.3.2 Fingermark collection procedures

Fingermark donors were instructed to lightly rub their hands together and provide natural fingermarks by lightly pressing the finger on the substrate surface for ~5 seconds. Care was taken to ensure that the donor had not washed their hands, eaten or come into contact with chemicals in the 30 minutes prior to sample collection to reduce contamination and ensure a natural build-up of surface secretions. Multiple samples collected from the same fingermark donor were collected at least with 30 minutes intervals between depositions of each sample to allow secretions to replenish on donors' fingertips.

Natural fingermarks, which have not been intentionally groomed for increased material, were used throughout this study, except in Chapter 6, where charged fingermarks were used to investigate the selectivity and donor effect of the lipid-sensitive acid Nile blue method. For charged fingermarks, donors were instructed to wipe their fingers across the face or forehead prior to the deposition of marks.

In casework, fingermarks are not deliberately deposited; incidental fingermarks provide a more realistic sample in assessing fingermark detection method in line with phase 3 of the IFRG guidelines. Incidental fingermarks deposited through the process of inspecting the contents of an envelope, simulating activity in cases of theft, were used in Chapter 4. When charged or incidental fingermarks are used, it is clearly outlined in the relevant chapters.

Split depletion series fingermark collection method was used in this study unless stated otherwise. When comparing the effectiveness of two fingermark development techniques, the split depletion method is preferred for fingermark collection. It involves depositing a series of fingermarks, dividing it in half, and processing each half with a different technique before recombining for assessment. This allows for a direct comparison of the two techniques within a single fingermark, which reduces variability in the amount and composition of the deposited fingermark material.^{19, 24} The use of depletion series allows for testing for sensitivity, with progressively less material available for detection with successive deposition (Figure 2.1). To minimise donor biasing on one side of the finger, a second depletion series was collected from the same donor in order to reverse the development process on the opposite side.



Figure 2.1: Schematic illustrating split fingermark and depletion series methodology for fingermark collection and the reverse of development process A and B to minimise donor biasing on one side of the finger.

An additional technique that can be employed to quickly evaluate a formulation or make small adjustments is the quartered fingermark approach.¹⁹ This method entails

placing a single natural fingermark at the centre of a substrate that is sectioned into four equal parts. Subsequently, the fingermark is divided into four parts, with each quarter treated with different formulations or the same formulation that varies by only one component, such as the processing time. Quartered fingermarks methodology was used in Chapters 4 and 6.

2.3.3 Fingermark ageing and storage conditions

Fingermarks samples were stored under ambient conditions in an office cupboard (ca. 20 °C, 56 %RH) for at least 24 hours prior to development unless stated otherwise. To study the effects of sample age, samples were stored under the same conditions for 7 and 30 days. These periods were chosen because fingermark composition undergoes the most changes within the first 24 hours²¹, and the typical time frame casework samples are likely to be processed. Other storage conditions including samples stored under non-airconditioned environment to simulate the condition in Seychelles and samples submerged in water to simulate wet surfaces, was used in Chapter 4 and 6 respectively. The details of each condition are provided under the relevant chapters.

2.4 Substrate selection and environmental considerations

2.4.1 Substrate surface selection

The IFRG guidelines²⁴ recommends testing under local conditions and substrates. Since the experiments were conducted in WA, every effort was made to source representative samples encountered in casework in Seychelles. This includes cardboard, a common surface encountered by the Seychelles Police. However, most substrates encountered in Seychelles are imported from multiple overseas sources, creating a diverse and dynamic situation. The substrates used in this study are specified in the experimental section in Chapters 3-6.

2.4.2 Environmental considerations

As part of this study, it was intended that the final laboratory trials be conducted in Seychelles; however, due to COVID-19 travel restrictions the experiments had to be

conducted in WA. Considerations were made regarding the difference in temperature and humidity between the two locations, which may affect IND/Zn efficacy as outlined in Chapter 4. A preliminary experiment under elevated humidity and temperature to mimic the tropical conditions of Seychelles was conducted to evaluate the suitability of IND/Zn formulation under such conditions.

2.5 Fingermark development

In this study, several fingermark detection methods in line with some of the current capabilities (Table 2.1) of Seychelles Police were investigated. Chapters 3 and 4 investigated IND/Zn formulations, Chapter 5, carbon-based powder suspension formulations and Chapter 6, acid Nile blue, ORO, cyanoacrylate and BY40. The following section outlines the details of each development procedure, including reagent information, solution preparation and development methodology.

Detection method	Reagents	Equipment
Amino acid	Ninhydrin	Treatment: Spray bottle and extractor
		Development: Overnight using ambient temperature
Powder suspension	Wetwop black and Wetwop white	Treatment: Brush, wash bottle, glass tray
Superglue fuming	Cyanoacrylate	Treatment: Cyanoacrylate fuming chamber
Superglue fluorescent dye staining	Basic yellow	Treatment: Wash bottles and glass trays Visualisation: Polilight, Nikon DSLR camera

Table 2.1 Summary of some of the chemical detection capabilities of the Seychelles Police.

2.5.1 1,2-indanedione zinc

1,2-Indanedione is an amino acid sensitive fingermark reagent commonly utilised to develop fingermarks on porous surfaces such as paper. In this study, three operationally active IND-Zn formulations were used, published by i) the National Centre for Forensic Studies (NCFS),²²¹ in use in Australia, ii) Bundeskriminalamt

(BKA),²²² in use in Germany, and iii) the Centre of Applied Science and Technology (CAST),¹²⁷ in use in the UK.

Materials

1,2-Indanedione (Reddy Chemtech), ethyl acetate (\geq 99.7 %, UNIVAR - APS), glacial acetic acid (\geq 99.7 %, Lab-Scan), methanol (\geq 99.7 %, Honeywell – Burdick & Jackson), ethanol (\geq 99.5 %, UNIVAR - Ajax Finechem), zinc chloride (\geq 98 % Sigma Aldrich), HFE-7100 (3M Novec), petroleum ether 40-60 °C (VWR Chemicals), Solstice[®] PF (Honeywell). Reagents were used as received without further purification.

Sample preparation

NCFS formulation was prepared as outlined by Stoilovic *et al.*,²²¹ 0.598 g of 1,2indanedione dissolved in 124.8 mL of ethyl acetate and 5.2 mL of glacial acetic acid. 870 mL of carrier solvent (HFE-7100 or Solstice[®] PF) was added, followed by 4 mL of zinc chloride stock solution (8 g zinc chloride in 200 mL of ethanol).

BKA formulation was prepared as outlined by Becker *et al.*,²²² 1.0 g of 1,2indanedione dissolved in 45 mL of ethyl acetate and 10 mL of glacial acetic acid. 900 mL of carrier solvent (petroleum ether 40-60°C) was added, followed by 10 mL of zinc chloride stock solution (200 mg zinc chloride in 30 mL of ethanol).

CAST formulation was prepared as outlined by Nicolasora *et al.*,¹²⁷ 0.250 g of 1,2indanedione dissolved in 45 mL of ethyl acetate, 10 mL of glacial acetic acid and 45 mL of methanol. 1000 mL of carrier solvent (HFE-7100 or Solstice[®] PF) was added, followed by 1 mL of zinc chloride stock solution (100 mg zinc chloride in 4 mL of ethyl acetate and 1 mL acetic acid).

The developed SPF IND/Zn formulation was prepared using 0.60 g of 1,2-indanedione dissolved in 45 mL of ethyl acetate and 10 mL of glacial acetic acid. 900 mL of carrier solvent (petroleum ether 40-60 °C) was added, followed by 10 mL of zinc chloride stock solution (200 mg zinc chloride in 30 mL of ethanol).

All formulations of 1,2-indanedione zinc were stored in an amber glass bottle away from sunlight.

Sample development

The samples were developed by submerging them in IND/Zn working solution (~150) in a glass tray for 5-10 seconds, then air dried in a fume hood. Fresh working solution was used after every ten samples treatment or when the solution turned cloudy. All treated dry samples were processed in a heat press at 160 °C for 10 s in line with the NCFS and BKA methods. This processing condition was chosen as a simple and cost-effective method that aligns with frugal forensic approach compared to oven processing at 100 °C, which CAST recommends. The SPF method follows the same sample development as the BKA and NCFS methods. All treated samples were photographed and stored in a laboratory cabinet to limit light exposure.

2.5.2 Powder suspension

Powder suspension is a method that can be used to develop latent fingermarks on adhesive surfaces such as the sticky side of pressure sensitive adhesive (PSA) tapes. In this study, various powder suspension formulations were used; two commercial pre-mixed formulations Wetwop[™] black and Wet Powder black, and a recently published¹⁷⁵ WET UCIO formulation including modified versions of it outlined in Chapter 5.

Materials and sample preparation

Wet UCIO formulation was prepared as outlined by Claveria,¹⁷⁵ 1.5 g of carbon black powder (Sirchie) was mixed with 10 mL of 27 % w/v sodium dodecyl sulfate solution (Gran Velada) to make a paste similar in texture to thin paint.

Modified WET UCIO powder suspension formulations were prepared by substituting the Gran Velda solution with in-house surfactant solutions of various SDS concentrations and additives. SDS salt (laboratory grade, Chem Supply), absolute ethanol (analytical grade, Scharlab), polyethylene-400 (laboratory grade, Chem Supply) and deionised water were used to prepare in-house SDS surfactant solutions using volumetric glassware. The composition of the various modified Wet UCIO powder suspension formulations and the in-house surfactant solutions utilised in this study are specified in Chapter 5. Commercial powder suspension products Wetwop[™] Black (Lightning Powder Company) and Wet Powder Black (Kjell Carlsson Innovation) were manually agitated for 10 seconds before use. A working solution of approximately 10 mL was poured into a clean beaker.

Sample development

The application of all powder suspensions followed the same methodology. A separate clean squirrel brush was used for each powder suspension formulation to paint the paste onto the sticky side of adhesive tapes and left for 10-15 seconds. The tapes were rinsed with cold tap water, air dried, and photographed.

2.5.3 Oil red O

ORO is a lipid-sensitive reagent for development of fingermarks on porous surfaces. In this study, a modified ORO formulation reported by Frick *et al.*,¹³⁷ was used.

Materials and sample preparation

ORO was prepared as outlined by Frick *et al.*,¹³⁷ 0.50 g of ORO (Agros Organic, ChemSupply Australia) was dissolved in 1000 mL of propylene glycol (Laboratory grade, ChemSupply Australia) and heated to 95 °C with constant stirring until fully dissolved. The solution was cooled to room temperature and stored in an amber glass reagent bottle in laboratory cabinet.

Sample development

The samples were developed by submerging in ORO solution (~150 mL) in a glass tray for 15 minutes with manual agitation for 30 seconds at the beginning of the treatment. The samples were rinsed with deionised water, air dried on paper towels, and photographed.

2.5.4 Cyanoacrylate fuming

Cyanoacrylate fuming of latent fingermarks was conducted using an improvised nontemperature and humidity controlled chamber used in previous work by McGann *et al.*¹⁸⁷ and Boseley *et al.*¹⁸⁸ at Curtin University. The available laboratory conditions have demonstrated appropriate development on non-porous surfaces following the procedure outlined below.

Materials, sample preparation and development

Cyanoacrylate fumed print was prepared by placing samples in an improvised chamber (Ikea Socker Greenhouse, $35 \text{cm} \times 45 \text{ cm} \times 22 \text{ cm}$). Three drops (~1.5 g) of Loctite 401 instant adhesive ($\geq 99.5 \%$ w/w ethyl-2-cyanoacrylate, Henkel Australia; Blackwoods) contained in small piece of aluminium foil (Capri Heavy Duty Catering Foil, China) was placed in the middle of the chamber. The chamber was sealed with tape and the samples were allowed to develop for 2 hours (no fan). Following development, the samples were removed and allowed to cure overnight inside a laboratory cabinet.

2.5.5 Acid Nile Blue

In this study, acid Nile Blue was used as a lipid-sensitive reagent for the development of fingermarks on wet porous substrates as well as a post-cyanoacrylate stain. The treatment of porous surfaces and post-cyanoacrylate fumed prints follow the similar procedure as outlined below.

Materials and sample preparation

Acid Nile was prepared as outlined by Crocker *et al.*,¹⁵⁴ 0.05 g (6.8 μ mol) of Nile blue (analytical grade, Sigma-Aldrich, USA) was dissolved in 1000 mL of 0.3 M sulfuric acid (analytical grade, Ajax Finechem) and heated to 95 °C for 1 hour with constant stirring. The solution was cooled to room temperature and stored in an aluminium-wrapped glass reagent bottle in laboratory cabinet to avoid photo degradation.

Sample development

The samples were developed by submerging in acid Nile blue solution (~150 mL) in a glass tray for 20 minutes with agitation of the solution across the samples. The samples were rinsed with deionised water, air dried on paper towels, and photographed. All treated samples were stored in a laboratory cabinet to limit light exposure.

2.5.6 Basic Yellow 40

Basic Yellow 40 is a post-cyanoacrylate stain and was used in this study for comparative assessment with acid Nile blue.

Materials and sample preparation

BY40 was prepared as outlined by Bandey *et al.*,¹⁶ 1.0 g of BY40 (ChemSupply Australia) was dissolved in 500 mL ethanol (\geq 99.5 %, UNIVAR - Ajax Finechem). The solution stored in an aluminium-wrapped glass reagent bottle in laboratory cabinet.

Sample development

The samples were developed by submerging in BY40 solution (~100 mL) in a glass tray for 10-15 seconds. The samples were rinsed with deionised water, air dried on paper towels, and photographed.

2.6 Instrumentation

2.6.1 Photography of developed fingermarks

Developed fingermarks were photographed using a Nikon D300 camera on manual exposure mode, mounted on a Firenze Mini Repro stand at 35 cm above the samples and fitted with a 60 mm lens. Samples were illuminated using a Polilight PL500 (Rofin Australia Pty. Ltd., Australia) or dual incandescent light globes (Mirabella). For luminescence examination a long-pass barrier filter with 1% transmittance was fitted on the camera lens to block the reflected light and allow the emitted longer wavelength to pass through as illustrated in Figure 2.2. The camera settings, illumination and barrier filter specification optimised for each development method are shown in Table 2.2. Camera aperture f/11 and ISO200 were consistent across all methods. All digital images were recorded using Nikon Camera Control Pro (version 2.31.0).



Figure 2.2: Schematic of photographic setup for photoluminescence fingermarks examination.

Detection Treatment	Mode	Illumination	Camera filter	Shutter speed
1,2-indanedione zinc	Photoluminescence	Polilight 505 nm	550 nm	1/2 sec
Powder suspention	Absorbance	2 incandescent globes	None	1/10 sec
Cyanoacrylate fuming	Diffused reflectance	2 incandescent globes	None	1/10 sec
Acid Nile Blue	Photoluminescence	Polilight 505 nm	550 nm	1/2 sec
Basic yellow 40	Photoluminescence	Polilight 450 nm	495 nm	1/2 sec

Table 2.2: Photographic conditions optimised to each detection method.

2.6.2 Fluorescence spectroscopy

Fluorescence spectroscopy was used in Chapter 3 to investigate the observable variation in luminescence intensity with IND/Zn due to variation in 1,2-indanedione concentration. The measurements were conducted on amino acid spots containing

alanine, leucine and phenylalanine. The amino acids phenylalanine and leucine were used as representative for aromatic and aliphatic amino acids, and alanine is commonly used as a model for amino acids due to its faster reaction rate and fewer side reactions compared to glycine.^{104, 223}

Amino acid solution was prepared by dissolving 200 mg of L-alanine (Sigma Aldrich, 98% w/w) in 20 mL of MilliQ water, and the solution was mixed thoroughly. The procedure was repeated with L-phenylalanine and L-leucine (Sigma Aldrich, 98% w/w), and then 5 mL from each solution was combined to produce a mixture of amino acid solution. From the mixture of 10 mg/mL amino acid concentrations, series dilution was used to produce solutions of 5 mg/mL, 1 mg/mL and 0.5 mg/mL.

The substrate samples were prepared in strips, each with five spots in the form of circles (5 mm radius) spaced 20 mm apart, drawn with a pencil. Each spot was impregnated with 1 μ L aliquots of the amino acid solution using a micropipette. The samples were stored and treated similarly to other fingermark samples. Each strip was treated with BKA IND/Zn formulation containing different concentration of 1,2-indanedone. One strip was also treated with a formulation of IND/ZN without 1,2-indanedione zinc. This was used as control samples to verify for contamination that may cause photoluminescence.

Fluorescence spectra were obtained using a Cary Eclipse Fluorescence Spectrophotometer (Agilent) combined with a fibre optic probe for *in situ* measurements. The extended probe tip was placed on the substrate ensuring a consistent distance between the probe to the amino acid spot during measurements (Figure 2.3). The excitation wavelength was set at 505 nm, and emission spectra were collected between 540 and 650 nm with a slit width of 5 nm. A background spectrum of each untreated substrate was collected prior to sample analysis. The data acquisition was carried out using the Agilent Cary Eclipse Scan Application (Version: 1.2(147)) and exported as C.S.V file for analysis in Excel.



Figure 2.3: Experimental setup using the fibre optic probe for fluorescence measurements of amino acid spots treated with IND/Zn formulation containing different 1,2-indanedione concentrations.

2.6.3 UV-vis spectroscopy

Visible spectroscopy was used in Chapter 4 to assess the stability of the SPF IND/Zn reagent by monitoring the 1,2-indanedione peak absorbance over time. Spectral scans of the SPF IND/Zn working solutions were performed using an Agilent Cary 60 UV-Vis Spectrophotometer. An Agilent rectangular quartz cuvette (10 mm pathlength) was used, and sample was scanned in the visible wavelength region (400–700 nm) with a scan interval of 0.5 nm and scan speed of 300 nm/s. A corresponding reference blank contained solvents as the sample but did not contain 1,2-indandione, was run as a blank-corrected baseline. The data acquisition was carried out using the Agilent Cary WinUV Scan Application (Version 2.00) and exported as C.S.V file for analysis in Excel.

2.6.4 ATR-FTIR spectroscopy

Attenuated total refection - Fourier transform infrared (ATR-FTIR) spectroscopy was used to analyse the composition of the backing and adhesive of PSA tapes in Chapter 5. A Thermo Scientific Nicolet iS50 Fourier Transform Infrared (FTIR) spectrophotometer with a single-bounce attenuated total reflectance (ATR) diamond crystal was used to obtain 4 cm⁻¹ resolution spectra in absorbance mode over a range of 4000 – 400 cm⁻¹, with 64 accumulated scans. For each tape, a ca. 5 cm strip was cut and placed over the crystal, and spectral analysis was performed using the pressure arm to maintain a consistent contact pressure. The backing and the adhesive side were analysed using separate strips with measurements taken at three locations for better representative composition. The crystal was cleaned with ethanol between different tape sample measurements, followed by a background scan. ATR correction was applied to each spectrum using the in-built software (Omnic 9, Thermo Fisher Scientific), with the replicate spectra then averaged and exported as C.S.V files for analysis.

2.6.5 Confocal Raman microscopy

To investigate the post-cyanoacrylate staining mechanism in Chapter 6, Confocal Raman microscopy was used to investigate the position of the stain in relation to the cyanoacrylate polymer. All samples were prepared on 1 mm borosilicate microscope slides (VWR International, Leuven). Dye samples were prepared by placing one drop of each dye onto a microscope slide. For the fingermark samples, charged split fingermarks were deposited on two microscope slides secured side by side with cellulose tape. Three types of fingermark samples were prepared: unstained cyanoacrylate fumed, acid Nile blue post-cyanoacrylate stain and BY40 post-cyanoacrylate stain fingermarks. All fingermarks were developed according to the methods outlined in sections 2.5.4 to 2.5.6. Care was taken to ensure the fingermarks were intact along the physical boundary between the two slides during fingermark development. After treatment the slides were carefully separated to examine the cross-section of the fingermarks using Confocal Raman Microscopy.

Confocal Raman microscopy studies were conducted using a confocal Raman microscope (alpha 300R, WITec, Ulm, Germany) equipped with a 532 nm frequency doubled Nd:YAG laser and a thermoelectrically cooled charge-coupled device (CCD) detector. The laser light was coupled into the microscope using a single-mode fibre and brought on to the sample using a dichroic mirror and a 100 × microscope objective (NA 0.9). The spatial resolution is about 300 nm with a spectral resolution of 0.02 cm⁻¹. The sample was mounted onto a piezo driven scanning stage with a position accuracy x and y directions. The WITec Control FOUR software was used for data acquisition and WITec Project FOUR 4.1 for spectral processing. Raman images were acquired with a scan speed of 18.6 s/line and 0.23 s integration time. A stress-free sample silicon wafer was used to calibrate the reference peak position.

2.7 Fingermark grading

To assess development performance, photographs of fingermarks were graded on ridge detail and contrast using a modified 5-point system from the UK Home Office CAST.¹⁹ A photographic representation of the CAST system 1,2-indanedione is shown in Table 2.3. Fingermark grades were further categorised into groups, particularly grades 3 or 4, which were considered 'useful' for identification.²²⁰ The UC (University of Canberra) scale shown in Table 2.4 was also used in Chapters 3 and 6 as a comparative scale to assess the relative performance between two treatments based on the difference in ridge details and contrast.²²⁴ However, the CAST scale was found to be more effective for providing additional information on the general trends across the different techniques. It is also compatible with categorizing grades and making comparisons with other studies. For this reason, all results presented are based on the CAST assessment scale unless stated otherwise. The fingermarks were graded by two evaluators with experience with fingermark grading system and each grading a different set of experiments. The use of a grading system is subjective to human bias; however, studies show that independent fingermark graders produce reliable data for assessing fingermark quality.²²⁵

Table 2.3: Fingermark grading scale and grading classification used in this study.^{19, 220}

Grade	Friction ridge detail developed	Contrast of ridge detail	Photographic examples	Classification
4	Very strong development: full ridge details	Very good contrast		Useful for comparison
3	Strong development more than 2/3 of fingermark continuous ridges	Good contrast		Useful for comparison
2	Limited development, ridge details present but not likely to be used for identification purposes	Moderate contrast		Detected, but not suitable for comparison
1	Weak development; evidence of contact but no ridge detail	Poor contrast		Detected, but not suitable for comparison
0	No fingermarks detected	No contrast		No fingermarks detected

Table 2.4: UC (University of Canberra) comparative scale to assess the relative performance of two

 detection methods for split fingermarks

Score	Definition
 2	Half-print treated with method A exhibits far greater ridge detail and or contrast than the corresponding half-print treated with method B
1	Half-print treated with method A exhibits slightly greater ridge detail and or contrast than the corresponding half-print treated with method B
0	No significant difference between the corresponding half-print
-1	Half-print treated with method B exhibits slightly greater ridge detail and or contrast than the corresponding half-print treated with method A
-2	Half-print treated with method B exhibits far greater ridge detail and or contrast than the corresponding half-print treated with method A

2.8 Statistical analysis

When performing statistical analysis on fingermark grades results, it is important to recognise that CAST grades are ordinal data hence a categorical scale. Therefore, most parametric tests and descriptive statistical analysis, such as means and averages are not appropriate.²²⁶ Instead, the non-parametric counterparts are

recommended. In this study, the non-parametric tests Mann-Whitney U test and Kruskal-Wallist test were used to assess whether the grades are higher or lower on average between two or more treatments. A chi-square test was used to assess whether the treatments have different distributions in fingermark grade frequencies. Additionally, the only accepted parametric test, the difference of proportions test, was performed to assess whether the proportion of grade 3s and 4s (useful fingermarks) of one treatment differs from the other.

Statistical analysis software R version 4.3.0 was used to assess whether there was significant difference in the fingermark grades between different IND/Zn treatments in Chapters 3 and 4, and between different powder suspension formulations in Chapter 5. The statistical analysis was performed by transforming the data into vectors in Excel. Vectors are the basic data structure in R and are created by using the c() function to combine variables of the same type, separated by commas. Repeated vectors or frequency of variables can be represented using the rep() function. Figure 2.4 shows an example of a fingermark grade frequency table transformed into vectors in Excel. The vectors were input as R codes to compute the statistical analysis in R. All statistical tests were conducted at 95% confidence level (p < 0.05). All statistical data analysis including fingermark grade frequency table, R codes and results are provided in Appendix A.

Grade frequency table			Grade frequency as vectors				
Treatment Treatment		nt Treatment	Vector turne				
Grades	А	В		Treatment A	Treatment B		
Grade 0	136	130	As repeated vectors to	c(rep(0,136), rep(1,67),	c(rep(0,130), rep(1,69),		
Grade 1	67	69	compute Mann-Whitney U	rep(2,93), rep(3,202),	rep(2,94), rep(3,192),		
Grade 2	93	94	test and Kruskal-Wallis test	rep(4,102))	rep(4,115))		
Grade 3	202	192	As frequency distribution to	c(136 67 93 202 102)	c(130 60 04 102 115)		
Grade 4	102	115	compute chi-squared test	(130,07,93,202,102)	(130,09,94,192,113)		

Figure 2.4: Illustrating transformation of fingermark grades frequency results (left) into vectors (right), the basic data structure in R code to compute statistical analysis.

2.9 Human ethics approval

This study required low-risk ethics approval to collect samples from human participants, in accordance with the National Health and Medical Research Council's (NHMRC) National Statement on Ethical Conduct in Human Research (2007). Ethics approval HRE2020-0162 was granted by the Curtin University Human Research Ethics Committee on the 6th of April 2020 prior to sample collection. As part of the requirements of approval, each participant received an information sheet and consent form, which provided information about; i) the study and its purpose, ii) privacy and confidentiality of their information, iii) the risk from participating iv) the voluntary participation in this study, and v) their rights to withdraw from the study at any time. The ethics approval, the participant information sheet and consent forms are provided in Appendix D.

Portions of this chapter have been published in the following articles:

Jemmy T. Bouzin, Aaron J. Horrocks, Georgina Sauzier, Stephen M. Bleay, and Simon W. Lewis. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry* **2022** *30 100439*.

DOI: https://doi.org/10.1016/j.forc.2022.100439

Jemmy T. Bouzin, Amanda A. Frick, Georgina Sauzier, and Simon W. Lewis. Preliminary evaluation of Solstice[®] PF as a replacement carrier solvent for Australian fingermark detection. *Forensic Science International* **2022** 340 111465. DOI: <u>https://doi.org/10.1016/j.forsciint.2022.111465</u>

3.1 Introduction

1,2-indanedione zinc (IND/Zn) has been widely recognised as one of the most effective amino acid treatments for the detection of latent fingermarks on porous substrates.^{82, 124, 125} Many jurisdictions have adopted it as a primary technique to increase the recoverability of latent fingermarks.^{122-126, 227} However, the Seychelles Police have not yet implemented the use of IND/Zn due to validation studies in the UK having only recently been carried out in 2018,^{125, 127, 128} noting that Seychelles Police generally adopts fingermark detection technique from the UK Fingerprint Visualisation Manual (FVM). Secondly, and more importantly, there are concerns regarding the cost and accessibility of the carrier solvent HFE-7100 (methyl nonafluorobutyl ether), as discussed in Chapter 1. This raises questions about the operational sustainability of the method in a small and remote jurisdiction like Seychelles. This chapter aims to identify and develop a sustainable formulation of IND/Zn that can be used in Seychelles.

There are several published IND/Zn formulations, and as a starting point, three active formulations used by police agencies worldwide were identified. These were a formulation recommended by the Australian National Centre for Forensic Studies (NCFS)¹⁷ and used by the Australian Federal Police; a formulation developed by the Bundeskriminalamt (BKA)²²² in use by the German Federal Police, and a formulation developed by the Centre of Applied Science and Technology (CAST)¹²⁷, presently Defence Science and Technology Laboratory, in use in the UK. The CAST formulation was chosen as the default formulation for adoption by the Seychelles Police, given their usual implementation from the UK FVM. The NCFS formulation was used as a benchmark for testing under Australian conditions, while the BKA formulation was chosen for its use of an alternative carrier solvent to HFE-7100.

The composition of IND/Zn formulations used in different countries can vary significantly for reasons ranging from reagent performance under local conditions to safety. These differences can be grouped into three categories: (i) the active ingredients consisting of 1,2-indanedione and zinc chloride; (ii) the co-solvents, which help dissolve the active chemical and provide the appropriate pH conditions

for reaction with amino acids; and (iii) the carrier solvent, which acts as a medium to transport the active ingredients to the amino acid sites of the fingermark deposit in the substrate and evaporate quickly. Table 3.1 and Figure 3.1 illustrate the differences in composition between the NCFS, BKA and CAST IND/Zn formulations.

Table 3.1: Mass in grams and molar ratio of active material present in one litre of NCFS, BKA and CASTIND/Zn working solution.





Active chemicals

Initial studies into 1,2-indanedione reagents showed that the interaction between 1,2-indanedione and amino acids is influenced by several factors, including the ambient humidity, the type of solvent used, the chemistry of the paper substrate and the presence of heat.^{110, 115, 117, 119-122, 223, 227-229} Compared to earlier formulations, the addition of zinc chloride by Australian researchers was found to increase

luminescence intensity and produce more reproducible results regardless of local temperature and humidity conditions.^{115, 122} Consequently, most 1,2-indanedione formulations contain zinc chloride in the working solution, usually in the order of 1:4 to 1:25 to indanedione (Table 3.1). Spindler *et al.*¹⁰⁴ demonstrated that zinc (II) plays a crucial role as a Lewis acid catalyst in stabilising the intermediate V in the Strecker degradation reaction of 1,2-indanedione with amino acids (Figure 3.2). The interactions of zinc (II) with the 1,3-dipole of intermediate V, accelerate the hydrolysis process to 2-amino-1-indanone (VI), which is otherwise a rate-limiting step. This was confirmed with intermediate V not detected when the reaction occurs in the presence of zinc (II) ions, indicating swift conversion of V to VI.



Figure 3.2: The reaction mechanism of IND/Zn with amino acid and zinc ions as a Lewis acid catalyst.¹⁰⁴

The study¹⁰⁴ also found that using a 10:1 excess of 1,2-indanedione, which imitates standard fingermark development, resulted in a slightly higher relative abundance of Joullie's Pink (JP) compared to stoichiometric reactions. However, this also resulted in an increased presence of yellow-brown coloured oligomer by-products. Furthermore, analysis of the products using both visible absorption and luminescence spectroscopy revealed an inverse correlation between their visible

absorbance and luminescence. Samples with strong chromophores exhibited low luminescence, and those with weak chromophores demonstrated high luminescence. This may be due to the non-fluorescent oligomeric side products masking the luminescence of JP.

Co-solvents

To maintain the solubility of 1,2-indanedione in non-polar carrier solvent, a small amount of polar solvents such as ethyl acetate, ethanol or methanol are included in all formulations. Early spectroscopic studies revealed that the presence of ethanol in the 1,2-indanedione formulation without zinc salts led to the formation of a hemiketal (2,2-diethoxy-1-indanone),²²⁸ resulting in reduced luminescence and reagent stability.¹¹⁷ However, recent studies by Nicolasora *et al.*¹²⁷ under UK conditions demonstrated that the inclusion of methanol in the CAST formulation led to stronger luminescence. This was attributed to the addition of zinc ions in the formulation, although the mechanism behind it is still not fully understood. It was also observed that excessive methanol can cause ink and ridge detail diffusion, which reduces the quality of developed fingermarks. Additionally, most formulations contain a small amount of acetic acid to counteract the alkaline nature of paper stocks and enhance the development of latent fingermarks. One exception is a formulation by Wiesner *et al.*,¹¹⁷ where acetic acid was found to cause diffused fingermarks on local paper stocks.

Carrier solvents

The carrier solvent makes up the bulk of the working solutions, with most formulations using HFE-7100.^{16, 17, 122, 230} This is because the previous standard solvent, CFC-113 (1,1,2-trichloro-1,2,2-trifluoroethane) was restricted in the 1990s due to its contribution to ozone depletion.^{231, 232} HFE-7100 has been generally preferred as it produces superior ridge detail and luminescence¹¹⁶ and is also non-flammable, low in toxicity and minimises diffusion of printed and handwritten inks.²³¹⁻²³³ However, concerns have been raised recently over a potential EU ban²³⁴ due to its contribution to fluorinated greenhouse gases and climate change as an organofluorine chemical. HFE-7100 is a hydrofluoroether (HFE) consisting of two isomers (Figure 3.3) with zero ozone depletion potential and low Global Warming

Potential (GWP).²⁰¹ Consequently, it falls under Annex II of Regulation (EU) No 517/2014 and reporting on its production, import, and export quantities is obligatory.²⁰⁰ More importantly, HFE-7100 belongs to the "forever chemicals" class of per- and polyfluoroalkyl substances (PFAS). PFAS are a class of synthetic organo-fluorine compounds that have gained attention from environmental regulators and the public due to their widespread environmental presence, resistance to degradation and potential adverse health effects.^{235, 236} Potential restrictions in the production and use of HFEs would result in limited supply of HFE-7100 globally. Consequently, there has been renewed interest in researching potential alternatives to HFE-7100 as a carrier solvent.^{233, 237, 238}



Figure 3.3: The molecular structure of HFE-7100 ($C_4F_9OCH_3$) consisting of two isomers $CF_3CF_2CF_2OCH_3$ and $(CF_3)_2CFCF_2OCH_3$.

Solstice[®] PF (trans-1-chloro-3,3,3-trifluoropropene) depicted in Figure 3.4, is marketed as an alternative to HFE-7100 in the context of an industrial degreaser.²³⁹ It has a lower GWP, as shown in Table 3.2, and is priced at half the cost of HFE-7100.^{233, 239} Recent trials conducted by Olszowska *et al.*²³³ in the UK, have indicated that Solstice[®] PF can serve as a substitute carrier solvent for HFE-7100 in amino acid sensitive fingermark detection reagents. Studies conducted using CAST IND/Zn revealed that formulations prepared with Solstice[®] PF performed comparably to the standard HFE-7100-based formulations, with no significant differences in fingermark appearance or ink diffusion.^{233, 238} On some substrates, such as brown paper, Solstice[®] PF-based IND/Zn was found to outperform HFE-7100, demonstrating improved sensitivity and ridge detail.²³⁸



Figure 3.4: The molecular structure of Solstice[®] PF (trans-1-chloro-3,3,3-trifluoropropene).

Petroleum ether has also been trialled, however, HFE-7100 has generally favoured,^{116, 117, 127} despite early studies by Bicknell *et al.*²²⁷ and Serrano *et al.*²⁴⁰ reporting both solvents to be equally effective. Petroleum ether 40-60 °C is a mixture of low boiling point aliphatic hydrocarbons primarily composed of pentanes and hexanes.²⁴¹ It is an inexpensive and readily accessible laboratory reagent^{124, 242} with the caveat of being a flammable solvent. However, its use in a properly fitted chemistry laboratory with a fume hood and safety procedures in place, particularly regarding the use of open flame and storage conditions, should significantly reduce the risk.

	HFE-7100	Solstice [®] PF	Petroleum ether 40 -60°C
Appearance	Colourless	Colourless	Colourless
Boiling Point (°C)	61	19	40-60
Heat of Vaporisation at Boiling Point (kJ/kg)	112	194	No data available
Freezing point (°C)	-135	-107	No data available
Flash Point (°C)	No data available	No data available	-35
Vapour pressure at 25 (°C) (kPa)	28	126	28
Liquid density at 25 (°C) (g/mL)	1.52	1.26	No data available
Solubility of water in solvent at 25°C (ppmw)	95	460	No data available
GWP (100 Year)	320	1	No data available
PFAS	Yes	No	No
HSP Value (δD, δP, δH)	13.7,2.2,1.0	15.5,4.8,2.2	14.9,0,0 (Hexane)

 Table 3.2: Comparison of physical properties of carrier solvents used in amino acid sensitive fingermark

 detection reagent.^{239, 241, 243, 244}

Table 3.2 also displays the Hansen solubility parameters (HSP) for each carrier solvent. Hansen's 3-parameter solubility model is commonly used as a practical guide when selecting solvents for coating systems or optimising solvents for cleaning applications.^{245, 246} HSP model predicts solubility based on three parameters: dispersion forces (δD), polarity (δP), and hydrogen bonding (δH). These parameters describe the solubility compatibility between two systems by evaluating the HSP distance (Ra) using the equation below. The HSP distance (Ra) between two molecules determines how compatible they are in terms of solubility. The smaller the Ra value between two molecules, the more likely they will be soluble with each other.

 $Ra^{2} = 4(\delta D^{2} - \delta D1)^{2} + (\delta P^{2} - \delta P1)^{2} + (\delta H2 - \delta H1)^{2}$

Hansen parameter tables are available and can also be determined experimentally.²⁴³⁻²⁴⁶ HSP can serve as an initial tool to identify potential solvent replacement, such as finding an alternative carrier solvent for amino acid reagent. For instance, if HFE-7100 is the target replacement solvent, Solstice[®] PF and petroleum ether's Ra values are 6.8 and 5.4, respectively, indicating that petroleum ether is a more compatible replacement.

Frugal forensics approach

Due to the variations in composition, different formulations of IND/Zn can vary markedly in properties such as polarity, pH and cost-effectiveness. These differences can affect the performance of the reagent and its viability for use in specific jurisdictions. As discussed in Chapter 1, techniques employed in Seychelles should ideally have minimal environmental impact, be cost-effective and have minimal supply chain disruption due to limited resources and remote location. Table 3.3 provides a cost analysis of each formulation and shows the cost for one litre of the working solution based on the prices of the reagents from at least two regular Seychelles Police suppliers in the UK and USA. Overall, the petroleum-based BKA formulation is at least two times more cost-effective than the HFE-based formulations and has a shorter lead time for reagent supply. As previously mentioned, due to environmental regulations, HFE-7100 is prone to supply chain disruption. Moreover, and rarely discussed in the open literature, is that HFE-7100 is

a proprietary fluid of 3M Novec[™], making it a single source chemical. This can have a direct impact on supply chain, as was seen in 2022 with a global supply chain concern for HFE-7100 due to manufacturing issues at the 3M production facility and exacerbated by the COVID-19 pandemic.^{202, 203} In contrast, petroleum ether is a readily available solvent from multiple suppliers, making it a more likely candidate for frugal forensics application in Seychelles.

Similar arguments in terms of the cost and availability of 1,2-indanedione can be raised. However, 1,2-indanedione is present in small amounts in the working solution, ranging from 0.227 g/L in the CAST formulation to 0.990 g/L in the BKA formulation. This results in a meagre cost factor, except in the BKA formulation where the relatively high concentration coupled with the low cost of the petroleum ether makes it a major cost factor (56 %), as shown in Table 3.3. Reducing the 1,2indanedione concentration would lead to additional savings, provided that it does not significantly affect its performance. As for supply, a remote jurisdiction can buy multiple years' worth of 1,2-indanedione crystal stock, which is reportedly highly stable.²⁴⁷ Likewise, no immediate issues are expected with zinc chloride salt and cosolvents; both commonly available reagents with trivial differences in cost. Ensuring an effective concentration ratio of the active chemicals¹⁰⁴ would be more important. However, some formulations may tolerate a given modification differently from others, with potential impact on reagent effectiveness and stability. Hence any modification should be evaluated in line with the IFRG guidelines before operationalisation.

Table 3.3: Approximate cost of one litre of 1,2-indanedione zinc working solution in USD. Chemical sourced from two regular Seychelles Police suppliers based in the UK and USA.

	Chemicals	NCFS		ВКА	ВКА			
Category		Chemical average cost	Category % cost	Chemical average cost	Category % cost	Chemical average cost	Category % cost	Price Source/Lead Time
Active	1,2-indanedione	\$43.00	15%	\$71.49) 58%	\$16.25	6%	Arrowhead Forensics, Fisher Scientific
ingreatents	Zinc chloride	\$0.04		\$0.02	2	\$0.01		Fisher Scientific, Sigma -Aldrich
	Ethyl acetate	\$7.56		\$5.42	2	\$2.51		Fisher Scientific, Sigma -Aldrich
Co-solvents	Acetic acid	\$0.22	3%	\$0.42	5%	\$0.77	2%	Fisher Scientific, Sigma -Aldrich
	Methanol	-	570		-	\$1.05	270	Fisher Scientific, Sigma -Aldrich
	Ethanol	\$0.09		\$0.23	}	-		Fisher Scientific, Sigma -Aldrich
Carrier colvents	HFE-7100	\$237.29	8.70/		- 270/	\$246.48	0.2%	Arrowhead Forensics (unavailable), Sigma-Aldrich (45 days)
Carrier solvents	Petroleum ether bp 40-60 ºC	-	8270	\$44.69	3776	-	5276	Fisher Scientific, Sigma -Aldrich (2 days)
T	otal cost	\$288.20	1	\$122.26	5	\$267.06	- -	

*Cost as of December 2022

3.2 Experimental

3.2.1 Preliminary consideration

The three active IND/Zn formulations require different processing conditions: it is recommended that items treated with CAST IND/Zn be processed in an oven at 100 °C, ²⁴⁸ while the NCFS and BKA method stipulates the use of a heat press set to 160 °C. ^{17, 222} In line with the frugal forensics approach, it was decided to use the heat press process which does not require major equipment and is more cost-effective.

3.2.2 Experimental methods

3.2.2.1 Substrate

Nine paper- and cardboard-based substrates commonly encountered by the Seychelles Police were chosen and summarised in Table 3.4. All substrates were sourced from Officeworks Australia as representative of substrates encountered in Seychelles.

Table 3.4: Details of substrates used in the head-to-head comparison of NCFS, BKA and CAST IND-Znformulations.

Substrate	Description	Colour	Weight (gsm)/Thickness
1	Winc Copy paper A4	White	80
2	PPS Envelope, 100% Recycled	White	80
3	Jburrows Notepad A4, 8mm ruled	Blue	70
4	PPS Cardboard	Brown	3.5 mm
5	PPS Heavy duty Cardboard (Fibreboard)	Brown	6.5 mm

3.2.2.2 Fingermark collection

Natural, split depletion fingermarks were collected from five donors (3 males and 2 females) between 22 and 51 years old in line with an IFRG phase 1/2 study. Each donor deposited 1st, 5th and 10th depletion series on each substrate. The remaining untreated depletions were deposited on a separate piece of the same type of substrate. This allowed an assessment of the sensitivity as well as the overall performance across multiple substrates whilst managing the number of samples. The

collection procedure was repeated to reduce donor biasing on one side of the finger, as discussed in section 2.3.2.

Samples were stored in a cupboard under office conditions (~20 °C) for at least 24 hours and cut into halves prior to IND-Zn treatment unless stated otherwise.

3.2.2.3 Fingermark development

The IND-Zn working solution and sample development for NCFS, BKA, and CAST were prepared as outlined by Stoilovic *et al.*,²²¹ Becker *et al.*,²²² and Nicolasora *et al.*¹²⁷ respectively and described in section 2.5.1. A number of modified formulations were also prepared and are specified under the relevant sections. All modified formulations followed the same preparation and development process as the parent formulation unless stated otherwise.

3.2.2.4 Photography and grading of developed fingermarks

Processed samples were re-combined and photographed as outlined in section 2.6.1 and graded using the CAST assessment and UC scale as outlined in section 2.7.

3.2.3 Head-to-head comparative assessment of existing IND/Zn formulations

The three active IND/Zn formulations were compared directly with each other on five substrates: white copy paper (Winc), white envelope (PPS), blue notepad (JBurrows), cardboard (PPS) and fibreboard (PPS). In order for each treatment to be compared head-to-head, three sets of fingermarks were collected on each substrate from each donor (1. NCFS|CAST, 2. BKA|NCFS, 3. CAST|BKA).

3.2.3.1 Investigation into co-solvents

To further investigate the role of polar solvents, the NCFS formulation, which exhibited the least luminescence in the head-to-head comparison, was compared to a modified NCFS formulation containing a higher proportion of polar solvents. The modified formulation was prepared by increasing the ethanol concentration to 4.1 % v/v, similar to the proportion of alcohol in the CAST working solution, via a 1:1 volumetric reduction in the amount of ethyl acetate. The comparison was carried out
on two substrates which showed the least (white copy paper) and the most (blue notepad) background development in the head-to-head comparison.

Methanol versus ethanol

The benefit of methanol in the CAST IND/Zn formulation reported by Nicolasora *et* $al.^{127}$ was also investigated by comparing the CAST formulation to a modified version in which methanol was directly substituted with ethanol in a 1:1 volumetric ratio. The same substrates as in section 3.2.3.1 were used.

3.2.3.2 Investigation into 1,2-indanedione concentration

To investigate the impact of reducing the 1,2-indanedione concentration, the BKA formulation containing 990 mg/L of 1,2-indanedione was compared to a modified version containing 227 mg/L of 1,2-indanedione (similar concentration as in the CAST IND/Zn). This was investigated on two substrates which IND/Zn was the most (white copy paper) and the least (cardboard) effective in developing fingermarks in the head-to-head comparison. To assess the subtle differences of the developed fingermarks between the two formulations, the UC scale was utilised with BKA 990 mg/L as the control.

Fluorescence spectroscopy studies

Interestingly, reducing the 1,2-indanedione concentration was observed to produce fingermarks with stronger luminescence. To verify this observation, fluorescence spectroscopy was carried out on amino acid spots treated with three BKA formulations containing 990 mg/L, 594 mg/L (approximately equivalent to the concentration in the NCFS formulation) and 227 mg/L of 1,2-indanedione respectively. A fourth BKA formulation without 1,2-indanedione was prepared and used as a control. The same two substrates as above (section 3.2.3.2) were used, and four separate strips of each substrate were impregnated with five 1 µL aliquots of 10 mg/L amino acid solution (alanine, leucine and phenylalanine). All strips were allowed to dry for 24 hours, mimicking the standard fingermark methodology. Each of the four BKA formulations was used to develop one strip of each substrate, following the same standard procedure for developing fingermarks with 1,2-

indanedione. The amino acid spots were then analysed using a Cary Eclipse Fluorescence Spectrophotometer (Agilent) combined with a fibre optic probe for *in situ* measurements, as outlined in section 2.6.2. Luminescence examination was also carried out using a Polilight (PL 500) for all strips at three stages: prior to impregnation with amino acids, prior to development with IND/Zn and after development to investigate for possible luminescence contamination.

3.2.4 Investigation into Carrier solvent – Solstice[®] PF as an alternative

Solstice[®] PF was investigated as an alternative carrier solvent using the NCFS IND/Zn. The NCFS formulation provides a direct comparison with the investigation²³³ carried out on the CAST formulation by substituting HFE-7100 with Solstice[®] PF in a 1:1 volumetric ratio. For the purposes of this study, the terms HFE-IND/Zn and Sol-IND/Zn are used to refer to IND/Zn reagents prepared using HFE-7100 and Solstice[®] PF, respectively and pertain specifically to the NCFS formulation unless stated otherwise. The performance of Sol-IND/Zn and HFE-IND/Zn were directly compared across a range of paper-based substrates and aged fingermarks. Additionally, the compatibility of the Solstice[®] PF based reagent with pen inks and its long-term stability was also investigated.

3.2.4.1 Overall performance of Solstice[®] PF

To assess the overall performance and sensitivity, fingermark samples were collected on three substrates: white copy paper (COS Premium), white envelopes (PPS) and cardboard (PPS, 100% FSC recycled). For this part of the experiment each donor deposited one set of three fingermarks on each substrate. For investigations into sample age, three sets of fingermarks from each donor were collected on copy paper (COS Premium) and allowed to age 1, 7 and 30 days, respectively.

3.2.4.2 Compatibility with inks

To investigate effects of the carrier solvent on ink diffusion, a collection of eighteen different pens (Table 3.5) was used. A printed template on A4 copy paper (COS Premium) was filled in with each of the pens and left to dry for at least 15 minutes

prior to immersion. The template was then divided in half, and each half was treated with HFE-IND/Zn or Sol-IND/Zn as described above, omitting the heat treatment. Samples were handled such that any diffused ink would run horizontally across the page as much as possible, rather than vertically across the other inks. The samples were left to air dry on paper towels before examination under both ambient and luminescence conditions using a Polilight (PL 500).

Pen type	Brand/model	Colour	Sample number
Whiteboard marker	Artline 577	Black	1
	Anko	Blue	2
	Staples Remarx	Green	3
Permanent marker	Sharpie Fine Point	Black	4
	Sharpie Fine Point	Blue	5
	Sharpie Fine Point	Pink	6
Gel pen	BIC Gel-ocity	Orange	7
-	BIC Gel-ocity	Purple	8
	BIC Gel-ocity	Red	9
Felt tip pen	Artline 200	Black	10
	COS Fineliner	Blue	11
	Texta Point 188	Red	12
Ballpoint pen	Artline Smoove	Red	13
	Artline Smoove	Blue	14
	Staedtler 430 M	Black	15
	BIC Economy	Black	16
	BIC Cristal M	Blue	17
	BIC Cristal M	Red	18

Table 3.5: Pen used in ink diffusion investigations.

3.2.4.3 Effect of co-solvents on ink diffusion

Based on the notably greater diffusion of samples immersed in Sol-IND/Zn, additional investigations were carried out to determine the potential contribution of each cosolvent. These investigations utilised pens (6, 8 and 14) that had shown differing degrees of visible and/or luminescent diffusion. Swatches from each pen were deposited on white copy paper and left to dry for at least 15 minutes prior to spotting with solvents using a capillary tube. Ink replicates were spotted with neat solvents as well as single and double 'knockout' mixtures based on Sol-IND/Zn. The 'knockout' mixtures were prepared according to the NCFS IND/Zn formulation without 1,2-indanedione or zinc chloride, omitting one (single) or two (double) of the co-solvent's ethyl acetate, acetic acid and ethanol. Omitted solvents were replaced by an equivalent volume of Solstice® PF. The samples were left to air dry on paper towels before being examined under ambient and luminescence conditions and then compared to control swatches of each ink. To investigate the effect of the drying time of inks, this test was repeated on ink samples left to dry for 5 days.

3.2.4.4 Reagent stability

A 500 mL batch of Sol-IND/Zn was prepared and immediately stored at 4 °C in a sealed Schott bottle. This reagent was used to treat latent fingermarks on copy paper (as described above) at 1.5, 2.5 and 4 months after preparation, and was not handled or used at any other time. Fingermarks were collected at each time interval of reagent testing. Sample halves treated with the aged reagent were compared to those treated with 'fresh' Sol-IND/Zn (less than 7 days old). Visual inspection of the solution was also conducted at these times.

3.2.5 Statistical analysis

Statistical analysis software R (version 4.3.0) was utilised to assess whether there was significant difference in the fingermark grades between the IND/Zn treatments. The non-parametric Kruskal-Wallis test was used for the head-to-head comparison (n= 450) of NCFS, BKA and CAST, as more than two treatments were being tested. The data was subjected to a chi-squared test to assess whether the treatments have

different distributions in fingermark grade frequencies. All statistical tests were conducted at 95% confidence (p < 0.05).

3.3 Results and discussion

3.3.1 Head-to-head comparative assessment of existing IND/Zn formulations

3.3.1.1 Overall performance and sensitivity

The head-to-head comparative assessment generated a total of 450 fingermark halves, and the results are summarised in Figure 3.5 for overall performance and Figure 3.6 for sensitivity. All three formulations showed comparable performance, with NCFS formulation developing 48.0 % of useful fingermarks, BKA 47.3 % and CAST 46.4 %.



Figure 3.5: Percentage of fingermarks graded (3&4), (1&2), and (0) out of 450 fingermarks developed in head-to-head comparison between NCFS, BKA and CAST 1,2-indanedione zinc formulation.

Similar trends of comparable performance were also observed when assessing the sensitivity of the three formulations. As expected, there was a gradual decrease in the number of fingermarks detected down the depletion series. All three formulations were highly sensitive with at least 28 % of the 10th depletion series was graded as useful fingermarks (Figure 3.6).

The result reveals that despite the differences in composition, the three IND/Zn formulations have similar effectiveness in developing latent fingermarks on

substrates used in this study. The high reproducibility is likely due to the addition of zinc salt, making IND/Zn reagents less susceptible to humidity.^{115, 122, 227} Spindler *et al.*¹⁰⁴ demonstrated that incorporating zinc ions into a 1,2-indanedione solution helps stabilise a crucial intermediate during the rate-limiting hydrolysis process involved in producing JP. This is an improvement from earlier non-zinc salt containing 1,2-indanedione formulations, as it makes it easier to adopt formulations and modify them. This is particularly important within the frugal forensics context, where the emphasis is on cost and supply chain rather than optimum performance.





3.3.1.2 Substrates

The NCFS, BKA and CAST formulations were all successful in developing fingermarks on each of the five substrates, as shown in Figure 3.7. Across the various substrates, the formulations exhibited similar trends, with the most useful fingermarks developed on notepad paper and the least number on cardboard surfaces.



Figure 3.7: Comparison of percentage fingermark grades assigned to fingermarks developed on five substrates in the head-to-head comparison between NCFS, BKA and CAST 1,2-indanedione zinc formulation.

However, it was observed that fingermarks developed with CAST and BKA formulations exhibited brighter luminescence than those developed with NCFS formulation. Conversely, NCFS produced fingermarks with slightly better contrast and less background development. This was particularly evident on the coloured notepad with printed lines, where the NCFS formulation exhibited a clean background and lower luminescence fingermark ridge details. In contrast, the CAST and BKA formulations produced a yellowish background, as shown in Figure 3.8, but this did not significantly impact the fingermark grades.



Figure 3.8: Fingermarks treated with (a) NCFS formulation, (b) BKA, (c) CAST on notepad paper showing difference in luminescence and background development.

As can be seen in Figure 3.8, the CAST formulation exhibited marginally stronger luminescence and background development. This is likely due to its higher percentage of polar solvents. Methanol and acetic acid make up the highest proportion of highly polar solvents in the CAST formulation (5.0% v/v), while the NCFS formulation contains the least amount in the form of ethanol and acetic acid (0.9% v/v). The observation is consistent with the findings of Nicolasora *et al.*¹²⁷, which showed that the combination of methanol and zinc chloride resulted in increased luminescence intensity, but excess methanol caused more background development. Previous studies have also reported that the carrier solvent can affect formulation properties.^{116, 122, 227} However, the NCFS formulation which also uses HFE-7100 but with a much lower proportion of highly polar solvents (alcohol and acetic acid), exhibited the least luminescence and background development compared to the CAST formulation. This suggests that although the carrier solvent may influence the luminescence intensity and background development, the co-solvents likely have a greater impact. Further investigation into the benefits of polar solvents was therefore conducted and discussed in section 3.3.2.

The decreased effectiveness on cardboard is consistent with other studies,^{125, 126} due to low quality paper and a less porous surface. The chipboard and fibreboard used in

this study are made from recycled paper, which has a lower cellulose content and shorter fibres compared to standard paper.^{249, 250} The cellulose matrix in paper has been suggested to play an important role in the reaction between the IND/Zn reagent and amino acids in latent fingermarks by directing JP's fluorophoric conformation via hydrogen bonding.¹⁰⁴ Furthermore, cardboard boxes are generally made with a layer of high grammage Kraft paper to reduce porosity and increase durability.^{250, 251} The level of porosity is particularly relevant in the development of fingermarks on paper and cardboard surfaces, since less porous surfaces tend to absorb fingermark deposits to a lesser extent,⁹¹ resulting in weaker development. This is likely because the fingermark deposits close to the surface are less protected causing a decrease in the target material. It was observed that BKA and CAST formulations were slightly more effective on cardboard than NCFS formulations. The marginally stronger luminescence exhibited by the CAST and BKA formulations may be beneficial in developing fingermarks on cardboard, as weak development is generally seen on less porous substrates.

3.3.1.3 Statistical Analysis

A Kruskal-Wallis test performed on the head-to-head comparative data showed that there is no significant difference in the fingermark grades between the three treatments (p = 0.93). Similarly, the chi-squared test showed that there is no significant difference in the distributions of fingermark grade frequencies between the different treatments (p = 0.73). The statistical results reinforce the comparative performance observed between the three active IND/Zn formulations (NCFS, BKA and CAST). The R code and results of the statistical analysis are provided in Appendix A-2.

3.3.1.4 Frugal forensics analysis of existing operational IND/Zn protocols

Although the formulations have similar performance, their composition can have significant implications when applied in a frugal forensics context. This is especially true when it comes to the carrier solvents, which present challenges related to cost and supply chain risks when implementing an IND/Zn formulation for the Seychelles Police. The use of HFE-7100 in the NCFS and CAST formulations is expensive and subject to supply chain disruption, jeopardising service continuity. In contrast, the BKA formulation's use of petroleum ether 40-60 °C is more cost-effective and readily available from multiple suppliers, allowing for better budget and supply chain risk management. Therefore, the BKA formulation was identified as the more sustainable method that the Seychelles Police Force can adopt with minimal modifications.

The publication of the head-to-head comparative findings in 2022 prompted the New South Wales (NSW) Police to conduct trials on the NCFS formulation.^{202, 252} Their study revealed that substituting HFE-7100 with petroleum ether resulted in comparable performance and an estimated annual savings of \$40,000 AUD.²⁵³ Additionally, in late 2022, 3M announced the discontinuation of PFAS production by the end of 2025 (see Appendix C). This has increased the urgency to find an alternative carrier solvent. Altogether, it highlights the thesis's timeliness and the potential impact of the frugal forensics approach to both the Global North and South.

3.3.2 Carrier solvent – Petroleum ether

The sustainability of the BKA formulation was established largely based on the carrier solvent - petroleum ether. Hence, further investigation was conducted to examine whether the co-solvents and the amount of 1,2-indanedione could be optimised to enhance the sustainability of the formulation.

3.3.2.1 Co-solvents

Investigation into the role of polar solvents revealed that the more polar modified NCFS formulation displayed marginally stronger luminescence on both substrates (white copy and notepad paper), similar to the CAST formulation as shown for the copy paper in Figure 3.9. The findings support the suggestion that increased polarity of the co-solvent enhances the luminescence of developed fingermarks, possibly due to greater solubility of 1,2-indanedione.¹²⁷ However, as mentioned previously, this also leads to increased background development and ridge diffusion. Therefore, there is a need to balance sensitivity and quality of the developed fingermarks when modifying the polar solvents in the IND/Zn formulation.

Methanol versus ethanol

Assessment of the developed marks indicated that the use of methanol or ethanol in the formulation resulted in similar effectiveness. However, marginally stronger luminescence was observed with the methanol-based formulation, as shown in Figure 3.10. This was attributed to the relatively higher polarity of methanol compared to ethanol.



Figure 3.9: Fingermarks on copy paper treated with NCFS formulation (left) and the more polar modified NCFS formulation (right) showing the marginally greater background development.



Figure 3.10: Fingermarks on copy paper treated with modified CAST formulation containing ethanol (left) and CAST formulation containing methanol (right).

Frugal forensics analysis of co-solvents in BKA formulation

With BKA identified as a more sustainable formulation, it was decided not to modify the polar solvents since it already strikes the right balance with similar luminescence intensity to CAST formulation and only minor background development compared to NCFS. With regards to the use of alcohol, ethanol was preferred over methanol due to the toxic nature of the latter. Nonetheless, the considerable cost of 1,2indanedione in the formulation (~ 58 %) prompted further investigation into the impact on effectiveness in reducing its concentration.

3.3.2.2 1,2-indanedione concentration

Interestingly, reducing the 1,2-indanedione concentration was found to improve the luminescence of developed fingermarks. The BKA 227 mg/L formulation exhibits better contrast compared to the BKA 990 mg/L formulation, as shown in Figure 3.11. This was more pronounced on white copy paper with a 69% of UC scale scores of 1 or 2 compared to 10 % on cardboard. The result was attributed to excess 1,2-indanedione in the BKA 990 mg/L formulation, forming non-fluorescent oligomeric side products masking the luminescence of JP, which is consistent with Spindler *et al.*'s¹⁰⁴ study investigating the reaction mechanism of 1,2-indanedione with amino acids. The lower paper quality and less porous surface of cardboard may explain why this observation is less pronounced on cardboard.







Spectroscopy studies

To verify this observation, spectroscopy studies were conducted. The results of the A4 the copy paper is depicted in Figure 3.12, with BKA 277 mg/L producing stronger fluorescence, which aligns with the previous observations. Examination of the amino

a)

acid spots showed that BKA 990 mg/L (highest concentration) exhibited a deep purple-pink colour under visible light and a dark orange colour under Polilight at 505 nm. This changed to light purple-pink and bright yellow, respectively, with BKA 227mg/L (lowest concentration). These findings reinforce the suggestion that excess 1,2-indanedione forms non-fluorescent oligomeric side products that mask JP luminescence. It is worth noting that the dual visualisation of fingermarks in absorbance and luminescence modes, provides a valuable advantage to practitioners to quickly identify where the fingermark is located on an item, leading to better preservation and quicker processing time.

The results for cardboard are depicted in Figure 3.13, indicating a similar outcome, although less pronounced as seen in earlier observation. The experiment was also repeated, reducing the amount of amino acid impregnated per spot from 10 μ g to 5 μ g, 1.0 μ g and 0.5 μ g to mimic a depletion series, and similar results were observed.

Frugal forensics analysis of 1,2-indanedione concentration in the BKA formulation

As discussed above, the BKA formulation has four times more 1,2-indanedione than the CAST formulation, which adds a considerable cost to the reagent. Additionally, the dual visualisation of fingermarks using visible absorbance and luminescence examination is advantageous in casework. Therefore, it was decided to reduce the 1,2-indanedione concentration in the BKA formulation to 594 mg/L - similar to the NCFS formulation. This lowers the cost of 1,2-indanedione in the formulation from \$71.49 to \$43.07 USD per litre of working solution (40 % decrease) and strikes a balance between cost-effectiveness and dual visualisation.



Figure 3.12: Amino acid spots on copy paper treated with BKA IND/Zn formulations containing different 1,2-indanedione concentrations, (a) emission spectrum excited at 505 nm illustrating the difference in luminescence intensity (b) difference in colour under visible light, (c) difference in colour under Polilight at 505 nm.



Figure 3.13: Amino acid spots on cardboard treated with BKA IND/Zn formulations containing different 1,2-indanedione concentrations, (a) emission spectrum excited at 505 nm illustrating the difference in luminescence intensity (b) difference in colour under visible light, (c) difference in colour under Polilight at 505 nm.

3.3.3 Carrier solvent - Solstice® PF as an alternative

3.3.3.1 Overall performance of Solstice[®] PF

General observation

In this part of the study, the inverse correlation between visible absorbance and luminescence of developed fingermarks was also observed, depicted in Figure 3.14. HFE-IND/Zn produced a deep purple-pink colour under visible light, while Sol-IND/Zn exhibited a light purple-pink colour and stronger luminescence. This suggests that the use of Solstice[®] PF as a carrier solvent can have an impact on the chemical interaction between 1,2-indanedione and the target amino acids, resulting in subtle differences in product colour and luminescence. While similar observations have been made regarding a variety of carrier solvents,^{116, 122, 227} the relevant mechanisms have yet to be identified.



Figure 3.14: Latent fingermarks on copy paper treated with HFE-IND/Zn (left) and Sol-IND/Zn (right), (a) viewed under white light and (b) in luminescence mode.

Overall performance and sensitivity

Overall, the two formulations showed similar levels of effectiveness and sensitivity, except for slight differences on cardboard surfaces (Figure 3.15). As discussed earlier, IND/Zn performance is negatively impacted by less porous substrates that do not effectively absorb fingermark residue, as well as substrates with lower cellulose content.^{254, 255} Sol-IND/Zn was more effective on cardboard, developing more useful fingermarks than HFE-IND/Zn (Figure 3.16). This is consistent with results reported by Zhao *et al.*²⁵⁶ on brown paper and newspaper with CAST IND/Zn.²⁵⁷ The stronger luminescence exhibited by Sol-IND/Zn treated fingermarks may provide better contrast on cardboard surfaces.



Figure 3.15: Comparison of percentage fingermark grades assigned to fingermarks treated with HFE-IND/Zn and Sol-IND/Zn on white copy paper, white envelope and cardboard.



Figure 3.16: Fingermark halves developed on (a) white envelope treated with HFE-IND/Zn (left) and Sol-IND/Zn (right) and (b) cardboard treated with Sol-IND/Zn (left) and HFE-IND/Zn (right).

Ageing samples

Sol-IND/Zn and HFE-IND/Zn were found to perform comparably on samples up to 30 days old developed on white copy paper (Figure 3.17). Overall, the quality of IND/Zn treated fingermarks for both reagents declined initially from 1 day to 7 days and then improved with increasing age over 30 days. While this could be a result of intra-donor variation in sample composition, a similar trend was noted by Zhao *et al.*²⁵⁶ when comparing CAST HFE-IND/Zn and Sol-IND/Zn on copy paper, suggesting ageing mechanisms may also play a part.²⁵⁷ Other reports indicate that IND/Zn efficiency declines with fingermark age over 3-4 weeks following deposition.^{258, 259} Further analysis of individual donors shows that the difference could be attributed to donors 4 and 5 as indicated by Figure 3.17b.



Figure 3.17: Comparison of percentage fingermark grades assigned to a) fingermarks aged 1, 7 and 30 days b) fingermarks aged for each donor developed on white copy paper and treated with HFE-IND/Zn and Sol-IND/Zn.

3.3.3.2 Compatibility with inks

Following immersion in the IND/Zn working solutions, visible and/or luminescent diffusion was exhibited by a number of inks, which can be attributed to differences in their composition (Figure 3.18).^{260, 261} Most of the pens contained inks that were alcohol- or water-based (according to manufacturer information), with the exception of the ballpoint pens, which were based on glycol and benzyl alcohol, or oil. Red ballpoint pens, red and purple gel pens and pink permanent markers appeared to be

particularly affected, which may be due to the solubility of red colourants such as the dye eosin.²⁶¹ Significantly greater diffusion was exhibited by ink samples immersed in the Sol-IND/Zn reagent, which was unexpected given the results reported by Olszowska *et al.*²³³ and the less polar nature of the NCFS formulation compared to that used by CAST.



Figure 3.18: Comparison of the effect of Sol-IND/Zn and HFE-IND/Zn reagents on inks, photographed under ambient light and luminescent mode.

Olszowska *et al.*²³³ conducted a similar investigation, primarily using blue and black ballpoint pens as examples of commonly used pens.²⁶² Their results showed that similar, minor diffusion effects were caused by CAST IND/Zn, regardless of carrier solvent type. By total volume, the NCFS IND/Zn reagent contains a significantly lower volume of highly polar co-solvents: about 20 % of that of the CAST formulation. Additionally, the CAST reagent contains methanol, a solvent not used in the NCFS formulation. The greater proportion of polar co-solvents in the CAST formulation, particularly the inclusion of methanol, is likely to have a greater effect than any differences in polarity between Solstice[®] PF and HFE-7100. Methanol is used in the CAST formulation to increase luminescence intensity; however, increased methanol concentration is known to cause diffuse ridge detail as well as unwanted ink bleeding.^{116, 263}

3.3.3.3 Effect of co-solvents on ink diffusion

Following these results, the cause of the excess diffusion during Sol-IND/Zn treatment was investigated by spotting neat solvent and a series of 'knockout' solvent mixes (replacing one or two of the co-solvents with an equivalent volume of Solstice® PF) onto a subset of inks. It was found that Solstice® PF and all three co-solvents are sufficiently polar to dissolve at least one type of ink, while HFE-7100 alone had no visible effect (Figure 3.19). Similar results were observed when the experiment was repeated on 5 day old ink samples, indicating minimal effect of ink drying time on the results. However, it should be noted that longer term trials on a larger number of inks are required.



Figure 3.19: Inks spotted with neat solvent and photographed under ambient light and luminescence mode.

Further investigations utilising solvent mixes (equivalent to concentration in the IND/Zn reagent) were inconclusive regarding the exact cause(s) of diffusion for some ink types. Though greater ink diffusion was seen with Sol-IND/Zn compared to HFE-IND/Zn, the differences between reagent formulations cannot be solely attributed to the carrier solvent. As an example, ink 8 (a gel ink) did not appear to be affected by Solstice[®] PF alone but exhibited diffusion when treated with solvent mixes containing acetic acid. Gel inks are typically thickened with agents such as xanthan gum,²⁶⁴ which is soluble in dilute acids.²⁶⁵ It is unclear at this stage whether the differences in the degree of ink diffusion caused by the HFE-7100 and Solstice[®] PF formulations are due to a simple additive effect of Solstice[®] PF and the co-solvents, or whether there is additional interaction between the carrier solvent and the more polar co-solvents. The proportion of highly polar co-solvents in the NCFS formulation may simply be low enough that a change in carrier solvent polarity has a more noticeable effect. Noting from the HSP values in Table 3.2 the polarity and hydrogen bonding are stronger for Solstice[®] PF than HFE-7100.

3.3.3.4 Reagent stability

After 1.5 months in storage (refrigerator), the Sol-IND/Zn reagent was visually unchanged compared to its appearance immediately after preparation. However, after 2.5 months, the reagent appeared cloudy (Figure 3.20). A white precipitate (presumed to be 1,2-indanedione) was suspended in the solution, which could not be redissolved by allowing the reagent to come back to room temperature or by manual agitation. The precipitate continued to form during storage, such that the solution was almost completely opaque 4 months after preparation. In comparison, a batch of HFE-IND/Zn prepared and stored contemporaneously remained visually unchanged during this time. Further batches of Sol-IND/Zn were prepared to confirm these results, and in these cases, precipitation was first observed at 2-3 weeks. Regardless, the aged Sol-IND/Zn reagent still performed equally to fresh solution at all investigated time periods.



Figure 3.20: Sol-IND/Zn after 2.5 months in refrigerated storage, showing white precipitate.

These observations contrast significantly with those of Olszowska *et al.*²³³ regarding CAST Sol-IND/Zn, which has been reported to remain stable for at least 6 months.²³³ However, the standard HFE-7100 based NCFS and CAST IND/Zn formulations have

very different shelf lives. The CAST formulation has a reported shelf life of up to one year,¹²⁷ whereas the NCFS working solution has a shelf life of 3 months.¹⁷ This is potentially due to the greater polarity of the UK formulation, which would enable 1,2-indanedione and zinc chloride to remain more stable in solution.

It remains unclear at this stage why Sol-IND/Zn appears to deteriorate faster than HFE-IND/Zn. The low storage temperature (4 °C) may be a contributing factor as room temperature storage is recommended for the NCFS formulation,¹⁷ though both reagents were stored under the same conditions. Notably, the NCFS Workshop Manual recommends preparing separate stock solutions of 1,2-indanedione and zinc chloride, each with a shelf life of one year. This approach may present a solution for longer-term storage of Sol-IND/Zn, with the working solution prepared as needed.

3.3.3.5 Frugal forensics analysis of Solstice[®] PF as a carrier solvent for fingermark development

These results indicate that Solstice[®] PF is a satisfactory alternative carrier solvent to HFE-7100 particularly on cardboard surfaces. However, from a frugal forensics perspective it presents several challenges in terms of transportation, storage and cost-effectiveness coupled with similar drawbacks as HFE-700.

Solstice[®] PF has a low boiling point of 19 °C, however it has a high heat of vaporisation of 194 kJ/kg, enabling it to maintain a liquid state in a climate-controlled laboratory environment.²³⁹ However, because of this property, it must be stored in a pressurised cylinder. This can be problematic for small jurisdictions like Seychelles due to additional freight charges and use of more storage space for pressurised cylinders. Additionally, Solstice[®] PF is also a single source chemical from Honeywell,²³⁹ consequently suffering from similar supply chain risks as HFE-7100. Therefore, it was determined that Solstice[®] PF was not the best alternative carrier solvent. Instead, the BKA formulation containing petroleum ether 40-60 °C and a modified 1,2-indanedione concentration of 594 mg/L was brought forward as a more sustainable option for laboratory trials.

3.4 Conclusions

This chapter investigated existing 1,2-indanedione zinc formulations to develop a sustainable version suitable for use in Seychelles and Global South jurisdictions. The study began with a head-to-head comparative assessment of three operational formulations used in Australia (NCFS), Germany (BKA) and the UK (CAST) on representative substrates commonly encountered in Seychelles. Despite the composition variations primarily associated with the difference in the carrier solvents, all three formulations were highly sensitive and gave similar performance. The BKA formulation was identified as the most cost-effective and supply-chain-friendly option, making it an ideal choice for resource-limited micro-jurisdictions that require minimal adaptation. The IND/Zn formulations were then modified using a frugal forensics approach, specifically focusing on modifying the co-solvents, active chemicals and carrier solvents.

Investigation into the role of polar solvents revealed that the more polar-modified IND/Zn formulations produce marginally stronger luminescence of the developed fingermarks, particularly with the use of methanol. The result supports the suggestion that boosting the co-solvents polarity can improve the luminescence of fingermarks, likely due to the increased solubility of 1,2-indanedione. However, an increase in polar solvent was also found to cause more background development and ridge diffusion, hence, it is crucial to have a balance between sensitivity and quality of fingermarks when altering polar solvents in the IND/Zn formulation. Consequently, the amount of co-solvents in the BKA formulation was maintained as it was seen to strike the right balance, and the use of ethanol was chosen over methanol for safety considerations, in line with the frugal forensic principles.

The concentration of 1,2-indanedione in the BKA formulation was found to be a significant cost factor, and that reducing it improves the effectiveness of the reagent. The result was attributed to excess 1,2-indanedione forming non-fluorescent oligomeric side products masking the luminescence of JP. Therefore, the formulation was modified to contain 594 mg/L of 1,2-indanedione, reducing the cost due to 1,2-indanedione by 40 %.

Replacing the expensive and supply chain-disruptive solvent HFE-7100 in IND-Zn formulation is the main challenge in developing a sustainable formulation. To address this challenge, Solstice[®] PF was investigated as a potential replacement and showed comparable effectiveness to HFE-7100 in the NCFS formulation and producing stronger luminescence, making it better for visualising fingermarks on surfaces such as cardboard. However, it can cause issues with the physical integrity of paper evidence due to relative higher diffusion of inks observed. Increased ink diffusion suggests chemical interactions with other reagent components that may affect stability. Solstice[®] PF-based reagent was found to form a precipitate within a month of storage, but it did not affect performance over 4 months. Moreover, Solstice[®] PF presented similar drawbacks as HFE-7100 in terms of supply chain risks.

Based on these results and considering the HSP values and comparable performance of the BKA-petroleum ether 40-60°C based formulation in the head-to-head comparative assessment, no further modification was made with respect to the carrier solvent. Although highly flammable, petroleum ether 40-60°C is an inexpensive and readily available solvent manufactured by multiple sources, providing a better alternative in managing the cost and exposure to supply chain risks associated with HFE-7100. For restricted-budget laboratories where cost and supply chain are key factors in deciding the formulation composition, petroleum ether 40-60°C presents a more sustainable carrier solvent.

As a result of the series of modifications using the frugal forensics framework, a sustainable method for further trials on its potential use in Seychelles was established and referred to as the Seychelles Police Force 1,2-indanedione zinc (SPF IND/Zn). The SPF IND/Zn formulation is as follows -

0.60 g of 1,2-indanedione
45 mL of ethyl acetate
10 mL of glacial acetic acid
900 mL of petroleum ether 40-60 °C
10 mL of zinc chloride stock solution
Zinc stock solution is as follows -

200 mg of zinc chloride 30 mL of ethanol Processing conditions are as follows – Heat press for 10 seconds at 160 °C

The SPF IND/Zn formulation was then subjected to laboratory trials to assess its suitability for operational implementation in Seychelles, as detailed in the next chapter. Furthermore, this investigation established a foundation for the experimental methodology employed in the laboratory trials.

Portions of this chapter have been published or to be submitted for publication in the following articles:

Jemmy T. Bouzin, Jason Merendino, Stephen M. Bleay, Georgina Sauzier, and Simon W. Lewis. New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc. *Forensic Science International: Reports* **2020** *2* 100145.

DOI: https://doi.org/10.1016/j.fsir.2020.100145

Jemmy T. Bouzin, Georgina Sauzier, and Simon W. Lewis. A sustainable 1,2indanedione/zinc formulation for Seychelles. To be submitted.

4.1 Introduction

As discussed in Chapter 2, the IFRG guidelines²⁴ divide the research process into four phases, from the initial proof-of-concept (phase 1) to operational evaluation (phase 4). Each research phase follows specific guidelines regarding the type and number of substrates, donors, and fingermarks. As a method progresses to the latter phases, using more realistic fingermark samples to assess operational suitability becomes important. This is because, in real-life scenarios, fingermarks are not intentionally deposited. Therefore, the use of incidental fingermarks provides more representative samples that are likely to be encountered. In Chapter 3, a modified version of 1,2-indanedione/zinc chloride (IND/Zn) formulated for sustainable use by the Seychelles Police Force (SPF) was developed using deliberately deposited fingermarks in line with a phase 1/2 study. In this chapter, the adapted formulation (referred to as SPF IND/Zn) is evaluated through laboratory trials, including the use of incidental fingermarks in line with a phase 2/3 study.

Another IFRG recommendation is to test reagents under local conditions which accounts for available resources in terms of, personnel, equipment, substrates and environmental conditions. The use of local resources is particularly important in a frugal forensics approach as it acknowledges and adapts to local challenges, which is the cornerstone of the concept. This was demonstrated in Chapter 3 using a more economical carrier solvent and processing conditions for IND/Zn that are more robust to cost or supply chain limitations. The consideration of substrates and environmental conditions is more related to its impact on the latent fingermark detection effectiveness. In Chapter 3, the issue of substrate was addressed by using representative samples commonly encountered in Seychelles.

More importantly, the environmental conditions between Seychelles and the United Kingdom (UK), where most techniques adopted by SPF are developed, differ in terms of humidity and temperature as discussed in Chapter 1.^{191, 266} For IND/Zn, the susceptibility to the environmental conditions seems reduced with the addition of zinc chloride to the formulation.¹⁰⁴ Additionally, on porous substrates, the eccrine residue absorbed into the substrate matrix provides a level of protection against

environmental factors. However, long-term exposure to humidity and temperature may affect the survivability of the latent fingermarks through an increased rate of degradation or diffusion processes. Studies have shown that humidity significantly affects the eccrine constituents; for example, chlorides have been determined to diffuse faster under high relative humidity.^{35, 70, 267} In another study investigating the potential for degradation of selected amino acids present in eccrine residue, thermal degradation was observed after 3 minutes at 100 °C,²⁶⁸ although this was not performed on an actual fingermark residue. The combined cyclic exposure to high humidity and temperature may influence the degradation and diffusion of eccrine components of fingermarks over time. This can potentially have consequences on the detection of old fingermarks between jurisdictions with significantly different climatic conditions.

Amino acid sensitive reagents such as IND/Zn, 1,8-diazafluoren-9-one (DFO), and ninhydrin have long been proposed for developing older fingermarks on paper ^{36, 269}. This has been practically demonstrated by Bleay *et al.* ⁸² where IND/Zn, DFO, and ninhydrin used in sequence were found effective in developing fingermarks on 32 year old cheques kept under controlled environments, while considerably less effective on older documents with an unknown history of environmental exposure. In contrast, Boudreault *et al.* ⁶⁹ reported a significant decrease in the quality of developed fingermarks over time on white copy paper treated with IND/Zn, although this was not observed for fingermarks treated in sequence with ninhydrin. Together, these studies suggest that while the binding of amino acids to cellulose is likely to provide some stability, amino acids are not completely stable.

At the initial stage of this thesis, the trends in the effectiveness of the three active IND/Zn formulations (BKA, NCFS and CAST) for visualising latent fingermarks on paper substrates of various ages were explored under two different climatic conditions; those of Western Australia (WA) and the UK. This study was performed on incidentally handled documents representing three ageing periods (< 2, 35-40, 75-90 years) stored under UK and Western Australian conditions. The results indicated that the IND/Zn formulation is highly effective, developing fingermarks on 80 year old documents, and significantly increasing the established timescale for fingermark

detection with amino acid sensitive reagents. Comparison across the three ageing periods showed a common trend of a decrease in useful fingermarks due to progressive diffusion of the target amino acids occurring over time. Interestingly, the difference in climatic conditions appears to be more influential on fingermark development success rate than the formulation used. Exposure to environmental factors is likely to accelerate progressive diffusion of amino acids, though the natural moisture and aging of paper may also be influential. Therefore, the exposure and storage conditions and type of paper are important parameters to consider when assessing the success rate in developing old fingermarks on paper.

This chapter aims to evaluate the SPF IND/Zn formulation, taking into consideration the IFRG guidelines and the environmental considerations. It begins with a preliminary investigation of SPF IND/Zn followed by laboratory trials, including the use of incidental fingermarks.

4.2 Experimental

4.2.1 Experimental consideration

In line with the IFRG guidelines,²⁴ the laboratory trials were intended to be conducted in Seychelles; however, due to the COVID-19 pandemic all investigations were carried out in WA (discussed in section 2.2.3). Considerations were made to differences in local conditions between WA and Seychelles which may affect IND/Zn efficacy; namely climate (particularly humidity), as well as the composition of paper and other commonly available porous substrates.^{24, 115, 223, 227, 255, 270} All substrates were sourced as representative of ones encountered in Seychelles, and additional experiments under humidity and elevated temperature to mimic the tropical conditions of Seychelles were conducted.

Given that the experiment was conducted in Australia, the National Centre of Forensic Science (NCFS) formulation was chosen for the comparative assessment with SPF IND/Zn.

4.2.2 Experimental methods

4.2.2.1 Substrate

Six substrates were used in this study, summarised in Table 4.1. All substrates were sourced from Officeworks Australia as representative of substrates encountered in Seychelles.

Table 4.1 : Details of substrates used in the SPF IND/Zn formulation laboratory trials.	
--	--

Substrate	Description	Colour	Weight (gsm)/Thickness
1	COS Premium Copy paper A4	White	80
2	OfficeMax Copy paper A4, 50% Recyled	White	80
3	Plain white Envelope D5	White	
4	JBurrows Notepad A4, 8mm ruled	Blue	70
5	PPS Cardboard	Brown	3.5 mm
6	Newsprint (The West Australian)	Grey	

4.2.2.2 Fingermark collection

Due to COVID-19 restricting the number of donors, natural fingermarks were collected from ten donors (six males and four females) between 22 and 60 years old. The comparative assessment was repeated using incidental fingermarks to increase the number of samples and obtain a reliable result. The fingermark collection procedure is outlined under each section.

4.2.2.3 Fingermark development

The IND-Zn working solution and sample development for NCFS was prepared as outlined by Stoilovic *et al.*²²¹ and described in section 2.5.1 together with that of SPF formulation.

4.2.2.4 Photography and grading of developed fingermarks

The photography conditions and grading are outlined in sections 2.6.1 and 2.7, respectively.

4.2.3 Preliminary investigations

A series of preliminary investigations of the SPF formulation were conducted before the laboratory trials.

4.2.3.1 Ink compatibility

The compatibility of petroleum ether with inks was also investigated following the same procedure as the experiment in section 3.2.6.2, except that the ink templates were allowed to dry for 5 days before each half was treated with SPF or NCFS IND/Zn.

4.2.3.2 Processing conditions

Processing samples at ambient temperature by leaving samples overnight can be beneficial when a large number of samples need to be processed at any one time. The effectiveness of the process condition was investigated on white copy paper (COS premium) and cardboard using three donors. Each donor deposited two sets of five depletion series fingermarks on each substrate. Samples were aged for 24 hours prior to treatment with SPF IND/Zn. After treatment, one set was processed with a heat press at 160 °C for 10 seconds and the other with ambient heat (20-25 °C) in a closed box under office conditions. Both samples were photographed following the procedure outlined in section 2.6.1 on day 0, 1, 2, 3 and 4 after treatment.

4.2.3.3 Simple comparative assessment

Prior to laboratory trials, a preliminary comparative assessment was conducted using natural quartered fingermarks to compare SPF to the three active IND/Zn formulations using a similar collection procedure as section 2.3.2. Five donors deposited fingermarks on two substrates: white copy paper (COS Premium) and blue notepad (JBurrows), using the increased surface area of the right and left thumb. Each quarter was treated with a different IND/Zn formulation.

4.2.4 Laboratory trials

4.2.4.1 Comparative assessment

The developed SPF IND/Zn was compared to the NCFS formulation through a laboratory trial. Following a similar collection procedure as section 2.3.2, a total of ten donors deposited fingermarks on five substrates: white copy paper (COS Premium), white copy paper (OfficeMax 50% recycled), blue notepad (JBurrows), cardboard (PPS) and newspaper. Each donor deposited two sets of six depletion series fingermarks on each substrate. The samples were stored for 1 week following the procedure outlined in section 2.3.3. The collection was repeated, except that the samples were aged for 4 weeks, covering a timeframe in which casework samples are more likely to be processed.

4.2.4.2 Incidental fingermarks

Incidental latent fingermarks were collected through an exercise where donors verified the content of an envelope, simulating activity in theft cases. This follows a previously developed method of using incidental fingermarks²⁷¹ for more realistic samples to enhance the accuracy of the result. Ten donors were instructed to verify the contents of a plain white envelope containing a 10 x 7 cm strip each of four substrates: white copy paper (COS Premium), white copy paper (OfficeMax 50% recycled), blue notepad (JBurrows) and cardboard (PPS). After verifying each item and placing it back into the envelope, all five paper substrates (including the envelope) were collected. Donors were asked to repeat the task after at least 30 minutes to allow skin secretions to replenish on the fingertips, resulting in two sets of samples and a total of ten items to be processed per donor.

All samples were stored between 24 to 30 hours in an office cupboard prior to development. One complete set of substrates from each donor was treated with SPF and the other with NCFS IND/Zn formulation.

4.2.4.3 Environmental considerations

To take into consideration the differences in local conditions between WA and Seychelles that may impact the efficacy of SPF IND/Zn, an experiment was conducted to compare fingermark samples stored in an office setting to those stored in a simulated Seychelles dwelling. This would provide insight into the effectiveness of using SPF in Seychelles. To simulate conditions in Seychelles, the attic (level 5) of the Curtin University Chemistry Building 500 was used. This non-air-conditioned space contains louvre blades that facilitate ventilation, which are typical of the dwellings in Seychelles. Ten donors donated a single set of fingermarks on two substrates: copy paper (OfficeMax 50% recycled) and cardboard. Following procedure 2.3.2, a total of four sets of fingermarks were collected on each substrate, one for each aging period of 1 day, 1 week, 1 month, and 3 months. Each half of the split fingermarks was placed in a carton box and stored/aged either in office conditions or in the level 5 space. Data loggers were placed at each site to monitor the temperature and humidity of the storage locations. The experiment was conducted in the late summer to take advantage of the high level of humidity during this season in WA.²⁷²

4.2.4.4 Reagent stability

To assess the reagent stability, SPF IND/Zn solution was used to treat latent fingermarks on copy paper (COS Premium) at 1, 2, 3 and 6 months after preparation. Five split depletion series fingermarks were collected from five donors at each time interval of testing. All samples were stored for at least 24 hours in an office cupboard prior to development. Sample halves treated with the aged reagent were compared to those treated with 'fresh' (less than 7 days old) SPF IND/Zn solution. Visual inspection of the solution was also conducted at these times.

4.2.4.5 UV-visible spectroscopy

Further investigation was performed using visible spectroscopy to assess changes in 1,2-indanedione peak absorbance due to observed differences in the SPF reagent's colour over time (Figure 4.1). SPF IND/Zn working solutions of more than 6, 12 and 18 months since preparation, together with a fresh solution (less than 1 day old) and

a reference blank were analysed using an Agilent Cary 60 UV-Vis Spectrophotometer as outlined in section 2.6.3.



Figure 4.1: SPF 1,2-indanedione working solution less than 1 day old (left) and more than 6 months old (right) showing difference in colour.

4.2.5 Statistical analysis

Statistical analysis software R version 4.3.0 was utilised to assess whether there was significant difference in the fingermark grades between the IND/Zn treatments. The non-parametric Mann-Whitney U test was used for laboratory trials comparing SPF and NCFS (n = 1200). The laboratory trials and incidental fingermarks data (n=100) were subjected to a chi-squared test to assess whether the treatments have different distributions in fingermark grade frequencies. Additionally, a difference in proportion test was also performed on the laboratory trial data, as it is one of the acceptable parametric tests²²⁶ that can be applied to fingermark grading comparison. All statistical tests were carried out at 95% confidence level (p < 0.05).
4.3 Results and discussion

4.3.1 Preliminary investigations

4.3.1.1 Inks compatibility

The inks compatibility experiment revealed that the petroleum ether carrier solvent in SPF IND/Zn and HFE-7100 used in the NCFS formulation caused diffusion on the same type of inks. The effect was slightly more pronounced with petroleum ether, as shown in Figure 4.2, but less compared to Solstice[®] PF evaluated in Chapter 3. Diffusion was more evident with the pink permanent marker (sample 6), which typically contains non-polar colourants and resins in a volatile solvent. Non-polar colourants are used to prevent dissolving of the inks when in contact with water and the resin secures the ink colourant to the paper once the solvent evaporates.²⁷³ The high alkane content in petroleum ether 40-60 °C is likely to increase diffusion of the non-polar constituents in permanent marker. Heptane, the major constituent of petroleum ether, is highly non-polar with a high dispersion force, as indicated by the HSP values (Table 3.2, Chapter 3). In general, most pens including ballpoints, did not show diffusion, indicating petroleum ether's effectiveness comparable to HFE-7100.



Figure 4.2: Comparison of the effect of SPF IND/Zn and NCFS IND/Zn reagents on inks, photographed under ambient light and luminescent mode.

4.3.1.2 Ambient temperature processing conditions

Figure 4.3 shows that fingermarks are fully developed after 24 hours following ambient temperature processing conditions. However, weaker fingermark deposits (5th depletions) showed differences in luminescence even after an extended time period of 4 days. This is likely due to the rate of reaction to form fluorophoric Joullie's pink being slower compared to using a heat press. Therefore, it is important to exercise caution when using ambient temperature, as weak fingermarks may be undetected due to reduced sensitivity. It is possible that using elevated outside ambient temperature (> 25 °C) in tropical countries like Seychelles could provide better sensitivity, but further investigation is necessary.



Figure 4.3: Comparison of fingermarks treated with SPF IND/Zn and processed with heat press at 160 °C (left) and ambient temperature of 25 °C (right) on copy paper.

4.3.1.3 Simple comparative assessment

A comparison of quartered fingermarks indicated that the SPF formulation was effective when compared to the three active IND/Zn formulations on both substrates. No background staining or ridge diffusion was observed, as shown in Figure 4.4. Based on the preliminary investigation results and findings from Chapter 3, the SPF IND/Zn formulation was brought forward for laboratory trials.



Figure 4.4: Comparison of quartered fingermarks on blue notepad paper, developed with SPF IND/Zn (top left), NCFS IND/Zn (top right), and BKA IND/ZN (bottom left) and CAST IND/Zn (bottom right).

4.3.2 Laboratory trials

4.3.2.1 Comparative assessment

The comparative assessment of SPF IND/Zn to NCFS generated a total of 1200 fingermark halves, providing significant data points for analysis. Figure 4.5 shows the overall fingermark grades and clearly indicates that the SPF formulation is equally sensitive and effective as NCFS IND/Zn, developing 50.7 % and 51.2 % useful fingermarks, respectively. The detection rate for both formulations was lower when compared to the head-to-head comparison of IND/Zn formulations in Chapter 3. This was attributed to the addition of newspaper as a difficult substrate.



Figure 4.5: Percentage of fingermarks graded (3&4), (1&2), and (0) out of 1200 fingermark halves developed with SPF and NCFS IND/Zn formulations on five different paper substrates.

Two main trends were observed in the grades of fingermarks developed between the different substrates used (Figure 4.6). High recovery trends were observed on white copy paper (Premium and Recycled) and notepad paper, while cardboard and newspaper gave low recovery rates. Both formulations were least effective on newspapers, due to the low-quality paper and fluorescent background. Newspaper is produced using 100% recycled fibres through a mechanical pulping method, which differs from the chemical process employed in paper production.^{249, 250, 274} The mechanical treatment of the pulp results in shorter fibre lengths, decreasing the paper's smoothness. Additionally, printed papers like newspapers can exhibit high fluorescence, making any fingermarks on these pages hard to discern from the background due to the poor contrast between the fingermark and substrate, as depicted in Figure 4.8e. Consistent results were observed on cardboard for both treatments. Nevertheless, the performance was inferior when compared to the head-to-head comparison in Chapter 3, particularly for grades 0 and 1&2. This may be due to the deposition pressure, which is evident from the consistent results between SPF and NCFS, given splint fingermarks were used. Overall, SPF had similar trends to NCFS IND/ZN, indicating that SPF performs comparably to NCFS IND/Zn on each type of substrate.



Figure 4.6: Comparison of percentage fingermarks grades across five paper substrates, developed with SPF and NCFS IND/Zn formulation.

The two formulations also showed similar effectiveness on samples that were 1 week and 4 weeks old. Figure 4.7 shows a comparable distribution in fingermark grades with a slight reduction in the percentage of useful fingermarks developed for the 4week samples. This difference is not considered significant as it was expected that the triangle of interaction between fingermark deposit, substrate, and environment over this period could cause slight changes in the fingermark composition. More importantly, the results indicate that the SPF formulation is effective on both 1 week and 4 weeks samples, covering a timeframe during which most casework samples are likely to be processed in Seychelles.



Figure 4.7: Percentage of fingermarks graded (3&4), (1&2), and (0) developed for 1 week and 4 weeks samples, treated with SPF and NCFS IND/Zn formulation.

Examination of ridge details and quality of fingermarks developed revealed that those developed using the SPF formulation showed slightly stronger luminescence compared to the NCFS IND/Zn (Figure 4.8). It is worth noting that both formulations have similar concentrations of 1,2-indanedione. Hence, the difference in luminescence is likely due to the SPF formulation having a more polar co-solvent, as discussed in section 3.3.2. The observed difference was more noticeable on cardboard and newspaper compared to white copy paper. On notepad paper, a slightly brighter background was also observed with the SPF formulation. However, these variations did not affect the quality of the developed fingermarks.



Figure 4.8: Examples of fingermark developed with NCFS IND/Zn (left half) and SPF IND/Zn (right half) on (a) COS Premium white copy paper, (b) OfficeMax recycled white copy paper, (c) JBurrows blue notepad, (d) PPS cardboard, (e) The West Australian newspaper.

4.3.2.2 Incidental fingermarks

A trial involving incidental fingermarks enabled the performances of the SPF and NCFS IND/Zn formulations to be compared across a range of substrates handled 'realistically' through a simulated exercise. Figure 4.9 shows the number of useful fingermarks developed across the various substrates and reveals that both formulations are highly effective at developing large numbers of incidental fingermarks. Some well-developed fingermarks were not graded as useful due to overlap of ridge impressions as shown in Figure 4.10. As expected, the highest number of fingermarks were developed on the envelope due to being the substrate that was the most handled during the exercise. Most fingermarks were concentrated at the edges of the substrates, which is associated with the activity,^{271, 275} as most donors handled the edge of the substrate while inspecting it. A similar quality of fingermarks was developed by both formulations and no ink diffusion was observed on the blue ruled notepad paper. The results based on developing incidental fingermarks on 100 items examined on both sides provide valuable insight into its operational suitability and reinforce the comparable performance between the two IND/Zn formulations observed during the laboratory trials.



Figure 4.9: Comparison of total number of useful incidental fingermarks detected with SPF and NCFS IND/Zn treatment on five different substrates.



Figure 4.10: Examples of incidental fingermarks developed with NCFS IND/Zn (left) and SPF IND/Zn (right half) on (a & b) white plain envelope, (c & d) JBurrows blue notepad. Fingermarks were deposited by the same donor on each type of substrate.

4.3.2.3 Environmental considerations

The conditions of the two storage locations registered on the data logger are provided in Appendix B. The air-conditioned location registered a relatively constant temperature of 21 °C and a relative humidity (RH) between 33 % to 67 %, while the non-air-conditioned storage registered a range of temperatures from 14-32 °C and 30-81 % RH. SPF IND/Zn treatment was effective on both samples despite the differences in temperature and RH between the two locations (Figure 4.11 and Figure 4.12). However, slight differences were noted in the quality of fingermarks and background development for samples that were 3 months old, as depicted in Figure 4.13. Fingermarks stored under a non-air-conditioned environment appear fainter with a darker background.

These differences were attributed to the more extreme conditions of the non-airconditioned environment impacting the fingermark deposit and substrates physiochemical properties. Although amino acids are well preserved in the paper matrix,⁸² exposure to cyclic humidity and temperature can cause gradual diffusion of amino acids over time. It has been suggested that water-soluble components such as urea and chlorides can migrate faster under high relative humidity,¹⁴ although this process would be expected to occur at a slower rate for amino acids. Additionally, changes in the natural moisture content of the paper may have a combined effect on the diffusion mechanism and the composition of the paper. It has been demonstrated that various cycle conditions of humidity and temperature change the hydrogen bonding network in paper and its capacity for moisture absorption and desorption.²⁷⁶ These physicochemical changes over time are likely to impact the diffusion of water-soluble components of latent fingermark residue, including amino acids.

The findings indicate that SPF IND/Zn is effective in high temperature and humidity conditions similar to those in Seychelles for samples less than 3 months old. The consistent performance of SPF IND/Zn on samples stored under two different environmental conditions suggests that the variation in samples over 3 months old is more likely due to the environment rather than the IND/Zn reagent.

117



Figure 4.11: Comparison of fingermarks on white copy paper stored under non-air-conditioned and air-conditioned environments and treated with SPF IND/Zn.



Figure 4.12: Comparison of fingermarks on cardboard stored under non-air-conditioned and airconditioned environments and treated with SPF IND/Zn.



Figure 4.13: Comparison of fingermark ridge details on recycled copy paper stored under an airconditioned environment (left), non-air-conditioned environment (right) and treated with SPF IND/Zn.

4.3.2.4 Reagent stability

Investigation into the SPF working solution over 6 months suggested it to be stable, developing a comparable percentage of useful fingermarks as fresh SPF solutions (Figure 4.14). However, the effectiveness decreased for samples developed with 3 months old reagent. This is possibly due to fingermark variability, since split fingermarks were used, and both fresh and aged SPF formulations showed a decrease in effectiveness. Additionally, the SPF working solution gradually became colourless, as seen in Figure 4.1, indicating possible chemical interactions. Further investigation was performed using visible spectroscopy to assess changes in the 1,2-indanedione peak absorbance.





4.3.2.5 UV-vis spectroscopy

Examination of the spectra of each working solution shows a small peak at 490 nm for fresh working solutions, which correlates with 1,2-indanedione absorption.¹⁷ However, this peak gradually decreased in size as the solution aged (Figure 4.15). 1,2-indanedione is likely responsible for the faint-yellow colour, and the hypochromic effect has been suggested to be due to the formation of a hemiketal structure in the presence of ethanol, which makes the structure less conjugated.^{127, 228}

In section 3.3.3, it was observed that a decrease in 1,2-indanedione improves luminescence. However, this effect is only expected to reach a threshold point. Any further decrease in 1,2-indanedione would result in a decrease in luminescence. The SPF formulation containing 594 mg/L of 1,2-indanedione appears to remain above this threshold for working solutions aged more than 6 months. This is based on the comparative performance with fresh working solutions. Therefore, a shelf-life of 6 months is suggested for the SPF formulation. However, it must be noted that working solutions should be made when required and not stored for long durations since it is known that 1,2-indanedione is not stable in non-polar solvents and can interact with other solvents to form compounds that do not visualise latent marks.



Figure 4.15: Absorption spectra for SPF IND/Zn working solutions of less than 7 days old, 6, 12 and 18 months since preparation, showing the decrease in 1,2-indanedione peak absorbance.

4.3.3 Statistical Analysis

A Mann-Whitney U test performed on the comparative assessment data showed that there is no significant difference in the fingermark grades between SPF and NCFS IND/Zn treatments (p = 0.56). Similarly, a chi-squared test showed that there is no significant difference in the distributions of fingermark grade frequencies between the two treatments when performed on comparative assessment data (p = 0.87) and also when assessing the distributions of useful fingermark frequencies for the incidental fingermark data (p = 0.58). Additionally, the difference in proportion test showed no significant difference in the proportion of useful fingermarks (grade 3&4) developed by SPF and NCFS IND/Zn treatments in the laboratory trial data (p = 0.86).

The statistical findings reinforce the above results, demonstrating that the developed SPF IND/Zn is as effective as active formulations used by different police agencies around the world. The R code and results of statistical analysis are provided in Appendix A-3.

4.4 Conclusions

In this chapter, the developed SPF formulation was assessed by comparing it to the NCFS formulation. The assessment was carried out through laboratory and incidental fingermarks trials across five substrates, two ageing periods of 1 and 4 weeks, and a range of ten donors in line with the IFRG guidelines.²⁴ The findings demonstrated the comparable performance between SPF and NCFS formulations, developing 50.7 % and 51.2 % of suitable fingermarks, respectively. Both formulations exhibited high sensitivity and effectively developed fingermarks that were 1 week and 4 weeks old. Comparison of incidental fingermarks on 100 items gave similar results and statistical analysis reinforced comparable performance. The study concluded that the SPF formulation is stable for at least 6 months, despite spectroscopy studies showing a gradual decrease in 1,2-indanedione peak absorbance over an 18 month period.

This study demonstrated the successful approach of modifying existing techniques within the frugal forensics framework. As a result, this can increase accessibility and facilitate the use of latent fingermark methods in operational laboratories to improve the recoverability of latent fingermarks. The conclusions above must be qualified with the recognition that the studies were conducted on limited number of fingermark samples and substrates.

Chapter 5. Modification of WET UCIO powder suspension for fingermark detection on pressure sensitive adhesive tape

Portions of this chapter to be submitted for publication in the following article:

Jemmy T. Bouzin, Georgina Sauzier, and Simon W. Lewis. Modification of Wet UCIO powder suspension for fingermark development on adhesive tapes for use in Seychelles. *Forensic Chemistry* **2023**. To be submitted.

5.1 Introduction

This chapter explores the adaptation of a carbon-based powder suspension for latent fingermark detection on non-porous pressure-sensitive adhesive (PSA) tape. The study demonstrates how the frugal forensics concept can be applied to a recently published method.

PSA tape is a commonly accessible everyday material that is often the subject of forensic analysis. PSA tapes are used in crimes to immobilise or gag victims in kidnapping and homicide cases, seal drug packaging, and construct improvised explosive devices.^{160, 277-279} When handled for any purpose, the sticky-side surface of the adhesive tape serves as a substrate for trace evidence such as fingermarks²⁷⁹⁻²⁸¹. However, the detection of these fingermarks can be challenging due to the intrinsic adhesive nature of PSA tapes. Moreover, the cost of conventionally recommended fingermark detection techniques may limit their accessibility in some jurisdictions.

PSA tapes generally have two primary components: a pressure-sensitive adhesive coated on a backing material. Acrylic and rubber-based (natural or synthetic) forms are the most common adhesives.²⁸² ²⁸³ Acrylic PSAs are generally manufactured through polymerisation of acrylate monomer with the general structure shown in Figure 5.1a. Various monomers are used, such as methacrylate (R'= methyl group), 2-ethylhexyl acrylate (R' = 2=ethyl hexyl), and copolymer blends.²⁸² Natural rubber PSA can be based on natural rubber with composition largely corresponding to cis-1,4-polyisoprene; however, due to the high production cost, most are made from synthetic rubber such as poly-cis-1,4-butadiene and styrene-butadiene copolymers.²⁸³ The chemical structures of the polymers are shown in Figure 5.1b-d.



Figure 5.1: The molecular structure of materials used in the adhesive of PSA tapes, a) acrylate monomer, b) natural rubber polyisoprene, c) synthetic rubber poly-cis-1,4-butadiene, d) synthetic rubber polystyrene-butadiene.

Depending on the tape application, various backing materials are used. These backings include paper for easy-to-tear tape, polyester for high-strength tapes, and other plastic polymers (Figure 5.2) such as polypropylene, polyethylene, and polyvinyl chloride (PVC).²⁸² Modifying additives, such as tackifier resins, plasticisers and fillers, may also be incorporated.^{283, 284} For example, PVC insulating tape typically contains phthalate ester such as diisononyl phthalate (Figure 5.3) as a plasticiser to impart flexibility along with inorganic materials such as titanium dioxide, calcium carbonate, barium sulfate, talc, kaolin, and antimony oxide; used as fillers, flame retardants, or stabilisers.²⁸⁴ The variation in chemical composition and the intrinsic adhesive nature, which is key to pressure-sensitive tape functions, makes fingermark development without background surface staining challenging.



Figure 5.2: The molecular structure of materials used in the backing of PSA tapes, a) polypropylene, b) polyethylene, c) polyvinyl chloride.



Figure 5.3: The molecular structure of diisononyl phthalate, a common plasticiser used in adhesive tape to impart flexibility.²⁸⁵

Powder suspension is an effective and recommended method for developing latent fingermarks on the adhesive side of non-porous tapes.¹⁶ It consists of a powder suspended by a surfactant in water to form a paste that is then lightly painted over the adhesive tape using a brush, followed by washing with tap water. The effectiveness of powder suspensions has been reported to depend on several factors. The surfactant critical micelle concentration (c.m.c), a concentration above which surfactant micelle structures are spontaneously formed,²⁸⁶ has been shown to influence powder suspension effectiveness. In a study by Downham *et al.*¹⁶⁶, a formulation containing 1 x c.m.c of Triton X-100 surfactant was found to cause background staining, whilst formulations with 2 x c.m.c exhibited discriminate powder deposition on fingermark, suggesting the possible role of surfactant micelles in the powder deposition mechanism.

Other factors include the physio-chemical properties of the PSA tape surface. The variation in chemical composition can have a direct impact on the effectiveness of powder suspension. Studies by Bacon et al.²⁸⁷ using carbon-based suspension powder on various polymer-based substrates, found that titanium dioxide pigments in polymer substrates cause surface-wide background staining. This was not observed for similar polymer substrates without titanium dioxide pigments or where the pigments were 30 nm below the surface. Additionally, the type of powder has also been suggested to play a critical role. In a study on different samples of iron oxide (II/III) powders with varying distributions of size, Downham et al.¹⁶⁴ reported better performance with powder samples exhibiting higher sub-micrometre particle populations. This suggests that the particle size and shape likely influence powder suspension performance. As a result of the variation, various types of black and white powders have been trialled, leading to several pre-mixed commercial proprietary powder suspension formulations. Black powder suspension formulation is either carbon or iron oxide-based. Commercial carbon-based formulations include Wetwop[™] Black from Lightning Powder Company¹⁷¹ and Wet Powder Black from Kjell Carlsson,¹⁷² whereas Adhesive Side Powder Dark from Sirchie¹⁷³ is iron-oxide-based. White titanium dioxide-based powder suspension is also available for treating darkcoloured non-porous adhesive tape.¹⁷⁴

Carbon-based powder suspension is currently the recommended method for treating light-coloured adhesive side of non-porous tape.¹⁶ However, its operational use may be limited in small laboratories with restricted budgets due to its high cost (Wetwop[™] costs \$340 USD per litre). Another consideration when using a pre-mixed product is the general lack of control over formulation quality and shelf-life.¹⁶⁸ This has led to interest in developing cheaper in-house alternatives. A recent example is the development of Wet UCIO powder suspension,¹⁷⁵ consisting of carbon black Sirchie powder mixed with a commercial Gran Velada surfactant solution. This formulation has been reported to be more effective than commercial Wetwop[™] on eight types of adhesive tape.¹⁷⁵ while only costing c.a \$64 USD per litre (at least five times cheaper than Wetwop[™]) (Figure 5.4).



Figure 5.4: The products and cost of Wet UCIO and two commercial powder suspension formulations Wet Powder black Wetwop™ black.

However, the Gran Velada surfactant solution is not readily accessible outside Europe,²⁸⁸ limiting its use in Seychelles or other non-European jurisdictions. Nonetheless, the commercial Gran Velada surfactant solution contains 27 % w/v of sodium dodecyl sulfate (SDS),¹⁷⁵ an anionic surfactant consisting of a hydrophobic alkyl tail of 12 carbon atoms attached to a hydrophilic sulfate group (Figure 5.5).



Figure 5.5: The molecular structure of sodium dodecyl sulfate (SDS).

SDS is a low-cost and common surfactant used in various detergent, cosmetic, pharmaceutical and food products.²⁸⁹⁻²⁹² Given the accessibility, cost-effectiveness and low toxicity,²⁹³ SDS presents a sustainable alternative to the Gran Velada surfactant solution. Lab-grade SDS is available in different forms, including aqueous solutions of concentration 10 % or 20 % w/v. Another option is lab-grade SDS powder to produce in-house surfactant solution. The latter offers better transportation, storage, and quality control. Additionally, preparing fresh or small-volume stock solutions, as and when required, provides better shelf-life and reduces wastage, which aligns with a frugal forensics approach.

SDS is relatively hydrophilic with a water solubility for commercial salts ranging between 130 to 150 g/L at 20 °C,^{293 294} and a generally accepted c.m.c of 8.1 mM at 25 °C.²⁹⁵⁻²⁹⁸ To obtain a high SDS surfactant concentration similar to that of Gran Velada solution (27 % w/v), organic solvents, (e.g. ethanol, glycols or glycol ethers) are typically used as additives in the composition.²⁹⁹ The use of organic solvent has been shown to influence the micellisation process and wettability properties of the surfactant,²⁹⁸⁻³⁰¹ which may impact the powder suspension effectiveness.

This study aims to enhance the accessibility of the WET UCIO powder suspension by exploring readily available SDS salts to produce an in-house surfactant solution as a substitute for the Gran Velada solution. The first part of the study involves characterisation of pressure-sensitive tapes using ATR-FTIR analysis to assess and consider the surface's chemical composition and possible impact. The second consists of modifying the WET UCIO formulation and comparative assessment, which can be categorised into three stages.

- Stage A: Formulation modification and simple comparison to WET UCIO powder suspension.
- Stage B: Supplementary studies investigating the role of additives and background staining on insulating tapes.
- Stage C: Comparative assessment of modified formulation with commercial
 Wetwop[™] black powder suspension.

5.2 Experimental

5.2.1 Experimental methods

5.2.1.1 Powder suspension preparation

The preparation of the various powder suspension formulations used in this study (Wet UCIO, Wetwop[™] black, Wet Powder black, and modified Wet UCIO formulations) are described in section 2.5.2. The composition and approximate c.m.c of the modified formulations are specified in Table 5.1. In this study, Wetwop[™] black and Wet Powder black is referred to as Wetwop[™] and Wet Powder unless otherwise stated.

Table 5.1: Composition and approximate c.m.c of in-house surfactant solution used in this study. The

 c.m.c of SDS was taken to be 8.2 mM.²⁹⁷

Designation	Curfectory colution composition	SDS c.m.c
Designation	Surfactant solution composition	(approx.)
1 % SDS	1 g SDS in 100 ml deionised water	4 x c.m.c
5 % SDS	5 g SDS in 100 ml deionised water	20 x c.m.c
10 % SDS	10 g SDS in 100 ml deionised water	40 x c.m.c
27 % SDS	27 g SDS in 5 ml ethanol and 95 ml deionised water	100 x c.m.c
15 % SDS (Wet SPF)	15 g SDS in 5 mL ethanol and 95 mL deionised water	60 x c.m.c
10 % Ethanol	15 g SDS in 10 mL ethanol and 90 mL deionised water	
5 % PEG-400	15 g SDS in 5 mL PEG-400 and 95 mL deionised water	
10 % PEG-400	15 g SDS in 10 mL PEG-400 and 90 mL deionised water	

5.2.1.2 Substrates

Twelve PSA tapes of various brands and types were purchased in new condition for this study (Table 5.2). Investigation primarily made use of six tapes (#1 to #6), with additional insulating tapes (#7 to #12) subsequently purchased for supplementary studies, investigating background staining observed on this type of tape. Before sample preparation, the length circumference of each tape was discarded to ensure a clean surface for IR (infrared) analysis and fingermark deposition.

Table 5.2: Details of tapes used in this study.

Sample	Τορο Τγρο	Brand	Colour	Sourco	
Code	таре туре	brand	COIOUI		
Tape #1	Packaging	Scotch 3M	Transparent	Officeworks	
Tape #2	Packaging	Scotch 3M	Transparent	Officeworks	
Tape #3	Packaging	Bear	Brown	Bunnings	
Tape #4	Duct Tape	Paint Partner	Silver	Bunnings	
Tape #5	Masking	PPS	White	Officeworks	
Tape #6	Insulating	Nitto	Yellow	Bunnings	
Tape #7	Insulating	Click	Red	Bunnings	
Tape #8	Insulating	Click	Blue	Bunnings	
Tape #9	Insulating	Click	Black	Bunnings	
Tape #10	Insulating	Click	White	Bunnings	
Tape #11	Insulating	Deta	Yellow	Bunnings	
Tape #12	Insulating	Nitto	White	Bunnings	

5.2.1.3 ATR-FTIR Tape Analysis

The backing and adhesive of each PSA tape was analysed using Thermo Scientific Nicolet iS50 Fourier Transform Infrared (FTIR) spectrophotometer with a single-bounce attenuated total reflectance (ATR). The methodology is described in section 2.6.4.

5.2.1.4 Fingermark collection, photography, and grading

Tape samples of approximately 20 cm strips were cut, and acetate backing tabs were placed at each end on the adhesive side for labelling and handling purposes. The tape strips were secured with the adhesive side facing up on a corflute sheet with a background grid to indicate position to deposit fingermarks. Each tape sample consisted of a set of five fingermark depositions. Two samples of the same tape were secured side by side for split fingermark deposition as illustrated in Figure 5.6.



Figure 5.6: Schematic of latent fingermark deposition on tape samples.

Fingermarks were collected from a range of ten donors (four females and six males aged between, 22-60) following procedure outlined in section 2.3.2. The number of donors and tapes used for each experiment are specified in Table 5.3. Samples were developed following the procedure outlined in section 2.5.2. Visual examination and photography are outlined in section 2.6.1 and grading of fingermarks is described in section 2.7.

Experiment stage	Experiment type	Number of Donors	Tapes used	Powder suspension used
А	SDS solubility	5	#1, #2	1% SDS, 5% SDS, 10% SDS, 27% SDS
	Wet SPF formulation	3	#1, #2	Wet SPF, 27% SDS
В	Comparison Wet SPF to Wet UCIO	5	#2, #4, #6	Wet SPF, Wet UCIO
С	Role of additives	5	#3, #5, #6, #7	Wet SPF, Wet SPF - 10 % Ethanol, Wet SPF - 5 % PEG- 400, Wet SPF - 10 % PEG- 400
	Insulating tape background staining	5	#6, #7, #8, #9, #10	Wet SPF, Wet UCIO, Wet Powder
D	Comparative assessment Wet SPF to Wetwop	10	#2, #3, #4, #5, #6. #10. #11. #12	Wet SPF, Wetwop
-	Sensitivity test	5	#1, #4, #7	Wet SPF, Wetwop

Table 5.3: Details of the number of donors, tapes and powder suspension formulations used in each experiment.

5.2.2 Stage A – Formulation modification

Experiments were conducted to explore the effectiveness of modified WET UCIO powder suspension prepared with SDS concentration below the water solubility value, specifically 1 %, 5 %, and 10 % SDS surfactant solutions. Due to limited Gran Velada SDS surfactant solution available for this study, an in-house modified Wet UCIO powder suspension produced from 27 % SDS was used for the preliminary comparison. To achieve 27 % w/v SDS concentration, the surfactant was dissolved in 5 % v/v ethanol/deionised water mixture. Each donor deposited four sets (one for each formulation) of single fingermarks on each tape (acrylic and rubber-based adhesive tapes).

Additionally, a modified Wet UCIO formulation produced from 15 % w/v SDS formulation, designated "Wet SPF" was compared with the 27 % w/v SDS formulation, using the same tapes as above. Each donor deposited two sets of split five-depletion series fingermarks on each tape to provide an initial insight into the sensitivity of this formulation.

The Wet SPF formulation was subsequently compared to WET UCIO in line with phase 1 of the IFRG guidelines. Each donor deposited two sets (one for each formulation) of five-depletion series fingermark on each tape (packaging-, duct- and insulating tapes).

5.2.3 Stage B – Supplementary studies

To investigate the effect of additives on powder deposition, the Wet SPF formulation was compared to three modified formulations with varying concentrations and types of additives; namely 10 % ethanol, 5 % PEG-400 and 10 % PEG-400 (Table 5.3). Each donor deposited four sets (one for each formulation) of single fingermarks on each tape (brown packaging, Nitto insulating, Click insulating, and masking tape).

Background staining was investigated using three different powder suspensions (Wet SPF, WET UCIO and commercial Wet Powder) on five insulating tapes. The same

fingermark procedure as above was followed, except that three sets of fingermarks were collected, one for each formulation.

5.2.4 Stage C - Comparative assessment to Wetwop™

An assessment was made of the relative performance of Wet SPF powder suspension with the commercial Wetwop[™], which is the current operational technique in Seychelles. The assessment was carried out using ten donors on eight adhesive tapes and two ageing periods (1 week and 4 weeks) in line with phase 2/3 of the IFRG guidelines.²⁴ Each donor deposited two sets (one for each ageing period) of single split fingemarks on each tape. Following fingermark collection, each strip was stuck to the backing of another strip of the same tape to simulate the condition of tape packages operationally encountered. The extra strip was removed before fingermark development.

To assess sensitivity, ten depletion series of natural split fingermarks were collected from each donor on three tapes. The samples were aged for 1 week following a similar procedure as above.

5.2.5 Statistical analysis

Statistical analysis software R version 4.3.0 was utilised to assess whether there was significant difference in the fingermark grades developed by different powder suspension formulation treatments. The non-parametric, Mann-Whitney U and chi-squared tests were used to assess the overall performance (n= 320) and sensitivity (n= 300) between Wet SPF and Wetwop[™]. Additionally, a difference in proportion test was also performed on both data sets to assess whether the proportion of grade 3s and 4s (useful fingermarks) of one treatment differs from the other. All statistical tests were carried out at 95 % confidence level (p < 0.05).

5.3 Results and discussion

5.3.1 Tape Characterisation

As discussed in section 5.1, the chemical composition of PSA tapes may affect powder suspension performance, hence characterisation of the tapes was carried out. The IR spectra obtained allowed the backing and adhesive of the 12 tapes used in this study to be grouped based on the main polymer present. The backing was divided into four groups: polypropylene, polyethylene, cellulose acetate, and PVC whereas the adhesive was differentiated into two groups, namely acrylic- and synthetic rubber-based polymer (Table 5.4). The characteristic peak assignment for the different polymers is provided in Table 5.5.

Table 5.4 Classification of adhesive and backing composition based on IR analysis of the main polymertype.

Sample	Tana Tuna	Prand	Colour	Adhesive	Backing	Sourco
Code	таре туре	Brand	Colour	Composition	Composition	Source
Tape #1	Packaging	Scotch 3M	Transparent	Acrylic	Polypropylene	Officeworks
Tape #2	Packaging	Scotch 3M	Transparent	Rubber	Polypropylene	Officeworks
Tape #3	Packaging	Bear	Brown	Acrylic	Polypropylene	Bunnings
Tape #4	Duct Tape	Paint Partner	Silver	Rubber	Polyethylene	Bunnings
Tape #5	Masking	PPS	White	Rubber	Cellulose	Officeworks
Tape #6	Insulating	Nitto	Yellow	Rubber	PVC	Bunnings
Tape #7	Insulating	Click	Red	Rubber	PVC	Bunnings
Tape #8	Insulating	Click	Blue	Rubber	PVC	Bunnings
Tape #9	Insulating	Click	Black	Rubber	PVC	Bunnings
Tape #10	Insulating	Click	White	Rubber	PVC	Bunnings
Tape #11	Insulating	Deta	Yellow	Rubber	PVC	Bunnings
Tape #12	Insulating	Nitto	White	Rubber	PVC	Bunnings

5.3.1.1 PSA tape backing

The spectra of the backing of packaging tape #1, #2 and #3 all show the characteristic polypropylene profile (Figure 5.7) with quadruplet absorption bands at 2950, 2868 cm⁻¹ and 2916, 2840 cm⁻¹ attributed to C-H stretching of CH₃ and CH₂ groups, peaks at 1452, 1377 cm⁻¹ attributed to C-H bending of CH₂ and CH₃ groups, and minor peaks at 1168, 974 cm⁻¹ assigned to C-C stretching and C-H rocking of CH₃ and at 840 cm⁻¹ assigned to C-H rocking of CH₃ groups.^{302, 303}

The spectrum of the backing of duct tape #4 shows the typical pattern of linear lowdensity polyethylene (LLDPE) (Figure 5.8a) with peaks at 2914 and 2846 cm⁻¹ attributed to C-H stretching and characteristic peaks of both high-density polyethylene (HDPE) pair of splitting peaks at 1468,1462 cm⁻¹ and 726,716 cm⁻¹ attributed to CH₂ in-phase and out-of-phase rocking, and low-density polyethylene (LDPE) single peak at 1379 cm⁻¹ (attributed to methyl umbrella mode from the side chain).^{304, 305}

Tape side	Polymer type	Absorption band (cm ⁻¹)	Assingnment	
		840	CH rocking	
		974	CH3 rocking	
		1168	C-C stretching	
	Polypropylene	1377	CH bending symmetrical	
		1452	CH bending symmetrical	
		2950, 2868 2916, 2840	C-H stretching of CH2 and CH3	
		726	CH2 rocking	
	Polyethylene	1468	CH2 scissoring	
Backing		2914	C-H stretching Symmetrical	
		2846	C-H stretching Symmetrical	
	Cellulose acetate	1026	C-O stretching	
		1230	C-O-C stretching	
		1377	CH2 deformation	
		1726	C=O stretching	
		3400 (weak)	O-H stretching	
	Polyvinylchloride	615, 698	Stretching	
		955	CH2 rocking	
		1316, 1252	C-H bending of CHCl	
		1423	CH2 bending	
		2915, 2850	CH2 stretching	
Adhesive		1160	 –C-O-C stretching asymmetrical 	
	Acrylate	1240	C-O stretching	
		1730	 – C=O stretching 	
		2958, 2930, 2872	C-H stretching	
		750, 695	C H deformation in ring (polystyrene)	
	Styrene-isoprene	1451, 1376	C-H deformation	
	copolymer	1600, 1490 (weak)	Ring vibrations	
		2956, 2930, 2854	C-H stretching	

 Table 5.5: PSA tapes characteristic IR peak assignment for PSA tapes backing and adhesive.³⁰²⁻³¹¹

The backing of masking tape #5 conforms to a cellulose acetate polymer (Figure 5.8b) with a characteristic weak broad peak at 3400 attributed to O-H stretching, peaks between 2920 and 2851 cm⁻¹ assigned to CH stretching, typical strong peak at 1726 cm⁻¹ assigned to C=0 stretch of the ester group, 1377 cm⁻¹ assigned to CH₂ deformation vibration, typical peaks 1230, 1026 cm⁻¹ attributed to C-O-C stretching of the acetyl group and C-O stretching of pyranose ring respectively.^{303, 306, 307}

The insulating tapes #6-12 were identified as PVC backing (Figure 5.9) with characteristic peaks at 2915 and 2850 cm⁻¹ assigned to CH₂ stretching, 1423 cm⁻¹ attributed to CH₂ bending, 1316 cm⁻¹ and 1252 cm⁻¹ to C-H bending of CH-Cl, 955 cm⁻¹ assigned to CH₂ rocking and peaks at 698 cm⁻¹ and 615 cm⁻¹ to C-Cl stretching.^{306, 307} Additional absorption bands were attributed to phthalates plasticiser, particularly weak doublet at 1591 and 1582 cm⁻¹ and stronger peak at 744 cm⁻¹ assigned to phthalate aromatic ring vibration. Other peaks include 1726 cm⁻¹ and 1120 cm⁻¹, 1068 cm⁻¹ attributed to the ester group.^{304, 308, 309}



Figure 5.7: ATR-FTIR spectra of polypropylene backing of packaging tape #1-3. Spectra have been offset for clarity.



Figure 5.8: ATR-FTIR spectra of (a) polyethylene backing of duct tape #4 and (b) cellulose acetate backing of masking tape #5. Spectra have been offset for clarity.



Figure 5.9: ATR-FTIR spectra of polyvinylchloride backing of insulating tapes #6-#12. Spectra have been offset for clarity.

5.3.1.2 PSA tape adhesive

The spectra of adhesive for packaging tape #1 and #3 show typical pattern of acrylicbased polymer (Figure 5.10), with peaks at 2958, 2930 cm⁻¹ and 2872 cm⁻¹ assigned to C-H stretching of CH₂and CH₃ groups, a characteristic strong absorption band at 1730 cm⁻¹ attributed to C=O stretching of ester group, a broad peak at 1240 cm⁻¹ and the strong absorption at 1160 cm⁻¹ assigned to stretching vibration of the ester group.^{303, 306}

The adhesive of packaging tape #2, duct tape #4, masking tape #5 and insulating tapes #6-12 were all identified as synthetic rubber (styrene-isoprene copolymer) (Figure 5.11), showing aliphatic absorption bands at 2956, 2930, 2854 cm⁻¹ attributed to C-H stretching, at 1451, 1376 cm⁻¹ assigned to C-H deformation, and aromatic diagnostic bands at 1600, 1490 cm⁻¹ assigned to ring vibration, and peaks at 750, 695 cm⁻¹ assigned to bending of aromatic =C-H and C=C groups of polystyrene.^{306, 310}



Figure 5.10: ATR-FTIR spectra of acrylic adhesive packaging tape #1 and #2.



Figure 5.11: ATR-FTIR spectra of synthetic rubber adhesive for packaging tapes #2, duct tape #4, masking tape #5 and insulating tape #6-#12. Spectra have been offset for clarity.

Compared to other synthetic rubber adhesive tapes, insulating tapes #6-12 displayed additional peaks in their spectra, suggesting possible contributions from the tape backing in the adhesive spectra. Figure 5.12 shows the overlay of the fingerprint region of both backing and adhesive spectra of synthetic rubber adhesive tapes. Examination of the spectra of packaging tapes #2, duct tape #4, and masking tape #5 showed no significant contributions from backing in the adhesive spectra. However, with all insulating tapes, striking similarity was observed between the tape and adhesive spectra, especially the carboxylate bands. This was attributed to the leaching of the plasticizer diisononyl phthalate into the adhesive glue, illustrated in Figure 5.13. Diisononyl phthalate possesses an ortho-substituted aromatic ring, which results in a strong absorbance at 745 cm⁻¹ associated with the ortho aromatic group. Additionally, the characteristic weak doublet peaks at 1600 cm⁻¹ and 1585 cm⁻¹ are present due to the aromatic ring quadrant stretching vibration.³¹¹



Figure 5.12: Overlay of the fingerprint region of both backing and adhesive IR spectrum of rubber adhesive tapes showing significant contribution of the backing in the adhesive spectrum of tape#6 to #12.



Figure 5.13: Diisononyl phthalate IR spectrum³¹² offset over rubber adhesive tape's backing and adhesive spectrum of tape #9. Showing the contribution of diisononyl phthalate in the backing and adhesive tape spectra.

A visual examination of the spectra showed that despite originating from three distinct manufacturers, the insulating tapes displayed high similarities with only slight variations in the fingerprint region as shown in Figure 5.14. Notably, the Nitto tapes #6 and #12 were distinguishable from the other brands based on these differences in chemical composition, which are attributed to titanium dioxide and carbonates pigments.^{304, 309, 313} As discussed earlier these differences may cause variation in the effectiveness of powder suspension between the different tapes.



Figure 5.14: Comparison of infrared spectra of different insulating tape synthetic rubber adhesive. The main differences are highlighted (blue shaded regions).

5.3.2 Formulation modification

The water solubility of SDS is generally reported in the range of 130 to 150 g/L (13 % to 15 % SDS w/v) at 20 °C.²⁹³ ²⁹⁴ Preparation of in-house 27 % w/v SDS surfactant water mixture (equivalent concentration to the Gran Velada solution) formed crystalline structures a day after preparation as shown in Figure 5.15. This indicates that the Gran Velada surfactant solution likely contains an additive to bring the high concentration of SDS into solution. However, this was not stated on material and safety data sheet.³¹⁴


Figure 5.15: Sodium dodecyl sulfate solutions (a) insoluble solution of 27 % w/v in deionised, (b) clear solution of 15 % w/v in 5 % ethanol deionised water mixture.

Powder suspensions prepared with differing concentrations of SDS were thus investigated, with the results presented in Figure 5.16. Formulations containing 5 % (~20 x c.m.c) SDS surfactant concentration and above were effective in developing fingermarks with grades 3&4 (useful fingermarks). In contrast, no useful fingermarks were developed with powder suspension prepared from 1 % SDS (~ 4 x c.m.c). The variation in fingermark quality with SDS concentration was observable when examining the degree of powder deposition, as shown in Figure 5.16b. The powder suspension with 1 % SDS exhibited low contrast due to high background staining and minimal powder deposition on the fingermarks. Increasing the SDS concentration reversed the trend, with the 27 % SDS powder suspension resulting in high deposition on the fingermarks and a relatively clean background. This pattern was observed on both acrylic and rubber-based tapes but was more noticeable on the former. The result underlines the critical role that surfactant concentration plays in powder suspension effectiveness and is consistent with a previous study by Downham et al.,¹⁶⁵ which showed that using Triton X-100 surfactant in an iron oxide powder suspension requires a surfactant concentration above the critical micelle concentration (2 x c.m.c) to prevent indiscriminate staining. This study indicates that a surfactant concentration above 4 x c.m.c of SDS is necessary for carbon-based powder suspension. Compared to the non-ionic Triton X-100, the ionic nature of SDS surfactant likely increases its interaction with the background surface, resulting in a greater disruption of the surfactant micelles, hence necessitating a higher concentration.



Figure 5.16: Comparison of (a) the number and grades of fingermarks developed (b) the degree of powder deposition on fingermarks and background after treatment with modified WET UCIO powder suspensions prepared with 1 %, 5 %, 10 % and 27 % w/v SDS surfactant solution.

Whilst 27 % SDS formulation developed highest quality fingermarks, the difference with the 10 % SDS formulation was minor. Moreover, 27 g of SDS to produce 100 mL of surfactant solution was a sizeable amount given the low density of SDS (1.01 g/cm³).³¹⁵ Additionally, reducing the surfactant concentration would be more cost-effective and relatively less likely to precipitate over time. Therefore, a 15 % w/v surfactant solution containing 5 % v/v ethanol/deionised water mixture (Wet SPF) was investigated. Ethanol, a readily available laboratory reagent, was used as an additive to avoid precipitation of SDS. Wet SPF was found to be equally effective and sensitive in developing fingermarks as the formulation containing 27 % SDS, on both acrylic- and rubber-based PSA tapes as shown in Figure 5.17.



Depletion 1

Depletion 3

Depletion 5



The Wet SPF was then compared to Wet UCIO and both formulations were found to be highly effective on packaging tape #2 and duct tape #4, with all fingermarks graded as useful (Figure 5.18a). Overall, the percentage of useful fingermarks developed was 68 % for Wet SPF and 60 % for WET UCIO, slightly lower than the 86.5 % effectiveness reported by Claveria *et al.*¹⁷⁵ using WET UCIO on five rubber-based adhesives. The reduced effectiveness was attributed to the indiscriminate background staining observed on Nitto insulating tape #6 (Figure 5.18b). This is likely due to the variations in substrate chemical composition, as a result of titanium dioxide in the adhesive tape, as indicated by the visual examination of the IR spectra. Pigments such as titanium dioxide have been reported to cause widespread background staining with carbon-based powder suspension.²⁸⁷

Another observation was that the powder deposition was more intense with Wet SPF than with WET UCIO formulation (Figure 5.18b). This may explain why 24 % of the fingermarks developed with WET UCIO on insulating tape #6 were undetected (grade 0). In contrast, all fingermarks were detected with Wet SPF formulation, but a high percentage were graded as not useful, due to low contrast caused by background staining. Although subtle, the differential powder deposition on fingermarks was also observed on packaging tape #2 and duct tape #4 with Wet SPF formulation exhibiting darker fingermarks compared to WET UCIO. One possible cause is the difference in SDS concentration. Wet SPF has a concentration of 15 % w/v while WET UCIO has 27 % w/v. During these experiments, no significant differences were observed between

the two SDS concentrations, indicating that other factors, such as the presence of additives, may be influencing the results.



Figure 5.18: Comparison of (a) percentage of fingermarks graded (3&4), (1&2), and (0) out of 150 fingermarks (b) the degree of powder deposition on fingermarks and background after treatment with WET UCIO and Wet SPF powder suspensions on packaging tape, duct tape and insulating tape.

5.3.3 Supplementary studies

To investigate the role of additives, a set of modified Wet SPF formulations was evaluated using ethanol and PEG-400 as additives. Both additives gave similar results in terms of the number and quality of developed fingermarks across four tapes (Figure 5.19). This indicates that the two additives appeared to have minimal impact on the effectiveness of Wet SPF formulation, although this does not refute the possibility that the additive in Wet UCIO may cause a minor reduction in powder deposition. In line with frugal forensics, ethanol is a readily available and economical laboratory reagent, almost half the price of PEG-400. Nonetheless, the results suggest that PEG-400 is equally effective as an alternative additive.



Figure 5.19: Comparison of fingermark ridge details and contrast developed on four different tapes after treatment with Wet SPF powder suspensions prepared with different additives; 5 % ethanol, 5 % PEG-400, and 10 % PEG-400 v/v.

Investigation into the five insulating tapes revealed that the background staining was specific to Nitto tape #6, with all formulations exhibiting similar results. The IR analysis showed that all insulating tapes were of PVC backing and rubber-based adhesive, but Nitto tapes showed slight difference in chemical composition compared to other brands. This strongly suggests that background staining is substrate related and more likely due to additives in the adhesive as discussed above. Additionally, the earlier observation that powder deposition on fingermarks was higher with Wet SPF formulation, was also seen when compared to the commercial Wet Powder formulation. The findings suggest that Wet SPF may be more effective than Wet Powder in developing weaker fingermarks on rubber-based insulating tapes.

Interestingly, the insulating tape #9 used in this study was black-coloured, which is generally processed with white powder suspension for better contrast. However, in this study, the carbon-black-based powder was used, and visualisation was achieved by exploiting the difference in absorbance between the treated fingermark and tape surfaces.¹⁷ The treated black-coloured tape was examined using a Polilight under

different excitation wavelengths and filters to obtain the best viewing conditions. The combination of 505 nm excitation and 550 nm filter was found to be optimum, as the visualised marks closely resemble black and white fingermarks, which are generally preferred by practitioners¹⁷ (Figure 5.20). The result indicates that carbon-based powder suspension may be used for light and dark tapes, offering a cheaper alternative to commercial white powder suspensions. However, further investigation is required on a larger subset of samples of various types of dark-coloured tapes.



Figure 5.20: Fingermark ridge details and contrast after treatment with Wet SPF, WET UCIO and Wet Powder on (a) Click white insulating tape #10 under white light, (b) Click black insulating tape #9 under 505 nm excitation and 550 nm filter.

5.3.4 Comparative assessment to Wetwop™

5.3.4.1 Overall performance

The comparative assessment of Wet SPF and Wetwop[™] powder suspension in developing 1 week and 4 weeks old fingermarks deposited on eight adhesive tapes by a range of 10 donors are summarised in Figure 5.21. In total, 320 split fingermarks were graded to provide insight into overall performance with Wet SPF formulation developing 68 % of useful fingermarks and Wetwop[™] 61 %. The result clearly shows the comparative effectiveness of Wet SPF to the commercial Wetwop[™] powder suspension on both acrylic and rubber-based PSA tapes.





The results for the two aged samples of 1 week and 4 weeks are presented in Figure 5.22. Both formulations appeared to be more effective on older fingermarks, with Wet SPF increasing from 65 % to 70 % and WetwopTM from 58 % to 64 %. The effectiveness of powder suspension on older fingermarks has been suggested to be due to the loss of the water-soluble eccrine component over time, exposing and reducing the interaction distance between the eccrine components encapsulated within the non-water soluble constituents and particles of powder suspension.¹⁶ The

result demonstrates the applicability of Wet SPF on aged fingermarks that are likely to be encountered in casework.

In terms of substrates, both formulations exhibited high effectiveness in developing fingermarks on acrylic and rubber-based PSA tapes (Table 5.6), with exception of Nitto rubber-based insulating tapes #6 and #12, and rubber-based masking tape #5. As expected, due to differences in chemical composition, Nitto insulating tapes #6 and #12 exhibited background staining, resulting in poor effectiveness. In contrast, both formulations were highly effective in developing useful fingermarks on similarly coloured insulating tapes #10 and #11 but from a different manufacturer. The background staining on Nitto insulating tapes has been discussed above and these results reinforced the suggestion that this is due to the chemical composition of the Nitto adhesive PSA tapes. Excluding the Nitto insulating tapes, Wet SPF and Wetwop[™] percentage effectiveness of Wet UCIO method.¹⁷⁵

On the rubber-based masking tape #5, Wet SPF and Wetwop[™] were found to be 40 % and 25 % effective, respectively. A similar effectiveness of 37.5 % was reported for rubber-based masking tape with WET UCIO powder suspension.¹⁷⁵ Examination of the developed fingermarks showed that despite a relatively clean background, poor contrast was observed since the space between ridges was partly stained (Figure 5.23). This is likely because during the drying process, the solution partitioned into the adhesive layer and diffuses in the cellulose matrix to stain the backing, reducing the overall quality of the developed marks.



Figure 5.22: Percentage of fingermarks graded (3 & 4), (1 & 2), and (0) out of 320 fingermarks developed with Wet UCIO and Wet SPF on eight different types of tape aged for 1 week and 4 weeks.

Table 5.6: Percentage of fingermarks	graded 3&4 on each o	of the eight types of	^c adhesive tape.
--------------------------------------	----------------------	-----------------------	-----------------------------

Sample Code	Tape Type and colour	Brand	Adhesive	Wot SPE	Wetwop
		Dialiu	Composition	WELSFI	
Tape #2	Packaging (Transparent)	Scotch 3M	Rubber	100%	85%
Tape #3	Packaging (Transparent)	Bear	Acrylic	100%	100%
Tape #4	Duct Tape (Silver)	Paint Partner	Rubber	100%	90%
Tape #5	Masking (White)	PPS	Rubber	40%	25%
Tape #6	Insulating (Yellow)	Nitto	Rubber	5%	0%
Tape #10	Insulating (White)	Click	Rubber	100%	95%
Tape #11	Insulating (Yellow)	Deta	Rubber	95%	90%
Tape #12	Insulating (White)	Nitto	Rubber	0%	0%



Figure 5.23: Examples of fingermark developed with Wet SPF (left) and Wetwop™ (right) on eight pressure sensitives tapes #2 Scotch 3M clear packaging tape, #3 Scotch 3M brown packaging tape, #4 Paint Partner silver duct tape, #5 PPS white masking tape, #6 Nitto yellow insulating tape, #10 Click white insulating tape, #11 Delta yellow insulating tape, #12 Nitto white insulating tape.

5.3.4.2 Sensitivity

A total of 300 split fingermarks comprising up to 10th depletion were processed to assess the sensitivity of Wet SPF to Wetwop[™]. Figure 5.24 shows the sensitivity results with grouping of 1st to 5th depletions and 6th to 10th depletions. The result indicates Wet SPF to show superior sensitivity particularly for the 6th to 10th depletion series. As discussed earlier, this is due to higher powder deposition on fingermarks exhibited by Wet SPF resulting in better effectiveness in developing weaker fingermarks.



Figure 5.24: Percentage of fingermarks grades out of 300 fingermark halves developed with Wet UCIO and Wet SPF on three different types of tape grouped into depletion series 1-5 and 6-10.

Regarding reagent stability, Wet SPF that has been prepared with SDS solution over 6 months old was found to be as effective as suspension powder prepared with fresh SDS solution. It has also been reported that aqueous SDS solution can remain stable for up to three years.³¹⁶ Although further studies are required to test the shelf life of the Wet SPF suspension powder mixtures, preparing the formulation fresh and when needed is recommended to avoid wastage, given its ease of preparation.

5.3.5 Statistical analysis

A Mann-Whitney U, chi-squared and difference in proportion tests performed on the overall performance data, showed no significant difference in the fingermark grades (p = 0.051), its distribution of frequencies (p = 0.057), or proportion of grades 3s and 4s (p = 0.2) between Wet SPF and WetwopTM. For the sensitivity data, all three tests revealed significant differences $(p = 0.2 \times 19^9, 0.7 \times 10^8, 0.2 \times 10^8 \text{ respectively})$. The statistical results reinforce the observation that Wet SPF's performance is comparable and shows better sensitivity to commercial WetwopTM black powder suspension. The R code and results of all statistical analysis are provided in in Appendix A-4.

5.4 Conclusion

This chapter demonstrates how the frugal forensics concept can be applied to a recently published method through the adaptation of a carbon-based powder suspension for latent fingermark detection on non-porous PSA tapes. Successful modification of the recently published WET UCIO powder formulation by replacing the inaccessible Gran Velada surfactant solution with an in-house sodium dodecyl sulfate surfactant solution has been shown to be effective in the recovery of latent fingermarks on PSA adhesive tapes.

The Wet SPF powder suspension containing 15 % SDS in a 5 % ethanol/water mixture has been shown to be as effective as the commercial Wetwop[™] powder suspension on eight different tapes using ten donors. Other benefits include an inexpensive formulation and readily available surfactant, accessible from multiple suppliers. In respect of remote-location jurisdictions, the use of salts compared to commercial solution surfactants allows for easy transportation, storage and better control of quality and shelf-life.

The use of Wet SPF combined with forensic light sources appeared effective in visualising fingermarks on challenging surfaces of dark-coloured adhesive tapes. Further studies on the comparative assessment with commercial white powder

suspension are suggested, as this may allow the use of a single formulation suited to both light- and dark-coloured substrates.

Based on this successful demonstration in line with IFRG phase 2/3 studies, the Wet SPF method is recommended for validation test as an operational technique that can be used in Seychelles for developing latent fingermarks on adhesive tapes. The formulation is as follows:

1.5 grams Sirchie silk black fingerprint powder
10 mL of 15 % sodium dodecyl sulfate solution
15 % sodium dodecyl sulfate solution is as follows –
15 grams of sodium dodecyl sulfate
100 mL of 5 % ethanol/deionised water mixture.

Chapter 6. Investigation into acid Nile blue as an emerging lipid-sensitive and post-cyanoacrylate detection method

6.1 Introduction

Chapters 3 and 4 investigated a well-established amino acid sensitive reagent for porous surfaces, and Chapter 5 a recently published method on non-porous surfaces. This chapter explores the benzophenoxazine dye Nile blue, as an emerging lipid-sensitive latent fingermark detection technique within the construct of a frugal forensics approach.

Nile blue, as a lipid-sensitive reagent has several benefits in a frugal forensics context. It is a non-toxic water-based solution that is simple to use, low-cost (ca. \$3.20 USD per gram)¹⁵³ and an easily accessible forensic method. This makes it a promising option for jurisdictions with limited resources.

As discussed in Chapter 1, Nile blue is a cationic dye used for staining neutral fats, phospholipids, and fatty acids in histology.¹⁴⁹ Its behaviour in water is influenced by its acid form, which can form hydrogen aggregates that do not fluoresce.^{142, 317} However, a small amount of Nile blue undergoes spontaneous hydrolysis to Nile red^{142, 143} as shown in Figure 6.1. Nile red is also used in histology for staining triacylglycerols, phospholipids, esteryl esters, and lipoproteins. It is a neutral molecule that is highly fluorescent, particularly in non-polar solvents.¹⁴²



Figure 6.1: Reaction pathway proposed for the spontaneous hydrolysis of Nile Blue to Nile red in aqueous solution.¹⁴⁶

The presence of the two stains creates a dual-purpose fingermark detection reagent. Nile red dissolves preferentially in neutral lipids,¹⁴⁸ producing photoluminescence when excited at 505 nm and viewed through a 550 nm long pass barrier filter.^{146, 318} Nile blue interacts with the acidic components, resulting in visible dark blue development and suppression of background fluorescence.¹⁴⁶ The properties of Nile red and Nile blue depend on the solvent polarity and dielectric constant. In non-polar media, they fluoresce brightly in the visible to near-infrared range, whilst in polar solvents, their fluorescence redshifts with lower intensity due to hydrogen bonding.^{142, 319}

Recent studies by Frick *et al.*¹⁴⁶ demonstrated aqueous Nile blue's effectiveness in developing lipid-rich fingermarks on a range of wet porous and non-porous surfaces. Further studies by Crocker *et al.*¹⁵⁴ have shown that modifying the aqueous Nile blue with sulfuric acid increases fingermark fluorescence intensity due to increased hydrolysis of Nile blue into Nile red. However, comparative assessment of acid Nile with an operational lipid-sensitive technique, such as Oil red O (ORO), is yet to be carried out. This can establish the feasibility of using Nile blue as an operational lipid-sensitive technique for porous surfaces.

Another potential application of acid Nile blue is as a post-cyanoacrylate stain. In an initial investigation conducted by Chester *et al.*,¹⁵¹ the use of Nile blue as a stain after cyanoacrylate treatment was found to have limited effectiveness. However, this was likely due to the short processing time of only 20 seconds. The underlying mechanism of post-treatment stain was not discussed, nor does it commonly appear in the literature. Other preliminary studies^{187, 188} compared Nile blue with other luminescent dyes including rhodamine 6G (R6G). However, this did not include Basic Yellow 40 (BY40), which is the current operational method used by the Seychelles Police for enhancement of cyanoacrylate fumed fingermarks.

Basic yellow 40 (7-(diethylamino)-3-(1,3-dimethylbenzimidazol-3-ium-2-yl)chromen-2-one;chloride) is a common basic fluorescent dye introduced as an alternative to R6G due to suspected carcinogenic risk of R6G.⁶⁸ BY40 generally refers to several dyes, of which the recommended dye for fingermark development is the chloride base (Figure 6.2). Recently there has been concern over the industry's lack of clarity around individual powders' specifications, due to reports of ethanolic-BY40 solution becoming cloudy.¹⁶⁸ These were based on a sulfate salt of BY40 dyes (CAS No: 35869-60-4) that does not dissolve completely. Moreover, the mechanism of post-treatment stains is poorly understood as it is unclear whether the stain partitions into the fingermark residue or only stains the polycyanoacrylate structure. One explanation is that the fibrous network of polycyanoacrylate works like a sieve, trapping the dye molecules. Another possibility is that the cyanide (CN⁻) anions of polycyanoacrylate may weakly interact with the cations of basic dyes through van der Waals forces.⁶⁸ A fundamental understanding of the staining process could direct the use of new dyes and the treatment sequencing process.



Figure 6.2: The molecular structure of Basic Yellow 40 (CAS Number 29556-33-0) recommended dye for fingermark enhancement.

This study has two main objectives, firstly to conduct a comparative assessment of acid Nile blue to ORO as a lipid-sensitive reagent on wet porous substrates. Secondly, to assess the potential of acid Nile blue as a post-cyanoacrylate fingermark stain and explore the underlying staining mechanism using confocal Raman microscopy. This will provide information on the frugal forensics use of Nile blue dye as a fingermark enhancement technique.

6.2 Experimental

6.2.1 Experimental methods

6.2.1.1 Reagent preparation and sample development

Acid Nile blue was prepared, and samples were developed as per Crocker *et al.*'s¹⁵⁴ method outlined in section 2.5.5. ORO was prepared and samples were developed as per the modified formulations in Frick *et al.*,¹³⁷ outlined in section 2.5.3.

Cyanoacrylate fumed prints were prepared as outlined in section 2.5.4. The samples were allowed to cure for 24 hours before post-cyanoacrylate staining with acid Nile blue or BY40 following the procedure in sections 2.5.5 and 2.5.6 respectively.

6.2.1.2 Substrates

For the comparative assessment to ORO study, four types of substrates commonly encountered by the Seychelles Police were used and are summarised in Table 6.1.

For the post-cyanoacrylate stain study, three types of substrates (glass, metal, and plastic) that are generally processed with cyanoacrylate were used and summarised in Table 6.2.

Table 6.1 : Detail of substrates used in the comparative assessment to ORO.	

Substrate	Description	Colour	Weight (gsm)/Thickness
1	Winc copy paper A4	White	80
2	OfficeMax copy paper A4, 50% recycled	White	80
3	PPS Brown kraft paper	Brown	50
4	Newsprint (The West Australian)	Grey	

 Table 6.2: Details of substrates used in the post-cyanoacrylate study.

Substrate	Туре	Description	Source
1	Glass	Borosilicate microscope slides 1 mm	VWR International, Leuven
2	Metal	Aluminium sheet 0.5 mm	Bunnings
3	Plastic	Polypropylene plastic lid	Red Dot

6.2.1.3 Fingermark collection

Single split fingermarks were collected from a range of six donors (three females and three males, aged 22-60) following procedure outlined in section 2.3.2. Samples were stored in cupboard in office space under ambient average temperature of 21 °C and RH of 56 %.

6.2.1.4 Sample visualisation and evaluation

Visual examination and photography are outlined in section 2.6.1 and grading of fingermarks is described in section 2.7.

6.2.2 As a lipid-sensitive reagent on porous substrates

6.2.2.1 Preliminary Study

Crocker *et al.*¹⁵⁴ reported issues with Nile blue not dissolving completely, during the development of acid Nile blue. The excess Nile blue reacted with sulfuric acid, leading to yellow staining that could impact fluorescence intensity. However, this issue appears less problematic with 0.3 M sulfuric acid. To investigate the issue of yellow staining, 50 mL of acid Nile blue solution was filtered through filter paper to remove any sediments and then compared with unfiltered acid Nile blue solution.

Additionally, the quartered fingermark methodology was used to determine the optimal processing time for acid Nile blue treatment. Sample treatment was performed as described by Crocker *et al.*¹⁵⁴ with each quarter of a fingermark sample immersed for different time intervals of 5 mins, 10 mins, 20 mins or 30 mins.

All preliminary studies were conducted using Winc copy paper. While natural fingermarks were initially used, the experiments were repeated with charged fingermarks due to poor results. Consequently, charged fingermarks were also used for comparative assessments with ORO.

6.2.2.2 Comparison to Oil red O

Acid Nile blue was compared to ORO under dry and wet conditions as well as three aging periods. Two sets of split fingermarks from each donor on recycled copy paper were collected, with one set submerged in tap water and the other stored under dry conditions in an enclosed box under office environment. Both dry and wet samples were aged for 1 day before each half of the split fingermarks was processed with either acid Nile blue or ORO. To minimise damage while handling paper substrates under wet conditions, all split fingermarks were cut into halves before storage and wet samples were allowed to dry before treatment. The experiment was repeated for aging periods of 10 days and 30 days.

For substrate investigation, donors deposited split fingermarks on four paper types listed in Table 6.1. The fingermark samples were stored for 7 days under wet conditions following a similar procedure to the wet samples as above.

6.2.3 As a post-cyanoacrylate stain

6.2.3.1 Comparison to Basic Yellow 40

To compare acid Nile blue and BY40, natural fingermarks were collected from five donors on three substrates listed in Table 6.2. To investigate donor effects, charged fingermarks were collected under the same conditions.

6.2.3.2 Confocal Raman microscopy investigation into post-cyanoacrylate staining mechanism.

As an initial study using confocal Raman microscopy, three sets of charged single split fingermarks were collected from a good donor on microscope slides. To ensure the integrity of the fingermark ridges during development, the side-by-side microscope slides for split fingermark collection were secured with cellulose tape at each end. After cyanoacrylate fuming and curing time, one set was stained with acid Nile blue, one with BY40, and one left unstained. The microscope slides were carefully separated allowing lateral and transverse examination of the stained and unstained fingermarks. The samples were examined with confocal Raman microscopy as outlined in section 2.6.5.

6.3 Results and discussion

6.3.1 As a lipid-sensitive reagent on wet porous substrates

6.3.1.1 Preliminary investigations

The preliminary investigation revealed that whilst Nile blue residue was observed on the filter paper there was no major difference between filtered and unfiltered solution. More importantly, no yellow staining was observed (Figure 6.3). This indicates that using 0.3 M sulphuric acid solution is sufficient to dissolve Nile blue and not cause yellow staining. For all donors, the 20 minutes immersion time developed the best quality fingermarks. For the processing time, the 5 minutes immersion developed the least number of fingermarks whilst the 20 minutes immersion time consistently developed the best quality fingermarks. This suggested that the optimal processing time for the Nile blue to partition into the fingermark residue is 20 minutes. Therefore, the processing time was kept at 20 minutes for all other experiments.



Figure 6.3: Comparison of fingermarks developed by filtered acid Nile blue solution (left half) and unfiltered acid Nile blue solution (right half): (a) under white light with no observable yellow stains and (b) at 450 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter.

6.3.1.2 Comparison to Oil red O

Frick *et al.*¹³⁷ modified Beaudoin's ORO formulation, resulting in a simple, non-toxic, low-cost lipid-sensitive fingermark detection method that aligns with the frugal forensics concept similar to acid Nile blue. The comparative assessment of the two methods was investigated using split fingermarks and three aging periods (1, 10 and 30 days) under dry and wet conditions. The results shown in Figure 6.4 and Figure 6.5, indicate that ORO outperforms acid Nile blue in developing fingermarks subjected to wet or dry conditions.



Figure 6.4: Percentage of fingermarks graded 3&4 out of 72 fingermark halves developed with ORO or acid Nile blue under wet and dry conditions.

It was observed that with increasing aging period both reagents were less effective, with no fingermarks detected after 30 days. The result is consistent with other studies that have shown ORO and Nile blue target the 'fragile fraction' of the lipid component and hence are only effective on fingermarks that are less than a few weeks old.^{107, 139, 154, 320} As expected, the two reagents appeared to be more effective on wet fingermarks, likely due to reduced exposure to air for the lipid component

when submerged, slowing degradation of the fragile fraction.^{15, 35} This result suggests that like ORO, acid Nile blue should be applied to fingermarks less than 4 weeks old.



Figure 6.5: Comparison of fingermarks aged 1 day, 10 days and 30 days developed by ORO (left half) under white light and acid Nile blue (right half) at 505 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter.

An investigation was performed on four substrates using split fingermarks submerged in tap water for 1 week to simulate casework conditions where lipidsensitive reagents are likely to be used. The results are shown in Figure 6.6, and reinforce the slight superiority of ORO except on recycled paper, which showed a reversed trend compared to the previous result. This could originate from intradonor variation in sample composition, potentially compounded by the limited number of donors.

Both reagents were most effective in developing fingermarks on A4 premium copy paper and recycled paper with the least effective development seen on brown paper and newspaper (Figure 6.6 and Figure 6.7). The low porosity of the brown paper and newspaper is less likely to absorb and retain the lipid fragile fraction, leaving it more susceptible to being washed off over 1 week. Previous work by Fritz *et al.*³²¹ reported a decrease in fingermark quality with increasing water submersion periods.

Although ORO seems more sensitive as a lipid reagent, the added value of the photoluminescent properties of acid Nile blue prompted further investigation as a post-cyanoacrylate stain on non-porous substrates.



Figure 6.6: Number and grades of fingermarks developed by ORO and acid Nile blue on A4 copy paper, recycled A4 copy paper, kraft brown paper and newspaper after 1 week of submersion in tap water.



Figure 6.7: Comparison of fingermarks on different substrates submerged in tap water for 1 week and developed by Oil red O (left half) under white light and acid Nile blue (right half) at 505 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter: (a) white copy paper (b) 50 % recycled copy paper (c) brown kraft paper and (d) newspaper.

6.3.2 Suitability as a post-cyanoacrylate stain and exploring the staining mechanism

6.3.2.1 Preliminary investigation

Due to the COVID-19 pandemic, an improvised fuming chamber was used instead of the intended commercial fuming chamber at the Western Australia Police Force facility, which was inaccessible due to access restrictions. Previous work by McGann *et al.*¹⁸⁷ and Boseley *et al.*¹⁸⁸ at Curtin University, used an Ikea Sokker Greenhouse (35cm x 45 cm x 22 cm) as a sealed terrarium container (Figure 6.8a) with a cupshaped aluminium foil to hold the superglue (~1.5 g) in the centre as a fuming chamber. The quality of cyanoacrylate marks developed after 2 hours of fuming was on par with fingermarks developed at the Western Australia Police Force facility.¹⁸⁷ Preliminary investigation using this method was conducted and reproducible results were obtained (Figure 6.8b)



Figure 6.8: (a) Improvised cyanoacrylate fuming chamber (b) quality of cyanoacrylate fingermarks developed after 2 hours of fuming on borosilicate microscope slides.

The processing condition for staining with acid Nile blue follows the procedure of simple immersion in solution as when used as a lipid-sensitive reagent on porous substrate. As part of the method development, the staining processing time was reduced to ten minutes since it gave comparable results to the suggested twenty minutes processing time in Frick *et al.*¹⁴⁶ (Figure 6.9). The reduced processing time is likely due to fingermark deposits on non-porous surfaces being more accessible for Nile blue to partition into, compared to fingermark residues in a cellulose matrix for porous surfaces.



Figure 6.9: Comparison of cyanoacrylate fumed fingermarks stained by acid Nile blue for ten minutes (left half) and twenty minutes (right half), at 505 nm excitation wavelength of Polilight and viewed with 550 nm long-pass barrier filter.

6.3.2.2 Comparison to Basic Yellow 40

To directly compare acid Nile blue and BY40, natural split fingermarks from five donors on three substrates were treated with both methods. The results for all substrates and donors are summarised in Figure 6.10. These results show that acid Nile blue has potential as post-cyanoacrylate stains with over 50 % of fingermarks graded as identifiable, although BY40 is a more effective option.



Figure 6.10: Percentage of fingermarks grades developed out of 30 fingermark halves with acid Nile blue and Basic Yellow 40 on borosilicate glass, aluminium sheet, and polypropylene plastic lid.

It was observed that BY40 consistently performed well across all donors and substrates, while acid Nile blue varied across donors, exhibiting weak development with some donors. This is illustrated in Figure 6.11 which shows a comparison of donor 1 and donor 3 across all three substrates. BY40 developed all fingermarks, however with acid Nile blue, no useful fingermarks were developed for donor 3. Interestingly on plastic, donor 1 inadvertently overlaid fingermarks on a sample deposited by donor 3. The development of this sample is shown in Figure 6.11e, demonstrating the donor effect, as only donor 1's fingermarks were developed using acid Nile blue. Further investigation into natural and charged fingermarks for all donors showed that fingermarks treated with acid Nile blue indicated preferential staining of lipid-rich fingermarks, whereas no donor effect was observed with BY40.



Figure 6.11: Comparison of fingermarks developed with acid Nile blue (left half) and Basic Yellow 40 (right half) illustrating donor effect: (a) on borosilicate glass for donor 1 (b) on aluminium sheet for donor 1 (c) borosilicate glass for donor 3 (d) aluminium sheet for donor 3 (e) on polypropylene superimposed of donor 1 (top) and donor 3 (bottom).

These results suggest that staining with acid Nile blue involves the stain partitioning into the sebaceous component of the fingermark deposit. A possible avenue to increase the technique's sensitivity is to explore the more hydrophobic derivates of Nile red and Nile blue, such as dialkylated derivatives. These have been demonstrated through initial studies³²² to produce more luminescent and visible fingermarks. It must be noted that specific photographic conditions were set for the experiment and that the weak development with acid Nile blue can be further enhanced by increasing exposure time or using enhancement software.

Another inference is that two different stain mechanisms are at play. In contrast to the general idea that the dyes stain the polymer structure, the results seem to suggest that in the case of acid Nile blue, the dye is likely slip through the polymer structure to interact with the fingermark deposit. This prompted further investigation into the staining mechanism and the benefit of sequencing acid Nile blue with BY40 to improve detection of overlapping fingermarks, in addition to the initial results (Figure 6.11e).

6.3.2.3 Sequencing to enhance detection of overlapping fingermarks

Overlapping fingermarks is known to reduce their recoverability. This portion of the study explored whether using two different staining mechanisms could enhance detection through sequencing. Two types of overlapping fingermarks were collected on microscope slides (Figure 6.12); one set having natural fingermarks (right thumb) overlapping a charged one (left thumb), and the other in the reverse position, with each set collected twice. Fingermarks were collected from three donors and developed using the methodology outlined in section 6.2.3. Each type of overlapping cyanoacrylate fumed fingermarks was stained using the sequence acid Nile blue \rightarrow BY40 and the reverse sequence.



Figure 6.12: Overlapping fingermark collection (a) natural over charged fingermarks (b) charged over natural fingermarks.

Given that fingermark quality would depend on the sequence of the two techniques, the University of Canberra (UC) scale was adapted to provide a more relevant evaluation. Under treatment 1, UC scores of (+2) were given if both charged and natural fingermarks were developed or (+1) if only one fingermark was developed, which would be classified as a useful fingermark under the CAST scale. If both fingermarks are only partially developed, a score of (0) is given, while a score of (-1) is assigned if only one fingermark is partially developed and (-2) if no fingermarks are developed. In treatment 2, scores are adjusted based on whether the quality of fingermarks improved or deteriorated. The results of this study are summarised in Figure 6.13 and Figure 6.14.

As can be seen for both types of overlapping fingermarks, using the sequence BY40 \rightarrow acid Nile blue resulted in the first treatment (BY40) being effective in staining natural and charged fingermarks, but the second (acid Nile blue) not being as successful. This may be due to BY40 staining the polycyanoacrylate polymer structure, as it has been suggested that the cyanide (CN⁻) anions of polycyanoacrylate form weak van der Waals bonds with basic dye cations.⁶⁸ Consequently, this could prevent the acid Nile blue, as the second treatment from partitioning into the underlying fingermark deposit.



Figure 6.13: Overlapped fingermarks developed with cyanoacrylate and stained using the sequence $BY40 \rightarrow acid$ Nile blue, (top) image of results at different development stages and (bottom) UC scale percentage scores assessment result.



Figure 6.14: Overlapped fingermarks developed with cyanoacrylate and stained using the sequence acid Nile blue \rightarrow BY40, (top) image of results at different development stages and (bottom) UC Scale percentage scores assessment result.

When using sequence acid Nile blue \rightarrow BY40, as expected acid Nile blue exhibited preferential staining of the charged fingermarks, providing better resolution with the overlapped natural fingermarks (Figure 6.14). The use of BY40 as a second treatment, in most instances, led to the development of both natural and charged fingermarks. This reinforces the suggestion that two different mechanisms are at play, with acid Nile blue staining the underlying fingermark deposit and not preventing the staining of the polymer structure by BY40. Based on this result, acid Nile blue would be recommended in the sequence acid Nile blue \rightarrow BY40.

6.3.2.4 Confocal Raman microscopy investigation into post-cyanoacrylate staining mechanism.

The staining mechanism was investigated using confocal Raman microscopy to explore where the different staining dye is sitting in post-cyanoacrylate stained fingermark. The fundamental understanding has the benefit of directing the development of post-cyanoacrylate staining and its position in the sequencing process.

To begin with, an assessment was conducted to ensure that both dyes gave a response with the 532 nm laser of the confocal Raman microscope. This was accomplished by scanning a drop of each dye on a microscope slide. The results depicted in Figure 6.15 showed that both dyes produced a fluorescence response, which was characterised by a broad peak compared to sharper peaks observed during Raman scattering.³²³ More importantly, the maximum intensity of each peak occurred at a distinct wavenumber, making it possible to differentiate between the two dyes.

Subsequently, a fumed cyanoacrylate fingermark was analysed to investigate possible interference of the cyanoacrylate polymer by comparing unstained cyanoacrylate fingermarks with stained ones. All three types of polycyanoacrylate marks were distinguishable as shown in Figure 6.16. The unstained fingermark shows a maxima at 2300 cm⁻¹, while the BY40 stain mark has a maxima at 1400 cm⁻¹. On the other hand, acid Nile blue has a maxima at 3500 cm⁻¹, which corresponds to a Raman

shift wavelength of 650 nm, the expected maxima for Nile red. This outcome was to be expected since Nile red is responsible for the photoluminescence properties exhibited by Nile blue, as discussed previously.



Figure 6.15: Raman spectra of acid Nile blue and Basic Yellow 40 liquid smear on glass slide, acquired using WITec alpha 300SAR confocal Raman microscope equipped with a 532 nm NdYAG laser.



Figure 6.16: Raman spectra of unstained cyanoacrylate fingermark (red), cyanoacrylate fingermark stained with acid Nile blue (blue) and cyanoacrylate fingermark stained with Basic Yellow 40 (yellow)

on a glass slide, acquired using WITec alpha 300SAR confocal Raman microscope equipped with a 532 nm NdYAG laser.

However, examination of post-cyanoacrylate-stained fingermarks did not provide conclusive information about the staining mechanism. The cross-section of the fingermarks did not reveal the structure of the polymer (Figure 6.17), possibly due to reasons such as the fingermark not being in focus, the instrument's spatial resolution, or the polymer's transparent nature. However, overlay Raman images show fluorescent and non-fluorescent regions, indicating the fingermark ridges and furrows (Figure 6.18).



Figure 6.17: Visual image of the cross-section of post-cyanoacrylate stain fingermark on glass slide. The fingermark and polymer structure not apparent. The glass side was mounted edge-up.

Spectra taken across different positions of the fingermarks' cross-sections gave similar profiles, and the cyanoacrylate polymer structure could not be distinguished from the fluorescent stain of acid Nile blue (Figure 6.19). The fluorescence distribution of the fumed fingermark suggested a flat morphology, likely due to the non-humidity-controlled chamber used for the cyanoacrylate process in this study. Previous studies have shown that the microstructures formed during the development of the marks depend on the relative humidity, with 70-90 % producing noodle-like structures while 60 % produced flat, film-like structures. The film-like structure is expected to be shorter in height than the noodle-like structure, possibly too small and beyond the instrument's spatial resolution. Nonetheless, the results in section 6.3.2.2 suggest that acid Nile blue and BY40 stained cyanoacrylate fingermarks via two different mechanisms.



Figure 6.18: Visual and Raman image overlay of the cross-section of post-cyanoacrylate stain fingermark on glass slide showing fluorescent and non-fluorescent regions.



Figure 6.19: Overlay of Raman spectra at two opposite ends of a post-cyanoacrylate fingermark stained with acid Nile blue on a glass slide, acquired using a WITec alpha 300SAR confocal Raman microscope equipped with a 532 nm NdYAG laser.
6.4 Conclusion

This chapter explored acid Nile blue as an emerging fingermark detection technique that is low-cost, non-toxic, easily accessible, and easy to use, making it ideal as a frugal forensics method. It was compared to ORO as a lipid-sensitive reagent and investigated as a potential post-cyanoacrylate fingermark stain.

Like ORO, acid Nile blue was found to successfully develop fingermarks on wet porous substrates, with its effectiveness decreasing as the fingermark ages, resulting in no fingermarks detected after 30 days. The result reinforces the suggestion that Nile blue targets the 'fragile fraction' of the sebaceous component and should be applied to fingermarks less than 4 weeks old.

As a post-cyanoacrylate stain, acid Nile blue was effective in developing lipid-rich fingermarks whereas no preferential development was observed with BY40. These findings indicate that acid Nile blue interacts with the sebaceous component of the fingermark deposit. It appears that acid Nile blue penetrates through the polymer structure to interact with the fingermark residue, contrary to the belief that dyes stain the polymer structure.

Due to the limited sensitivity of acid Nile blue, it is unlikely to replace the Seychelles Police's current operational techniques. However, its frugal forensics attributes can be applied as a training tool and in outreach programs given that it is low-cost, waterbased, and low toxicity, particularly when using the non-acidified version. Moreover, the insight gained from understanding the post-cyanoacrylate staining mechanism can aid in developing improved post-cyanoacrylate stains. It is important to note that the aforementioned conclusions must be qualified with the recognition that the studies were conducted on a limited number of fingermark samples and substrates. Therefore, further analysis is required to establish generalisation. Portions of this chapter have been published in the following articles:

Jemmy T. Bouzin, Thais Lópes, Anna L. Heavey, Jessie. Parrish, Georgina Sauzier, and Simon W. Lewis. Mind the gap: The challenges of sustainable forensic science service provision. *Forensic Science International Synergy* **2023** *6* 100318. DOI: <u>https://doi.org/10.1016/j.fsisyn.2023.100318</u>

7.1 Introduction

In Chapters 3, 4, 5 and 6, various methods were identified and modified to achieve sustainability in line with the frugal forensic approach. Sustainability was evaluated based on a particular attribute of frugal forensics. For example, the SPF 1,2-indanedione zinc (IND/Zn) was found to be more sustainable than the UK CAST formulation due to the cost of the working solution and the difference in availability and accessibility of the carrier solvents.^{233, 324} However, the counterargument is that in terms of safety, the CAST formulation is more sustainable as it uses a non-flammable solvent HFE-7100, compared to SPF IND/Zn which uses the flammable solvent petroleum ether 40-60 °C.^{201, 241} This raises the question of how to assess the sustainability of a method comprehensively and objectively within the frugal forensics framework.

As discussed in Chapter 1, frugal forensics focuses on three core principles (Resilient, Economical and Quality) and six attributes (Performance, Accessibility, Availability, Cost, Simplicity and Safety) which take into consideration jurisdictional local challenges to develop sustainable forensic science provisions. Regardless of the local challenges, the development and assessment of forensic science methods must be carried out in a manner reflecting international best practices and minimum requirements.^{2, 325} For latent fingermark detection, the IFRG guidelines²⁴ provide such an evaluation and reporting framework for the performance assessment of new or modified methods. The guidelines specifically recommend testing methods in their local operational environment. The frugal forensics approach can be viewed as an extension of this recommendation by assessing performance and sustainability in a local context.

To address the aforementioned question, an assessment tool was developed that draws upon the IFRG fingermark performance framework through the use of a similar scoring system as the fingermark grading scale.^{19, 224} This tool consists of two components (depicted in Table 7.1 and Table 7.2) to assess the sustainability between two methods. Table 7.1 provides a comparative scoring system to assess the sustainability of a new or modified method against an existing method referred

to as "alternative" and "control" methods, respectively. Table 7.2 provides a guide on how likely the alternative method would be recommended for implementation.

PAACSS comparative score						
Score	Description					
2	Major benefit of alternative method					
1	Significant benefit of alternative method					
0.5	Minor benefit of alternative method					
0	No difference between methods					
-0.5	Minor benefit of control method					
-1	Significant benefit of control method					
-2	Major benefit of control method					

 Table 7.1: Frugal forensics sustainability comparative score assessment tool

 Table 7.2: Frugal forensics method recommendation score assessment tool

Recommendation score						
Total Score Description						
≥3	Strongly recommended					
2	Recommended					
0.5	Maybe recommended					
≤ 0	Not Recommended					

The assessment works by evaluating each of the six frugal forensics attributes between the alternative and control method using Table 7.1. The sum of scores (total score) are calculated and based on this score, Table 7.2 provides a guide of how likely the alternative method would be recommended for implementation with respect to the control method. The attributes are classified into minor, significant, and major benefits, with their relevance being evaluated according to the local circumstances of each jurisdiction. For instance, a jurisdiction with a reliable supply chain may assign less weight to the availability factor. In other words, the weights are not uniform but rather contingent upon a jurisdiction's unique challenges. This approach enables a comprehensive assessment of the six attributes of frugal forensics and, consequently, the sustainability of the method for a specific jurisdiction.

This chapter evaluates the sustainability of the methods developed in previous chapters using the developed assessment tool. The application of the tool, results, and its benefits are discussed.

7.2 Experimental

The assessment using the evaluative tool was conducted as follows:

The SPF IND/Zn formulation, investigated in Chapters 3 and 4, was evaluated using the CAST formulation as a control method. CAST IND/Zn was chosen as the control since it is the method expected to be used by default in Seychelles.

The Wet SPF method, developed in Chapter 5, was evaluated using Wetwop[™] as a control method. Acid Nile blue as a post-cyanoacrylate stain, investigated in Chapter 6, was evaluated using Basic Yellow 40 (BY40) as a control method. These two control methods were chosen because they are the current operational techniques used in Seychelles.

7.3 Results and Discussion

Table 7.3 provides a summary of the assessment tool results for the sustainability evaluation of the various methods explored in this study under Seychelles conditions and limitations. Based on the results, SPF IND/Zn and Wet SPF powder suspension are recommended as replacements for CAST IND/Zn and Wetwop[™], respectively. However, acid Nile blue is not recommended as a substitute for BY40. The reasoning for each detection technique's score is explained below.

In Chapters 3 and 4, it was shown that the SPF IND/ZN and CAST formulations were equally effective compared to the Australian NCFS formulation. Therefore, a score of 0 was attributed to the performance of SPF against the CAST formulation. However, the main difference between these two formulations is the carrier solvent. CAST formulation uses HFE-7100, which poses a significant supply chain risk, while the SPF formulation uses petroleum ether 40-60 °C, which is more easily accessible, thus reducing this risk. The latter solvent hence provides a major advantage in managing the supply chain risk, resulting in a score of +2 attributed to both Availability and Accessibility. The difference in carrier solvents also results in a variation in cost, with one litre of working solution costing \$267 USD for CAST and \$122 USD for the SPF formulation. Inexpensive formulation is a major economic advantage for laboratories with limited budgets and gives the SPF formulation a score of +2 for cost. Petroleum ether is a highly flammable solvent and is a significant safety risk. However, the risk is reduced when used in a properly fitted laboratory with safety protocols in place, hence a score of -1 was attributed to safety. Both methods are quite similar in terms of equipment and processing, except that the CAST formulation requires oven processing while the SPF formulation uses a heat press. The latter is a less expensive and simpler process, so it was given a simplicity score of +0.5. Based on the total score of +5.5, SPF IND/Zn is the strongly recommended option over the CAST formulation under Seychelles conditions and limitations.

Table 7.3: Comparative score and recommendation results for SPF IND/Zn with CAST formulation, Wet SPF powder suspension with Wetwop[™] formulation and post-cyanoacrylate acid Nile blue method with BY40.

Fingermark detection method			Frugal forensics attributes assessed using Table 6.1							Recommendation assessed using Table 6.2	
Туре	Alternative method	Control method	Performance	Availability	Accessibility	Cost	Safety	Simplicity	Total score	Recommendation	
1,2- indanedione zinc	SPF	CAST	0	+2	+2	+2	-1	+0.5	+5.5	Strongly recommended	
Powder suspension	Wet SPF	Wetwop™	0	0	+1	+2	0	-1	+2	Recommended	
Post- cyanoacrylate staining	Acid Nile blue	BY40	-2	0	0	+1	0	0	-1	Not recommended	

In Chapter 5, the comparative effectiveness of Wet SPF to the commercial Wetwop[™] powder suspension on both acrylic and rubber-based pressure-sensitive adhesive tapes was clearly shown. Hence, a score of 0 was attributed to the performance of Wet SPF against Wetwop[™] formulation. The Wet SPF formulation was a modification of the WET UCIO powder formulation¹⁷⁵ with the inaccessible Gran Velada surfactant being replaced by an in-house surfactant solution prepared from a readily available SDS salt, accessible from multiple suppliers. In remote locations such as Seychelles, using in-house powder suspension (Wet SPF) compared to commercial formulation (Wetwop[™]) allows for easier transportation, storage, and better control of quality and shelf-life. This means that although both formulations are readily available, Wet SPF is more accessible. Therefore, a score of 0 and +1 was attributed to Availability and Accessibility, respectively. In terms of cost, the Wet SPF formulation presents a major advantage, being four times cheaper than Wetwop[™], hence a score of +2 was attributed to Cost. Both formulations are low-risk procedures but Wetwop[™] is a simpler method as it does not involve preparation. Therefore, Safety and Simplicity were rated 0 and -1, respectively. Based on the total score of +2, the Wet SPF powder suspension formulation is the recommended option over the commercial Wetwop[™] formulation for operational use in Seychelles.

In Chapter 6, although acid Nile blue was shown to be effective as a postcyanoacrylate stain – particularly for lipid-rich fingermark deposits - BY40 was found to be superior in developing all types of fingermarks. Therefore, a score of -2 was attributed to performance. The approximate cost of BY40 dye in one litre of working solution is \$6.32 USD, while Nile blue costs only \$0.21 USD. This represents a significant economic saving, hence a score of +1 was given for Cost. Both dyes are readily available and accessible, safe and simple, resulting in a score of 0 for all of these attributes. Based on the total score of -1, acid Nile blue is not recommended as a replacement for BY40 under Seychelles conditions.

In essence, the assessment tool provides a more objective and holistic evaluation for the sustainability of latent fingermark methods in consideration of the jurisdictionspecific challenges. It provides a transparent process to evaluate and select methods for implementation. This is crucial to demonstrate the validity and weighting of results provided to end-users of forensic information and forms a critical component of quality assurance in forensic science service provision.

Furthermore, the assessment tool has benefits beyond decision-making, as it can also identify areas for sustainability improvement. For example, a graphical representation of the scores for the IND/Zn and acid Nile blue methods, with the control method given a base score of zero (Figure 7.1), reveals safety as an area for improvement for SPF IND/Zn and performance for acid Nile blue.



Figure 7.1: Comparison of frugal forensics attributes scores for (a) the SPF IND/Zn formulation with CAST as control method and (b) acid Nile blue as a post-cyanoacrylate stain with BY40 as control. All control methods were given a base score of zero.

Similarly, to the hierarchy of controls in managing health and safety risks,^{326, 327} a framework could be put in place to address the factors that contribute to the identified areas or negative scores. Subsequently, the modified method could be reevaluated using the sustainability tool, providing a methodical process for developing sustainable methods. Hence the assessment tool can be used both as decision-making and a planning tool. It is important to acknowledge that while the current tool is tailored towards latent fingermark methods in micro and remote jurisdictions, it can also be adjusted to suit other jurisdictions and other forensic science service provisions.

7.4 Conclusion

This chapter addresses the question of how to comprehensively and objectively evaluate the sustainability of a method using the frugal forensics approach. Frugal forensics focuses on three principles and six attributes to develop sustainable forensic science provisions. International best practices and minimum requirements must be followed for developing and accessing forensic science methods. For latent fingermark detection, the IFRG guidelines provide an evaluation and reporting structure. In consideration of these frameworks, a grading tool was developed to assess the sustainability of new or modified forensic science methods.

This approach uses a scoring system to evaluate the sustainability of a new or modified method based on the six attributes of frugal forensics and provides a recommendation for implementation. Based on the evaluative tool results in the context of Seychelles, the SPF IND/Zn was strongly recommended over the CAST formulation. Similarly, the Wet SPF powder suspension was recommended over the commercial Wetwop[™] formulation, while acid Nile blue was not recommended as a replacement for BY40.

Overall, the tool offers a more objective and comprehensive assessment that considers the unique challenges of each jurisdiction. This approach is both rigorous and thorough, ensuring a holistic evaluation of the frugal forensics sustainability attributes. Moreover, the objective approach promotes transparency in method selection and adheres to the quality assurance requirements.

Portions of this chapter have been published in the following articles:

Jemmy T. Bouzin, Aaron J. Horrocks, Georgina Sauzier, Stephen M. Bleay, and Simon W. Lewis. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry* **2022** *30* 100439.

DOI: https://doi.org/10.1016/j.forc.2022.100439

Jemmy T. Bouzin, Thais Lópes, Anna L. Heavey, Jessie. Parrish, Georgina Sauzier, and Simon W. Lewis. Mind the gap: The challenges of sustainable forensic science service provision. *Forensic Science International Synergy* **2023** 6 100318. DOI: <u>https://doi.org/10.1016/j.fsisyn.2023.100318</u>

8.1 Conclusion

This thesis aimed to improve latent fingermark detection protocols in remote and resource-limited jurisdictions, focusing on Seychelles as a small island state. In this project, the frugal forensics framework has been proposed as a means of enhancing the accessibility of latent fingermark detection techniques and enabling their use in the Global South. This approach emphasises the development of sustainable methods that are adapted to local challenges for a resilient and cost-effective forensic science service. Through the successful application of this concept to three different latent fingermark methods, two sustainable Seychelles Police Force (SPF) formulations (SPF 1,2-indanedione/zinc and Wet SPF powder suspension) that are well-suited for use in Seychelles were developed. Additionally, an evaluation tool was developed to facilitate a holistic assessment of the sustainability of methods to be implemented while satisfying the quality assurance requirements. Its utility as a strategic planning instrument for the systematic enhancement of the sustainability of a particular methodology is an added advantage.

In essence, this thesis presents the first systematic framework for sustainable forensic science service provision. In the context of latent fingermark provision, this project will improve the detection capabilities of latent fingermarks and facilitate their use in operational laboratories in the Global South, particularly jurisdictions with limited resources such as the Seychelles. The development of sustainable methods ensures service continuity where otherwise it would have been limited or absent. Implementation of effective latent fingermark detection techniques has the potential to increase the recovery of fingermark evidence and improve their effectiveness in criminal investigations. Altogether, this research will have a direct impact on the forensic reform in Seychelles, improving the forensic capacity and fostering research and development. The outcomes may also be used as a benchmark by jurisdictions facing similar circumstances, such as Mauritius, Madagascar, Philippines, and Timor-Leste. Hence, creating more consistency in the quality of forensic service provision across different jurisdictions. The generated knowledge will likely have a global impact, particularly as other regions search for alternative carrier solvents for amino acid sensitive reagents.³²⁴ Future work recommendations are presented below, in addition to the chapter-specific conclusions.

8.2 Alternative carrier solvent for amino acid sensitive reagent

In Chapter 3, the frugal forensics approach was used to improve an existing method for fingermark enhancement on porous surfaces. This was achieved by modifying the co-solvents, active chemicals, and carrier solvents of three active 1,2indanedione/zinc (IND/Zn) formulations to develop a sustainable formulation (SPF IND/Zn). The main challenge was to replace the expensive and supply chaindisruptive solvent, HFE-7100, in the IND/Zn formulation.²³³ Through the frugal forensics approach, petroleum ether 40-60 °C in the BKA IND/Zn formulation was found to be a better alternative for managing costs and reducing supply chain risks. Solstice® PF was also considered as an alternative carrier solvent, but it presented similar supply chain challenges to HFE-7100 whilst causing higher ink diffusion and possible chemical interactions with other reagent components that could affect stability. The benefit of frugal forensics was further demonstrated by reducing excess 1,2-indanedione in the BKA formulation, improving reagent effectiveness and cost. The importance of achieving the right balance of the polar solvent to ensure optimal luminescence while minimising background development and ridge diffusion was highlighted.

Future work should explore other alternative carrier solvents, especially with the recent development that the production of the standard carrier solvent for amino acid sensitive reagents, HFE-7100, will be discontinued by 3M in 2025.³²⁴ From a frugal forensics perspective, it would be beneficial to investigate other readily available and affordable solvents that have similar properties to a suitable carrier solvent. Additionally, it is important to gain a better understanding of how the carrier solvent Solstice[®] PF interacts with reagent components and how it affects reagent performance. This information will be valuable in the search for alternative carrier solvents.

8.3 Powder suspension and acid Nile blue improvements

The versatility of the frugal forensics approach was demonstrated on non-porous surfaces through its application to a recently published method. Chapter 5 focused on the development of an effective and accessible method for detecting latent fingermarks on pressure-sensitive adhesive (PSA) tape. The method involves modifying a recently published¹⁷⁵ powder suspension formulation by substituting the commercial surfactant solution with an in-house SDS solution, resulting in a more affordable and easily accessible reagent. The effectiveness of the modified formulation was shown to be comparable to that of the commercial Wetwop[™] powder suspension.

Future work should focus on re-assessing the effective ratio of powder to the modified surfactant solution. The in-house SDS surfactant solution was produced using a 5 % ethanol/water mixture as an additive. Variation in the concentration and type of additives was found to play a minimal role in the powder suspension performance. However, it is important to investigate if this holds true for other additives. Moreover, the modified formulation demonstrated efficacy in developing latent fingermarks on both light and dark-coloured tapes when used with a forensic light source. This suggests the possibility of using a single formulation for different coloured tapes. As this study was confined to carbon-based powder, it is suggested that further research be conducted to compare results with white powder suspension. Currently, there is ongoing research at Curtin University on improvised powders under the banner of frugal forensics. Combining the Wet SPF surfactant with viable improvised powders presents another avenue for future research.

In Chapter 6, acid Nile blue as an emerging lipid-sensitive fingermark detection technique with frugal forensics attributes was investigated. Through comparative analysis with Oil red O, the study revealed that acid Nile blue is most effective in detecting charged fingermarks that are less than 4 weeks old. Additionally, acid Nile blue is selective in its sensitivity towards sebaceous material when used as a post-cyanoacrylate stain, suggesting a different staining mechanism compared to Basic

Yellow 40. However, further research is needed to enhance acid Nile blue's sensitivity, which may involve exploring other benzophenoxazine dyes.

8.4 IFRG guidelines requirements

The IND/Zn formulation and powder suspension method developed in Chapters 3-5, were investigated in line with phases 2/3 of the IFRG guidelines.²⁴ Comparative assessment of each method against an operational technique was carried out through laboratory trials using split fingermarks on a range of ten donors and multiple substrates. The evaluation included incidental fingermarks and simulated casework substrates for meaningful results using more realistic samples. The sensitivity of the methods was tested using split depletion fingermarks.

Overall, the effectiveness of the methods was demonstrated using 1200 fingermark halves on five different porous substrates for SPF IND/Zn and 320 fingermarks on eight different PSA tapes for Wet SPF. Additionally, both formulations were also shown to be effective on 1 week and 4 weeks old samples, which is the general timeframe for processing casework samples in Seychelles. However, additional IFRG requirements must be satisfied before either method's application in casework. Further operational validation is needed on casework samples to test their operational suitability and this phase should be carried out at the operational laboratory.

8.5 Sustainable forensic science service provision

Chapter 7 addressed the question of how to systematically assess the sustainability of a method. An assessment tool was developed drawing on the same concept utilised in the fingermark grading scale. By objectively assessing all six attributes of frugal forensics, this tool promotes transparency in method selection and adheres to the quality assurance requirements. Not only does it aid in decision-making, but it also acts as a valuable planning tool by pinpointing areas in need of improvement. The application of the tool was demonstrated through the successful assessment of the three methods investigated in Chapters 3-6. Furthermore, the frugal forensics concept provides an opportunity to revisit current methods from a distinct perspective.

In summary, this thesis has demonstrated how frugal forensics can be applied and evaluated to develop sustainable latent fingermark detection techniques. While the focus is on fingermark detection, the economic, resilient, and quality-driven principles of frugal forensics can be extended to the broader forensic science service in the Global South. Through awareness of frugal forensics, investigation into and application of attainable, fit-for-purpose methodologies may bridge the gap in service provision in the Global South, promoting justice and aiding the public good, for present and future generations. The role of forensic science in achieving the United Nations Sustainable Development Goals is significant and calls for future research to promote an inclusive and outward-looking approach to forensic science, especially as the forensic community³²⁸ re-evaluates the current state and definition of this field.

1. Ubelaker DH. *The global practice of forensic science*. John Wiley & Sons; 2015.

Wilson-Wilde L. The international development of forensic science standards
 A review. *Forensic Science International*. 2018;288:1-9. doi:<u>http://doi.org/10.1016/j.forsciint.2018.04.009</u>

3. Cordner S, Bouwer H, Tidball-Binz M. The Ebola epidemic in Liberia and managing the dead-A future role for Humanitarian Forensic Action? *Forensic Sci Int*. 2017;279:302-309. doi:<u>http://doi.org/10.1016/j.forsciint.2017.04.010</u>

4. Jaymelee JK. An Alternative Approach to Forensic Anthropology: Findings from Northern Uganda. *African Conflict and Peacebuilding Review*. 2018;8(1)doi:<u>http://doi.org/10.2979/africonfpeacrevi.8.1.02</u>

5. Ahmad Syukran B, Muhammad Hazim A, Wan Abdul Fattah Wan I, Lukman Abdul M, Mohammad Amir Wan H. Catalysing global peace through the strengthening of forensic science application in Sharī'ah law. *Al-Shajarah: Journal of the International Institute of Islamic Thought and Civilization (ISTAC)*. 2019;0(0):77-103. <u>https://journals.iium.edu.my/shajarah/index.php/shaj/article/view/924</u>

6. Tidball-Binz MV, Cordner S. Humanitarian forensic action: A new forensic discipline helping to implement international law and construct peace. *WIREs Forensic Science*. 2021;4(1)doi:<u>https://doi.org/10.1002/wfs2.1438</u>

7. Shepitko VY, Shepitko MV. The role of forensic science and forensic examination in international cooperation in the investigation of crimes. *Journal of the National Academy of Legal Sciences of Ukraine*. 2021;28(1):179-186.

8. United Nations. *The Sustainable Development Goals Report 2021*. United Nations; 2021. <u>https://unstats.un.org/sdgs/report/2022/The-Sustainable-Development-Goals-Report-2022.pdf</u>

9. Lee BX, Kjaerulf F, Turner S, et al. Transforming Our World: Implementing the 2030 Agenda Through Sustainable Development Goal Indicators. *J Public Health Policy*. 2016;37 Suppl 1:13-31. doi:<u>http://doi.org/10.1057/s41271-016-0002-7</u>

10. Mitlin D, Satterthwaite D. *Urban Poverty in the Global South*. Routledge; 2012. doi:<u>https://doi.org/10.4324/9780203104316</u>

11. Smith M, Miller S. The Rise of Biometric Identification: Fingerprints and Applied Ethics. In: Smith M, Miller S, eds. *Biometric Identification, Law and Ethics*. Springer International Publishing; 2021:1-19. doi:10.1007/978-3-030-90256-8_1

12.Hazarika P, Russell DA. Advances in fingerprint analysis. Angewandte ChemieInternationalEdition.2012;51(15):3524-3531.doi:https://doi.org/10.1002/anie.201104313

13.Interpol.Fingerprintsfactsheet.2020.https://www.interpol.int/content/download/622/file/FS-
03%20FingerPrint%20FactsheetsEN2020-03.pdf

14. Champod C, Lennard CJ, Margot P, Stoilovic M. *Fingerprints and Other Ridge Skin Impressions, Second Edition*. 2 ed. 2016. doi:<u>https://doi.org/10.1201/b20423</u>

15. Ramotowski R. *Lee and Gaensslen's Advances in Fingerprint Technology, Third Edition*. 3 ed. 2012. doi:<u>https://doi.org/10.1201/b12882</u>

16. Bandey H, Bleay SM, Bowman VJ, Downham R, Sears V. *Fingermark Visualisation Manual*. Home Office CAST; 2014.

17. Stoilovic M, Lennard C. *Workshop Manual: Fingermark Detection & Enhancement*. 6th ed. Incorporating Light Theory and General Forensic Applications of Optical Enhancement Techniques. National Centre for Forensic Studies; 2012.

18. Bleay SM, Bailey MJ, Croxton RS, Francese S. The forensic exploitation of fingermark chemistry: A review. *Wiley Interdisciplinary Reviews: Forensic Science*. 2021;3(4):e1403. doi: <u>https://doi.org/10.1002/wfs2.1403</u>

19.Sears VG, Bleay SM, Bandey HL, Bowman VJ. A methodology for fingermarkresearch.Science& Justice.2012;52(3):145-160.doi:https://doi.org/10.1016/j.scijus.2011.10.006

20. Steiner R, Roux C, Moret S. Controlling fingermark variability for research purposes: A review. *WIREs Forensic science*. 2019;1(4):e1338-n/a. doi:<u>https://doi.org/10.1002/wfs2.1338</u>

21.Boseley RE, Vongsvivut J, Appadoo D, Hackett MJ, Lewis SW. Monitoring the
chemical changes in fingermark residue over time using synchrotron infrared
spectroscopy.2022;147(5):799-810.doi:https://doi.org/10.1039/D1AN02293H2022;147(5):799-810.

22. Gorka M, Thomas A, Bécue A. Differentiating individuals through the chemical composition of their fingermarks. *Forensic Science International*. 2023;346:111645. doi:<u>https://doi.org/10.1016/j.forsciint.2023.111645</u>

23. van Helmond W, van Herwijnen AW, van Riemsdijk JJH, van Bochove MA, de Poot CJ, de Puit M. Chemical profiling of fingerprints using mass spectrometry. *Forensic Chemistry*. 2019;16:100183. doi:<u>https://doi.org/10.1016/j.forc.2019.100183</u>

24. International Fingerprint Research Group. Guidelines for the Assessment of Fingermark Detection Techniques. *Journal of forensic identification*. 2014;64(2):174. <u>https://www.theiai.org/docs/JFI-2014-02-174.pdf</u>

25. Jelly R, Patton EL, Lennard C, Lewis SW, Lim KF. The detection of latent fingermarks on porous surfaces using amino acid sensitive reagents: a review. *Anal Chim Acta*. 2009;652(1-2):128-142. doi:<u>http://doi.org/10.1016/j.aca.2009.06.023</u>

26. Bécue A, Champod C. Interpol review of fingermarks and other body impressions (2019 – 2022). *Forensic Science International: Synergy*. 2023;6:100304. doi:<u>https://doi.org/10.1016/j.fsisyn.2022.100304</u>

27. Roberts MA, Tom KR, Knorr KB. Publication trends in forensic science research: Friction ridge discipline. *Forensic Science International: Synergy*. 2020;2:76-81. doi:<u>https://doi.org/10.1016/j.fsisyn.2020.01.001</u>

28. Ramotowski RS. *Composition of latent print residue*. 2001:63-104. doi:<u>https://doi.org/10.1201/9781420041347</u>

29. Bramble SK, Brennan JS. FINGERPRINTS (DACTYLOSCOPY) | Chemistry of Print Residue. In: Siegel JA, ed. *Encyclopedia of Forensic Sciences*. Elsevier; 2000:862-869. doi:<u>https://doi.org/10.1006/rwfs.2000.0737</u>

30. Baker LB. Physiology of sweat gland function: The roles of sweating and sweat composition in human health. *Temperature*. 2019;6(3):211-259. doi:<u>https://doi.org/10.1080/23328940.2019.1632145</u>

31. Wilke K, Martin A, Terstegen L, Biel SS. A short history of sweat gland biology. International Journal of Cosmetic Science. 2007;29(3):169-179. doi:<u>https://doi.org/10.1111/j.1467-2494.2007.00387.x</u>

32. Chen YL, Kuan WH, Liu CL. Comparative Study of the Composition of Sweat from Eccrine and Apocrine Sweat Glands during Exercise and in Heat. *Int J Environ Res Public Health*. 2020;17(10)doi:<u>https://doi.org/10.3390/ijerph17103377</u>

33. Justice USDo. McRoberts A, ed. *The Fingerprint Sourcebook*. U.S. Dept. of Justice, Office of Justice Programs, National Institute of Justice; 2014. <u>www.nij.gov</u>

34. Robinson S, Robinson AH. Chemical composition of sweat. *Physiological reviews*. 1954;34(2):202-220. doi:<u>https://doi.org/10.1152/physrev.1954.34.2.202</u>

35. Cadd S, Islam M, Manson P, Bleay S. Fingerprint composition and aging: A literature review. *Science & Justice*. 2015;55(4):219-238. doi:<u>https://doi.org/10.1016/j.scijus.2015.02.004</u>

36. Girod A, Ramotowski R, Weyermann C. Composition of fingermark residue: a qualitative and quantitative review. *Forensic Science International*. 2012;223(1-3):10-24. doi:<u>https://doi.org/10.1016/j.forsciint.2012.05.018</u>

37. O'Neill KC, Hinners P, Lee YJ. Potential of triacylglycerol profiles in latent fingerprints to reveal individual diet, exercise, or health information for forensic evidence. *Analytical methods*. 2020;12(6):792-798. doi:https://doi.org/10.1039/c9ay02652e

38. Ferguson LS, Wulfert F, Wolstenholme R, et al. Direct detection of peptides and small proteins in fingermarks and determination of sex by MALDI mass spectrometry profiling. *Analyst*. 2012;137(20):4686-4692. doi:<u>https://doi.org/10.1039/c2an36074h</u>

39. Croxton RS, Baron MG, Butler D, Kent T, Sears VG. Variation in amino acid and lipid composition of latent fingerprints. *Forensic Sci Int*. 2010;199(1):93-102. doi:<u>https://doi.org/10.1016/j.forsciint.2010.03.019</u>

40. Moraleda Merlo AB, Roux C, Bécue A, Weyermann C. A comparison of the natural and groomed fingermark lipid composition of different donors using GC/MS. *Forensic Science International*. 2023;348:111709. doi:<u>https://doi.org/10.1016/j.forsciint.2023.111709</u>

41. Keisar O, Cohen Y, Finkelstein Y, et al. Measuring the water content in freshlydeposited fingermarks. *Forensic Science International*. 2019;294:204-210. doi:<u>https://doi.org/10.1016/j.forsciint.2018.11.017</u>

42. Harker M, Coulson H, Fairweather I, Taylor D, Daykin CA. Study of metabolite composition of eccrine sweat from healthy male and female human subjects by 1H NMR spectroscopy. *Metabolomics*. 2006;2(3):105-112. doi:<u>https://doi.org/10.1007/s11306-006-0024-4</u>

43. Atherton T, Croxton R, Baron M, Gonzalez-Rodriguez J, Gámiz-Gracia L, García-Campaña AM. Analysis of amino acids in latent fingerprint residue by capillary electrophoresis-mass spectrometry. *Journal of Separation Science*. 2012;35(21):2994-2999. doi:<u>https://doi.org/10.1002/jssc.201200398</u>

44. Hadorn B, Hanimann F, Anders P, Curtius H-C, Halverson R. Free amino-acids in human sweat from different parts of the body. *Nature*. 1967;215(5099):416-417. doi:<u>https://doi.org/10.1038/215416a0</u>

45. de Puit M, Ismail M, Xu X. LCMS Analysis of Fingerprints, the Amino Acid Profile of 20 Donors. Journal of Forensic Sciences. 2014;59(2):364-370. doi:https://doi.org/10.1111/1556-4029.12327

46. Frick A, Fritz P, Lewis S. Siegel JA, Saukko PJ, eds. Chemistry of Print Residue. 2 ed. Academic Press; 2013:92-97. doi:https://doi.org/10.1016/B978-0-12-382165-2.00262-2

47. Cuthbertson F. The chemistry of fingerprints, United Kingdom Atomic Energy Authority, Atomic Weapons Research Establishment (AWRE) Report no. 013/69. HM Stationery Office; 1969.

Berglund L, Gallagher R, McNall P. Simulation of the thermal effects of 48. dissolved materials in human sweat. Computers and Biomedical Research. 1973;6(2):127-138. doi:https://doi.org/10.1016/0010-4809(73)90052-9

49. Baker C, Silverman R. Misperceptions about white gloves. International preservation news. 2005;37:4-9. https://core.ac.uk/download/pdf/276287599.pdf

50. Kent T. Water content of latent fingerprints – Dispelling the myth. Forensic 2016;266:134-138. Science International. doi:<u>https://doi.org/10.1016/j.forsciint.2016.05.016</u>

51. Yamashita B, French M. Barnes J, G, ed. Fingerprint Sourcebook, NCJ 225320. US Department of Justice; 2011.

52. Archer NE, Charles Y, Elliott JA, Jickells S. Changes in the lipid composition of latent fingerprint residue with time after deposition on a surface. Forensic Science International. 2005;154(2-3):224-239.

doi:https://doi.org/10.1016/j.forsciint.2004.09.120

Croxton RS, Baron MG, Butler D, Kent T, Sears VG. Development of a GC-MS 53. Method for the Simultaneous Analysis of Latent Fingerprint Components*. Journal of 2006;51(6):1329-1333. doi:https://doi.org/10.1111/j.1556-Forensic Sciences. 4029.2006.00203.x

54. Downing DT, Strauss JS. Synthesis and Composition of Surface Lipids of Skin. Journal of Investigative Dermatology. 1974;62(3):228-244. Human doi:https://doi.org/10.1111/1523-1747.ep12676793

55. Nicolaides N. Skin Lipids: Their Biochemical Uniqueness. Science (American Association for the Advancement of Science). 1974;186(4158):19-26. doi:https://doi.org/10.1126/science.186.4158.19

56. Mountfort KA, Bronstein H, Archer N, Jickells SM. Identification of Oxidation Products of Squalene in Solution and in Latent Fingerprints by ESI-MS and LC/APCI-MS. *Analytical chemistry (Washington)*. 2007;79(7):2650-2657. doi:<u>https://doi.org/10.1021/ac0623944</u>

57.Frick A. Chemical Investigations Into the Lipid Fraction of Latent Fingermark
Residue.Residue.CurtinUniversity;2015.https://espace.curtin.edu.au/handle/20.500.11937/758

58. Koenig A, Girod A, Weyermann C. Identification of wax esters in fingermark residues by GC/MS and their potential use as aging parameters. *Journal of Forensic Identification*. 2011;61(6):652-676. https://serval.unil.ch/resource/serval:BIB_DBAA2F688396.P001/REF.pdf

59.West MJ, Went MJ. The spectroscopic detection of exogenous material in
fingerprints after development with powders and recovery with adhesive lifters.
Forensic Science International. 2008;174(1):1-5.
doi:https://doi.org/10.1016/j.forsciint.2007.02.026

60. Szynkowska M, Czerski K, Rogowski J, Paryjczak T, Parczewski A. Detection of exogenous contaminants of fingerprints using ToF-SIMS. *Surface and Interface Analysis: An International Journal devoted to the development and application of techniques for the analysis of surfaces, interfaces and thin films*. 2010;42(5):393-397. doi: <u>https://doi.org/10.1002/sia.3193</u>

61. Boseley RE, Howard DL, Hackett MJ, Lewis SW. The transfer and persistence of metals in latent fingermarks. *Analyst (London)*. 2022;147(3):387-397. doi:<u>https://doi.org/10.1039/d1an01951a</u>

62. Boseley RE, Dorakumbura BN, Howard DL, et al. Revealing the Elemental Distribution within Latent Fingermarks Using Synchrotron Sourced X-ray Fluorescence Microscopy. *Analytical chemistry (Washington)*. 2019;91(16):10622-10630. doi:<u>https://doi.org/10.1021/acs.analchem.9b01843</u>

63. Ricci C, Kazarian SG. Collection and detection of latent fingermarks contaminated with cosmetics on nonporous and porous surfaces. *Surface and interface analysis*. 2010;42(5):386-392. doi:<u>https://doi.org/10.1002/sia.3098</u>

64. Bradshaw R, Bleay S, Wolstenholme R, Clench MR, Francese S. Towards the integration of matrix assisted laser desorption ionisation mass spectrometry imaging into the current fingermark examination workflow. *Forensic science international*. 2013;232(1):111-124. doi:<u>https://doi.org/10.1016/j.forsciint.2013.07.013</u>

65. Ng PHR, Walker S, Tahtouh M, Reedy B. Detection of illicit substances in fingerprints by infrared spectral imaging. *Analytical and bioanalytical chemistry*. 2009;394(8):2039-2048. doi:<u>https://doi.org/10.1007/s00216-009-2806-9</u>

66. Groeneveld G, De Puit M, Bleay S, Bradshaw R, Francese S. Detection and mapping of illicit drugs and their metabolites in fingermarks by MALDI MS and compatibility with forensic techniques. *Scientific reports*. 2015;5(1):11716-11716. doi:<u>https://doi.org/10.1038/srep11716</u>

67. Zhang T, Chen X, Yang R, Xu Y. Detection of methamphetamine and its main metabolite in fingermarks by liquid chromatography–mass spectrometry. *Forensic science international*. 2015;248:10-14. doi:<u>https://doi.org/10.1016/j.forsciint.2014.12.013</u>

68. Bleay S, Sears V, Downham R, et al. *Fingerprint Source Book v2.0 (second edition)*. 2nd ed. Home Office CAST; 2017.

69.Boudreault A, Beaudoin A. Pseudo-Operational Study on the Efficiency of
Various Fingermark Development Techniques During the Aging Process. Journal of
Forensic Identification.2017;67(1):85-117.https://www.proquest.com/docview/1871397360

70. De Alcaraz-Fossoul J, Mestres Patris C, Balaciart Muntaner A, Barrot Feixat C, Gené Badia M. Determination of latent fingerprint degradation patterns--a real fieldwork study. *International Journal of Legal Medicine*. 127(4):857-870. doi:<u>https://doi.org/10.1007/s00414-012-0797-0</u>

71. De Alcaraz-Fossoul J, Roberts KA, Johnson CA, Barrot Feixat C, Tully-Doyle R, Kammrath BW. Fingermark ridge drift: influencing factors of a not-so-rare aging phenomenon. *Journal of Forensic Sciences*. 2021;66(4):1472-1481. doi:<u>https://doi.org/10.1111/1556-4029.14710</u>

72.Williams DK, Brown CJ, Bruker J. Characterization of children's latentfingerprint residues by infrared microspectroscopy: Forensic implications. ForensicScienceInternational.2011;206(1):161-165.doi:<</td>https://doi.org/10.1016/j.forsciint.2010.07.033

73. Weyermann C, Ribaux O. Situating forensic traces in time. *Science & Justice*. 2012;52(2):68-75. doi:<u>https://doi.org/10.1016/j.scijus.2011.09.003</u>

74.Wertheim K. Fingerprint age determination: Is there any hope? Journal of
ForensicForensicIdentification.Jan/Feb2003;53(1):42-49.https://www.proquest.com/docview/194800313

75. Popa G, Potorac R, Preda N. Method for fingerprints age determination. *Romanian Journal of Legal Medicine*. 2010;18(2):149-154. <u>http://www.rjlm.ro/system/revista/14/149-154.pdf</u>

76. Weyermann C, Roux C, Champod C. Initial results on the composition of fingerprints and its evolution as a function of time by GC/MS analysis. *Journal of forensic sciences*. 2011;56(1):102-108. doi:<u>https://doi.org/10.1111/j.1556-4029.2010.01523.x</u>

77. van Helmond W, Weening M, Vleer V, de Puit M. Analysis of amino acid enantiomers from aged fingerprints. *Analytical Methods*. 2020;12(15):2052-2057. doi:<u>https://doi.org/10.1039/D0AY00096E</u>

78. Pleik S, Spengler B, Ram Bhandari D, et al. Ambient-air ozonolysis of triglycerides in aged fingerprint residues. *Analyst*. 2018;143(5):1197-1209. doi:<u>https://doi.org/10.1039/C7AN01506B</u>

79. Dorakumbura BN. Fundamental Studies into the Chemical and Physical Properties of Latent Fingermarks. Curtin University; 2017. https://espace.curtin.edu.au/handle/20.500.11937/69365

80. Mong G, Petersen C, Clauss T. *Advanced fingerprint analysis project fingerprint constituents*. Pacific Northwest National Lab.(PNNL), Richland, WA (United States); 1999. doi:<u>https://doi.org/10.2172/14172</u>

81. Spindler X, Shimmon R, Roux C, Lennard C. Visualising substrate–fingermark interactions: Solid-state NMR spectroscopy of amino acid reagent development on cellulose substrates. *Forensic Science International*. 2015;250:8-16. doi:<u>https://doi.org/10.1016/j.forsciint.2015.02.019</u>

82. Bleay S, Fitzgerald L, Sears V, Kent T. Visualising the past - An evaluation of processes and sequences for fingermark recovery from old documents. *Science & Justice*. 2019;59(2):125-137. doi:<u>https://doi.org/10.1016/j.scijus.2018.10.005</u>

83. Bouzin JT, Merendino J, Bleay SM, Sauzier G, Lewis SW. New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc. *Forensic Science International: Reports*. 2020;2doi:<u>https://doi.org/10.1016/j.fsir.2020.100145</u>

84. Frick A, Chidlow G, Goodpaster JV, Lewis S, Van Bronswijk W. Monitoring compositional changes of the lipid fraction of fingermark residues deposited on paper during storage. *Forensic Chemistry*. 2016;2:29-36. doi:<u>https://doi.org/10.1016/j.forc.2016.09.001</u>

85. Pleik S, Spengler B, Schäfer T, Urbach D, Luhn S, Kirsch D. Fatty Acid Structure and Degradation Analysis in Fingerprint Residues. *Journal of the American Society for Mass Spectrometry*. 2016;27(9):1565-1574. doi:<u>https://doi.org/10.1007/s13361-016-1429-6</u>

86. Frick AA, Kummer N, Moraleda A, Weyermann C. Changes in latent fingermark glyceride composition as a function of sample age using UPLC-IMS-QTOF-MS. *Analyst (London)*. 2020;145(12):4212-4223. doi:<u>https://doi.org/10.1039/d0an00379d</u>

87. De Alcaraz-Fossoul J, Einfalt MR, Dean ER, Vynalek RW, Chavez A, Roberts KA. Repeatability and reproducibility of the color contrast technique for fingermark age estimation applications. *Journal of Forensic Sciences*. 2021;66(3):1075-1084. doi:<u>https://doi.org/10.1111/1556-4029.14654</u>

88. Szabóová Ž, Galbavá P, Szabó AH, et al. GC–MS/MS method for age determination of fingerprints. *Monatshefte für Chemie - Chemical Monthly*. 2017;148(9):1673-1678. doi:<u>https://doi.org/10.1007/s00706-017-1984-y</u>

89. Brunelle E, Eldridge M, Halámek J. Determination of Time since Deposition of Fingerprints via Colorimetric Assays. *ACS Omega*. 2021;6(19):12898-12903. doi:<u>https://doi.org/10.1021/acsomega.1c01344</u>

90. Bobev K. Fingerprints And Factors Affecting Their Condition. *Journal of forensic identification : the official publication of the International Association for Identification*. 1995;45(2):176.

91. Almog J, Azoury M, Elmaliah Y, Berenstein L, Zaban A. Fingerprints' third dimension the depth and shape of fingerprints penetration into paper : cross section examination by fluorescence microscopy. *Journal of Forensic Sciences*. 2004;49(5):981-985. doi:<u>https://doi.org/10.1520/JFS2004009</u>

92. Bond JW. Visualization of Latent Fingerprint Corrosion of Metallic Surfaces. *Journal of forensic sciences*. 2008;53(4):812-822. doi:<u>https://doi.org/10.1111/j.1556-4029.2008.00738.x</u>

93. Wightman G, Emery F, Austin C, et al. The interaction of fingermark deposits on metal surfaces and potential ways for visualisation. *Forensic science international*. 2015;249:241-254. doi:<u>https://doi.org/10.1016/j.forsciint.2015.01.035</u>

94. Woodward H, Moret S, Chadwick S. Biodegradable plastics and their impact on fingermark detection methods. *Forensic science international*. 2023;344:111571-111571. doi:<u>https://doi.org/10.1016/j.forsciint.2023.111571</u>

95. Illston-Baggs G, Deacon P, Ivanova J, Nichols-Drew L, Armitage R, Farrugia KJ. An investigation into the detection of latent fingermarks on eco-friendly soft plastics packaging. *Forensic chemistry*. 2022;29:100427. doi:<u>https://doi.org/10.1016/j.forc.2022.100427</u>

96. Richmond-Aylor A, Bell S, Callery P, Morris K. Thermal Degradation Analysis of Amino Acids in Fingerprint Residue by Pyrolysis GC-MS to Develop New Latent Fingerprint Developing Reagents. *Journal of forensic sciences*. 2007;52(2):380-382. doi:<u>https://doi.org/10.1111/j.1556-4029.2007.00384.x</u>

97. Brown AG, Sommerville D, Reedy BJ, Shimmon RG, Tahtouh M. Revisiting the Thermal Development of Latent Fingerprints on Porous Surfaces: New Aspects and Refinements. *Journal of forensic sciences*. 2009;54(1):114-121. doi:<u>https://doi.org/10.1111/j.1556-4029.2008.00902.x</u>

98. Dominick AJ, Daeid NN, Bleay SM, Sears VG. The recoverability of fingerprints on paper exposed to elevated temperatures-Part 2: Natural fluorescence. *Journal of forensic identification*. 2009;59(3):340-355. <u>https://strathprints.strath.ac.uk/28320/</u>

99. Ricci C, Phiriyavityopas P, Curum N, Chan KLA, Jickells S, Kazarian SG. Chemical Imaging of Latent Fingerprint Residues. *Applied spectroscopy*. 2007;61(5):514-522. doi:<u>https://doi.org/10.1366/000370207780807849</u>

100. Wolstenholme R, Bradshaw R, Clench MR, Francese S. Study of latent fingermarks by matrix-assisted laser desorption/ionisation mass spectrometry imaging of endogenous lipids. *Rapid communications in mass spectrometry*. 2009;23(19):3031-3039. doi:https://doi.org/10.1002/rcm.4218

101. Kim Y, Choi Ws, Jeon B, Choi TH. The Effect of Temperature and Exposure Time on Stability of Cholesterol and Squalene in Latent Fingermarks Deposited on PVDF Membrane. *Journal of forensic sciences*. 2020;65(2):458-464. doi:<u>https://doi.org/10.1002/rcm.4218</u>

102. Kostyuk V, Potapovich A, Stancato A, et al. Photo-oxidation products of skin surface squalene mediate metabolic and inflammatory responses to solar UV in human keratinocytes. *PloS one*. 2012;7(8):e44472-e44472. doi:<u>https://doi.org/10.1371/journal.pone.0044472</u>

103. Paine M, Bandey HL, Bleay SM, Willson H. The effect of relative humidity on the effectiveness of the cyanoacrylate fuming process for fingermark development and on the microstructure of the developed marks. *Forensic Sci Int.* 2011;212(1-3):130-142. doi:<u>https://doi.org/10.1016/j.forsciint.2011.06.003</u>

104. Spindler X, Shimmon R, Roux C, Lennard C. The effect of zinc chloride, humidity and the substrate on the reaction of 1,2-indanedione-zinc with amino acids

in latent fingermark secretions. *Forensic Sci Int*. 2011;212(1-3):150-157. doi:<u>https://doi.org/10.1016/j.forsciint.2011.06.005</u>

105. Rawji A, Beaudoin A. Oil Red O Versus Physical Developer on Wet Papers: A Comparative Study. *Journal of Forensic Identification*. 2006;56(1):33-54. https://www.proquest.com/docview/194801377

106.Sauzier G, Frick AA, Lewis SW. Investigation into the Performance of Physical
Developer Formulations for Visualizing Latent Fingerprints on Paper. Journal of
ForensicForensicIdentification.https://espace.curtin.edu.au/handle/20.500.11937/21342

107.Salama J, Aumeer-Donovan S, Lennard C, Roux C. Evaluation of theFingermark Reagent Oil Red O as a Possible Replacement for Physical Developer.JournalofForensicIdentification.2008;58(2):203-237.https://www.proquest.com/docview/194806809

108. Almog J, Glasner H. Ninhydrin Thiohemiketals: Basic Research Towards Improved Fingermark Detection Techniques Employing Nano-Technology*. *Journal of Forensic Sciences*. 2010;55(1):215-220. doi:<u>https://doi.org/10.1111/j.1556-4029.2009.01231.x</u>

109. Almog J, Klein A, Davidi I, Cohen Y, Azoury M, Levin-Elad M. Dual Fingerprint Reagents with Enhanced Sensitivity: 5-Methoxy- and 5-Methylthioninhydrin. *Journal of Forensic Sciences*. 2008;53(2):364-368. doi:<u>https://doi/org/10.1111/j.1556-4029.2008.00671.x</u>

110. Almog J, Springer E, Wiesner S, Frank A. Latent Fingerprint Visualization by 1,2-Indanedione and Related Compounds: Preliminary results. *Journal of Forensic Sciences*. 1999;44(1):114-118. doi:<u>https://doi.org/10.1520/JFS14421J</u>

111. Ruhemann S. CXXXII.—Cyclic di- and tri-ketones. Journal of the Chemical
Society,Transactions.1910;97(0):1438-1449.doi:https://doi.org/10.1039/CT9109701438

112. Pounds C, Grigg R. The Use Of 1,8-Diazafluoren-9-One (DFO) For TheFluorescent Detection Of Latent Fingerprints On Paper. A Preliminary Evaluation.JournalofForensicSciences.1990;35(1):169-175.doi:https://doi.org/10.1520/JFS12813J

113. Ramotowski R, Cantu AA, Joullié MM, Petrovskaia O. 1,2-Indanediones: A Preliminary Evaluation of a New Class of Amino Acid Visualizing Compounds. *Fingerprint Whorld*. 1997;23(90):131-140.

114. Hauze DB, Petrovskaia O, Taylor B, Joullié MM, Ramotowski R, Cantu AA. 1,2-Indanediones: New reagents for visualizing the amino acid components of latent prints. *Journal of Forensic Sciences*. 1998;43(4):744-747. doi:<u>https://doi.org/10.1520/JFS14300J</u>

115. Stoilovic M, Lennard C, Wallace-Kunkel C, Roux C. Evaluation of a 1,2-Indanedione Formulation Containing Zinc Chloride for Improved Fingermark Detection on Paper. *Journal of Forensic Identification*. 2007;57(1):4-18. <u>https://www.proquest.com/docview/194805152</u>

116. Roux C, Jones N, Lennard C, Stoilovic M. Evaluation of 1,2-indanedione and 5,6-dimethoxy-1,2-indanedione for the detection of latent fingerprints on porous surfaces. *Journal of forensic sciences*. 2000;45(4):761. doi:<u>https://doi.org/10.1520/JFS14768J</u>

117. Wiesner S, Springer E, Sasson Y, Almog J. Chemical development of latent fingerprints: 1,2-indanedione has come of age. *Journal of Forensic Sciences*. 2001;46(5):1082-1084. doi:<u>https://doi.org/10.1520/JFS15102J</u>

118. Wilkinson D, MacKenzie E, Leech C, Mayowski D, Bertrand S, Walker T. The Results from a Canadian National Field Trial Comparing Two Formulations of 1, 8-Diazafluoren-9-one (DFO) with 1, 2-Indanedione. *Identification Canada*. 2003;26(2):8-18.

119.Merrick S, Gardner S, Sears V, Hewlett D. An operational trial of ozone-
friendly DFO and 1,2-indanedione formulations for latent fingerprint detection.
Journal of Forensic Identification. 2002;52(5):595-605.
https://www.proquest.com/docview/194803644

120. Gardner SJ, Hewlett DF. Optimization and initial evaluation of 1,2-indandione as a reagent for fingerprint detection. *Journal of forensic sciences*. 2003;48(6):1288. doi:<u>https://doi.org/10.1520/JFS2000381</u>

121. Wallace-Kunkel C, Roux C, Lennard C, Stoilovic M. The detection and
enhancement of latent fingermarks on porous surfaces - A survey. Journal of Forensic
Identification.2004;54(6):687-705.bttps://www.presevent.com/dominus/presevent

https://www.proquest.com/docview/194796637

122. Wallace-Kunkel C, Lennard C, Stoilovic M, Roux C. Optimisation and evaluation of 1,2-indanedione for use as a fingermark reagent and its application to real samples. *Forensic Science International*. 2007;168(1):14-26. <Go to ISI>://000246533700002

123. D'Elia V, Materazzi S, Iuliano G, Niola L. Evaluation and comparison of 1,2indanedione and 1,8-diazafluoren-9-one solutions for the enhancement of latent fingerprints on porous surfaces. *Forensic Sci Int*. 2015;254:205-214. doi:<u>https://doi.org/10.1016/j.forsciint.2015.07.036</u>

124. Levin-Elad M, Liptz Y, Bar-Or KL, Almog J. 1,2-Indanedione - A winning ticket for developing fingermarks: A validation study. *Forensic Sci Int*. 2017;271:8-12. doi:<u>https://doi.org/10.1016/j.forsciint.2016.12.007</u>

125. Luscombe A, Sears V. A validation study of the 1,2-indandione reagent for operational use in the UK: Part 3-Laboratory comparison and pseudo-operational trials on porous items. *Forensic Science International*. 2018;292:254-261. doi:<u>https://doi.org/10.1016/j.forsciint.2018.04.042</u>

126. Loh CN, Chia WLK, Foo SCS, Xu X, Tan WLS. Evaluation of the performance of IND/Zn and DFO on various porous substrates in Singapore context. *Forensic Science International*. 2020;315:110440. doi:https://doi.org/10.1016/j.forsciint.2020.110440

127. Nicolasora N, Downham R, Hussey L, Luscombe A, Mayse K, Sears V. A validation study of the 1,2-indandione reagent for operational use in the UK: Part 1 — Formulation optimization. *Forensic Science International*. 2018;292:242-253. doi:http://doi.org/10.1016/j.forsciint.2018.04.046

128. Nicolasora N, Downham R, Dyer RM, Hussey L, Luscombe A, Sears V. A validation study of the 1,2-indandione reagent for operational use in the UK: Part 2 - Optimization of processing conditions. *Forensic Science International*. 2018;288:266-277. doi:<u>https://doi.org/10.1016/j.forsciint.2018.04.044</u>

129. Beaudoin A. New Technique for Revealing Latent Fingerprints on Wet, Porous Surfaces: Oil Red O. *Journal of Forensic Identification*. 2004;54(4):413-421. https://www.proquest.com/docview/194797174

130. Bumbrah GS, Sodhi GS, Kaur J. Oil Red O (ORO) reagent for detection of latent fingermarks: a review. *Egyptian Journal of Forensic Sciences*. 2019;9(1)doi:<u>http://doi.org/10.1186/s41935-018-0107-1</u>

131. Bleay SM, Bradshaw G, Moore JE. *Fingerprint Development and Imaging Newsletter: Special Edition: Arson*. Home Office Scientific Development Branch; 2006. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/atta chment data/file/895952/Fingerprint Update Special Edition - Arson 26-06.pdf

132. Dominick AJ, Daéid NN, Bleay SM, Sears V. The Recoverability of Fingerprints on Paper Exposed to Elevated Temperatures - Part 1: Comparison of Enhancement Techniques. *Journal of Forensic Identification*. 2009;59(3):325-339. <u>https://search.proquest.com/docview/194839756</u> 133. de la Hunty M, Moret S, Chadwick S, Lennard C, Spindler X, Roux C.Understanding Physical Developer (PD): Part II--Is PD targeting eccrine constituents?ForensicScienceInternational.2015;257:488-495.doi:https://doi.org/10.1016/j.forsciint.2015.08.029

134. Defence Science and Technology Laboratory. Fingermark Visualisation Newsletter November 2019, 2019,

135. Proescher F. Oil red O pyridin, a rapid fat stain. *Stain Technology*. 1927;2(2):60-61. doi:<u>https://doi.org/10.3109/10520292709115655</u>

136. Castello A, Alvarez M, Miquel M, Verdu F. Long-lasting lipsticks and latent prints. *Forensic science communications*. 2002;4(2),

137. Frick AA, Fritz P, Lewis SW, van Bronswijk W. A Modified Oil Red OFormulation for the Detection of Latent Fingermarks on Porous Substrates. JournalofForensicIdentification.2012;62(6):623-641.https://www.proquest.com/docview/1241120090

138. Guigui K, Beaudoin A. The Use of Oil Red O in Sequence with Other Methods of Fingerprint Development. *Journal of forensic identification*. 2007;57(4):550. <u>https://www.proquest.com/docview/194805899</u>

139. Wood MA, James T. ORO. The physical developer replacement? *Sci Justice*. 2009;49(4):272-276. doi:<u>https://doi.org/10.1016/j.scijus.2009.02.006</u>

140. Simmons RK, Deacon P, Farrugia KJ. Water-soaked porous evidence: A comparison of processing methods. *Journal of forensic identification*. 2014;64(2):157-173. <u>https://www.proquest.com/docview/1511419457</u>

141.Beaudoin A. Oil Red O: Fingerprint Development on a 21-Year-Old Cold Case.JournalofForensicIdentification.2011;61(1):50-59.https://www.proquest.com/docview/847006272

142. Martinez V, Henary M. Nile Red and Nile Blue: Applications and Syntheses ofStructuralAnalogues.Chemistry.2016;22(39):13764-13782.doi:https://doi.org/10.1002/chem.201601570

143. Greenspan P, Mayer EP, Fowler SD. Nile Red: A Selective Fluorescent Stain for Intracellular Lipid Droplets. *The Journal of cell biology*. 1985;100(3):965-973. doi: <u>https://doi.org/10.1083/jcb.100.3.965</u>

144. Braasch K, de La Hunty M, Deppe J, et al. Nile red: Alternative to physical developer for the detection of latent fingermarks on wet porous surfaces? *Forensic*

Science

145. Sigma-Aldrich. Nile red, SKU 72485-1G. Accessed February, 2023. https://www.sigmaaldrich.com/SC/en/product/sigma/72485

146. Frick AA, Busetti F, Cross A, Lewis SW. Aqueous Nile blue: a simple, versatile and safe reagent for the detection of latent fingermarks. Chem Commun. 2014;50(25):3341-3343. doi:https://doi.org/10.1039/c3cc49577a

147. Smith JL. The staining of fat by Nile-blue sulphate. Journal of pathology and bacteriology. 1911;15(1):53-55. doi:https://doi.org/10.1002/path.1700150107

148. Cain AJ. The Use of Nile Blue in the Examination of Lipoids. Journal of cell science. 1947;s3-88(3):383-392. doi:https://doi.org/10.1242/jcs.s3-88.3.383

149. Dunnigan MG. The Use of Nile Blue Sulphate in the Histochemical Identification Phospholipids. Stain technology. 1968;43(5):249-256. of doi:https://doi.org/10.3109/10520296809115077

150. Menzel E, Fox K. Laser detection of latent fingerprints: preparation of fluorescent dusting powders and the feasibility of a portable system. Journal of Forensic Sciences. 1980;25(1):150-153. doi:https://doi.org/10.1520/JFS10950J

Chesher B, Stone J, Rowe W. Use of the omniprinttm 1000 alternate light-151. source to produce fluorescence in cyanoacrylate-developed latent fingerprints stained with biological stains and commercial fabric dyes. Forensic Science International. 1992;57(2):163-168.

Castello A, Alvarez-Segui M, Verdu F. Use of fluorescent dyes for developing 152. 2004;120(4):184-187. latent lip prints. Coloration technology. doi:<u>https://doi.org/10.1111/j.1478-4408.2004.tb00227.x</u>

Sigma-Aldrich. Nile blue A, SKU N5632-25G. Accessed February, 2023. 153. https://www.sigmaaldrich.com/SC/en/product/sigma/n5632

154. Crocker R. Detection of latent fingermarks on wetted porous surfaces. Curtin University; 2015.

155. AinswortháMitchell C. The detection of finger-prints on documents. Analyst. 1920;45(529):122-129. https://pubs.rsc.org/en/content/articlepdf/1920/an/an9204500122

210

156. Thomas GL. The physics of fingerprints and their detection. Journal of physics E, Scientific instruments. 1978;11(8):722-731. doi:https://doi.org/10.1088/0022-3735/11/8/002

157. Burns DS. Sticky-side powder: The Japanese solution. Journal of Forensic Identification. 1994;44(2):133-138. https://www.ojp.gov/ncjrs/virtuallibrary/abstracts/sticky-side-powder-japanese-solution

Gray ML. Sticky-Side Powder Versus Gentian Violet: The search for the 158. superior method for processing the sticky side of adhesive tape. Journal of Forensic Identification. 1996;46(3):268-272. https://www.ojp.gov/ncjrs/virtuallibrary/abstracts/sticky-side-powder-versus-gentian-violet-search-superior-method

159. Bratton R, Gregus J. A Black Powder Method to Process Adhesive Tapes. Fingerprint Whorld. 1996;22(83):28.

Martin BL. Developing latent prints on the adhesive surface of black electrical 160. Journal forensic identification. tape. of 1999;49(2):127. https://www.proquest.com/docview/194800537

161. Sneddon N. Black Powder Method to Process Duct Tape. Journal of Forensic Identification. 1999;49(347)(4), https://www.proquest.com/docview/194786105

162. Parisi KM. Getting the Most From Fingerprint Powders. Journal of Forensic 1999;49(5):494-498. Identification.

https://www.proquest.com/docview/194801418

163. Wade DC. Development of latent prints with titanium dioxide (TiO2). Journal of forensic identification. 2002;52(5):551-559. https://www.proquest.com/docview/194792061

Downham RP, Sears VG, Hussey L, Chu BS, Jones BJ. Fingermark visualisation 164. with iron oxide powder suspension: The variable effectiveness of iron (II/III) oxide powders, and Tween((R)) 20 as an alternative to Triton X-100. Forensic Science International. 2018;292:190-203.

doi:https://doi.org/10.1016/j.forsciint.2018.09.012

Downham RP, Tomasz MC, Hemant JD, Vaughn GS. Black Iron (II/III) Oxide 165. Powder Suspension (2009 CAST Formulation) for Fingermark Visualization, Part 1: Formulation Component and Shelf-Life Studies. Journal of forensic identification. 2017;67(1):118. https://www.proquest.com/docview/1871396809

Downham R, Ciuksza TM, Desai HJ, Sears VG. Black Iron (II/III) Oxide Powder 166. Suspension (2009 CAST Formulation) for Fingermark Visualization, Part 2: Surfactant Solution Component Investigations. *Journal of forensic identification*. 2017;67(1):145. <u>https://www.proquest.com/docview/1871397275</u>

167. Williams NH, Elliott KT. Development of Latent Prints using Titanium Dioxide (TiO2) in Small Particle Reagent, White (SPR-W) on Adhesives. *Journal of Forensic Identification*. 2005;55(3):292.

168. Defence Science and Technology Laboratory. Fingermark Visualisation Newsletter. 2020;(October 2020)(DSTL/PUB126361), https://assets.publishing.service.gov.uk/government/uploads/system/uploads/atta chment data/file/936196/2020 Oct Dstl Fingermark Visualisation Newseltter 6 v1.0 O.pdf

169.Daéid NN, Carter S, Laing K. Comparison of Three Types of White PowderSuspensions for the Recovery of Fingerprints on Wetted Nonporous Surfaces. JournalofForensicIdentification.2008;58(5):590-599.https://www.proquest.com/scholarly-journals/comparison-three-types-white-powder-suspensions/docview/2805239283/se-2

170. Daéid NN, Carter S, Laing K. Comparison of Vacuum Metal Deposition and Powder Suspension for Recovery of Fingerprints on Wetted Nonporous Surfaces. *Journal of Forensic Identification*. 2008;58(5):600-613. <u>https://www.proquest.com/scholarly-journals/comparison-vacuum-metal-</u> <u>deposition-powder/docview/194838953/se-2</u>

171. Safariland. Wetwop™ Black Lightning Powder Safety Data Sheet,. Revised 15
 February 2016,

https://sds.chemtel.net/webclients/safariland/finished_goods/Lightning%20Powde r%20-%20Wetwop%20Black.pdf

 172.
 KJELL CARLSSON INNOVATION. Wet Powder Black Safety Data Sheet,. Revised

 04
 November
 2011,

 https://sds.chemtel.net/webclients/safariland/raw
 materials/Wet%20Powder%20

 Black%20(Wetwop%20Black).pdf

173. Sirchie. ASP50D Adhesive Side Powder, Dark Safety Data Sheet,. Revised March 2012,

https://www.sirchie.com/media/resourcecenter/item/a/s/asp50d 2.pdf

174. Safariland. Wetwop[™] White Lightning Powder Safety Data Sheet,. Revised 15 February 2016,

https://sds.chemtel.net/webclients/safariland/finished_goods/Lightning%20Powde r%20-%20Wetwop%20White.pdf 175. Claveria S, Clares N, Fernández P, Heredia R, Godall A. WET UCIO - New Powder Suspension Formula for Fingerprint Development on the Adhesive Side of Tape. *Journal of forensic identification*. 2022;72(2):174-184. https://www.proquest.com/docview/2664099624

176. Wood L. The discovery of superglue fuming. *Fingerprint Whorld*. 1991;16(64):117-118.

177. Bumbrah GS. Cyanoacrylate fuming method for detection of latent fingermarks: a review. *Egyptian journal of forensic sciences*. 2017;7(1):4-4. doi:<u>https://doi.org/10.1186/s41935-017-0009-7</u>

178. Algaier D, Baskaran D, Dadmun M. The influence of temperature on the polymerization of ethyl cyanoacrylate from the vapor phase. *Reactive and Functional Polymers*. 2011;71(8):809-819. doi:https://doi.org/10.1016/j.reactfunctpolym.2011.03.015

179. Lewis LA, Smithwick RW, 3rd, Devault GL, Bolinger B, Lewis SA, Sr. Processes involved in the development of latent fingerprints using the cyanoacrylate fuming method. *Journal of Forensic Sciences*. 2001;46(2):241-246. doi:<u>https://doi.org/10.1520/JFS14956J</u>

180. Lide DR. CRC handbook of chemistry and physics. vol 85. CRC press; 2004.

181. Farrugia KJ, Fraser J, Friel L, Adams D, Attard-Montalto N, Deacon P. A comparison between atmospheric/humidity and vacuum cyanoacrylate fuming of latent fingermarks. *Forensic Science International*. 2015;257:54-70. doi:<u>https://doi.org/10.1016/j.forsciint.2015.07.035</u>

182. Mankidy PJ, Rajagopalan R, Foley HC. Influence of initiators on the growth of poly(ethyl 2-cyanoacrylate) nanofibers. *Polymer*. 2008;49(9):2235-2242. doi:<u>https://doi.org/10.1016/j.polymer.2008.03.018</u>

183. Mankidy PJ, Rajagopalan R, Foley HC. Facile catalytic growth of cyanoacrylatenanofibers.Chemicalcommunications.2006;(10):1139-1141.doi:http://doi.org/10.1039/B514600C

184. Kendall FG, Rehn BW. Rapid method of super glue[®] fuming application for the development of latent fingerprints. *Journal of forensic sciences*. 1983;28(3):777-780. doi:<u>http://doi.org/10.1520/JFS11573J</u>

185. Olenik J. Super glue[®], a modified technique for the development of latent fingerprints. *Journal of Forensic Sciences*. 1984;29(3):881-884. doi:<u>http://doi.org/10.1520/JFS11748J</u>

186. Martindale II W. Cyanoacrylate fuming as a method for rapid development of latent fingerprints utilizing anhydrous sodium carbonate as a dry catalyst. *Identif News*. 1983;13,

187. McGann J, Simon WL. *Novel Luminescent Staining for Visualisation of Latent Fingermarks on Metal Substrates*. Curtin University; 2015.

188. Boseley R, Frick A, Simon WL. *Detection of Latent Fingermarks on Non-Porous Substrates using Cyanoacrylate Fuming and Luminescent Stains*. Curtin University; 2016.

189. INTERPOL. Fingerprints 2023. <u>https://www.interpol.int/en/How-we-work/Forensics/Fingerprints</u>

190. Committee on Identifying the Needs of the Forensic Sciences Community, National Research Council. *Strengthening Forensic Science in the United States: A Path Forward*. 2009:348. doi:<u>https://doi.org/10.17226/12589</u>

191. Seychelles Meteorological Authority. Climate data. 2023. <u>www.meteo.gov.sc</u>

192. Seychelles National Bureau of Statistics. Population and vital statistics December 2021. 2021. Accessed Nov 2022, <u>https://www.nbs.gov.sc/</u>

193. Cellmark Forensic Services. *Forensic DNA Laboratory Project: A proposal for the Seychelles Police*. Seychelles Police; 2017.

194. Seychelles Police. *Scientific Support & Crimianl Records Bureau Strategic Plan* 2019-2021. 2019.

195. Seychelles Nation. Fight against crime gets major boost. 2013, <u>http://www.nation.sc/archive/239276/fight-against-crime-gets-major-boost</u>

196. Bouzin JT, Sauzier G, Lewis SW. Forensic science in Seychelles: An example of a micro-jurisdiction forensic delivery system. *Forensic Science International: Synergy*. 2021;3:100139. doi:<u>https://doi.org/10.1016/j.fsisyn.2021.100139</u>

197. Bécue A, Cantú AA. Fingermark detection using nanoparticles. CRC Press LLC; 2012. p. 307-379.

198. Bécue A, Eldridge H, Champod C. Interpol review of fingermarks and other body impressions 2016–2019. *Forensic Science International: Synergy*. 2020;doi:<u>https://doi.org/10.1016/j.fsisyn.2020.01.013</u>

199. Sears V, Bleay S, Batham R. The effectiveness of 1,2-indandione-zinc formulations and comparison with HFE-based 1,8-diazafluoren-9-one for fingerprint development. *Journal of Forensic Identification*. 2009;59(6):654-678.

200. Regulation (EU) No 517/2014 of The European Parliament and of the council of 16 April 2014 on fluorinated greenhouse gases and repealing Regulation (EC) No 842/2006. Off.J.Eur.Union (2014) L 150 195-230.

201. 3M[™] Novec[™] 7100 Engineered Fluid (2009). Electronics Markets Materials Division.; 2009. <u>https://multimedia.3m.com/mws/media/1998180/3m-novec-7100-engineered-fluid.pdf</u>. <u>https://multimedia.3m.com/mws/media/1998180/3m-novec-7100-engineered-fluid.pdf</u>.

202. Harvey-Walker N. The global shortage of HFE-7100, Personal Communications. NSW Police Force, June 2022.

203. Lothridge K. Global shortage of HFE-7100, Personal Communication. March 2022.

204. National Research Council. *National Research Council: Strengthening Forensic Science in the United States : A Path Forward*. National Academies Press, Washington DC; 2009.

http://ebookcentral.proquest.com/lib/curtin/detail.action?docID=3378513

205. President's Council of Advisors on Science and Technology. Forensic Science in Criminal Courts: Ensuring Scientific Validity of Feature-Comparison Methods: Report to the President. President's Council of Advisors on Science and Technology, Washington DC; 2016. <u>https://obamawhitehouse.archives.gov/sites/default/files/microsites/ostp/PCAST/</u> pcast forensic science report final.pdf

206. Edmond G, Towler A, Growns B, et al. Thinking forensics: Cognitive science for forensic practitioners. *Sci Justice*. 2017;57(2):144-154. doi:<u>http://doi.org/10.1016/j.scijus.2016.11.005</u>

207. Ribeiro G, Tangen JM, McKimmie BM. Beliefs about error rates and human judgment in forensic science. *Forensic Sci Int*. 2019;297:138-147. doi:<u>http://doi.org/10.1016/j.forsciint.2019.01.034</u>

208. Dror IE. Cognitive and Human Factors in Expert Decision Making: Six Fallacies and the Eight Sources of Bias. *Anal Chem.* 2020;92(12):7998-8004. doi:<u>http://doi.org/10.1021/acs.analchem.0c00704</u>

209. Catoggio D, Bunford J, Taylor D, Wevers G, Ballantyne K, Morgan R. An introductory guide to evaluative reporting in forensic science. *Australian Journal of*
Forensic Sciences. doi:https://doi.org/10.1080/00450618.2019.1568560

210. Biedermann A, Champod C, Willis S. Development of European standards for evaluative reporting in forensic science. *The International Journal of Evidence & Proof.* 2016;21(1-2):14-29. doi:<u>http://doi.org/10.1177/1365712716674796</u>

211. Etin-Osa D, Ce E-O. Forensic Science and the Nigerian Society. *Journal of Nuclear Sciences*. 2019;6(1):17-21.

212. Amankwaa AO, Nsiah Amoako E, Mensah Bonsu DO, Banyeh M. Forensic science in Ghana: A review. *Forensic Sci Int.* 2019;1:151-160. doi:<u>http://doi.org/10.1016/j.fsisyn.2019.07.008</u>

213. Mateen RM, Tariq A. Crime scene investigation in Pakistan: A perspective.ForensicSciIntSynerg.2019;1:285-287.doi:http://doi.org/10.1016/j.fsisyn.2019.06.046

214. Nderitu JM. Analysing The Efficiency Of Forensic Science Units Within Kenya Police Service In Solving Crimes. University of Nairobi; 2018.

215. Figueiredo ISd, Pareschi ACC. Diagnóstico da perícia criminal no Brasil. 2013,

216. Basu RR, Banerjee PM, Sweeny EG. FRUGAL INNOVATION. *Journal of Management for Global sustainability*. 2013;1(2),

217. Brundtland GH. Brundtland report. Our common future. *Comissão Mundial*. 1987;4(1):17-25.

218. United Nations. Sustainable Developments Goals. Department of Economic and Social Affairs. <u>https://sdgs.un.org/goals</u>

219. Kent T. Standardizing Protocols for Fingerprint Reagent Testing. *Journal of Forensic Identification*. 2010;60(3):371-379.

220. Chadwick S, Moret S, Jayashanka N, Lennard C, Spindler X, Roux C.Investigation of some of the factors influencing fingermark detection. ForensicScienceInternational.doi:https://doi.org/10.1016/j.forsciint.2018.06.014

221. Stoilovic M, Lennard C. *Worshop Manual: Fingermark Detection & Enhancement*. 6th ed. Incorporating Light Theory and General Forensic Applications of Optical Enhancement Techniques. National Centre for Forensic Studies; 2012.

222. Becker I, Heinrich M-L, Schwarz L, Bust M. *Process Instruction Indanedione/Zinc for the visualization of latent fingermarks version 4.1*. Bundeskriminalamt; 2018.

223. Spindler X, Stoilovic M, Lennard C, Lennard A. Spectral variations for reaction products formed between different amino acids and latent fingermark detection reagents on a range of cellulose-based substrates. *Journal of Forensic Identification*. 2009;59(3):308-324.

224. McLaren C, Lennard C, Stoilovic M. Methylamine pretreatment of dry latent fingermarks on polyethylene for enhanced detection by cyanoacrylate fuming. *Journal of forensic identification*. 2010;60(2):199-222.

225. Fritz P, Frick A, Van Bronswijk W, et al. Variability and subjectivity in the grading process for evaluating the performance of latent fingermark detection techniques. 2015, <u>https://espace.curtin.edu.au/handle/20.500.11937/46811</u>

226. Hockey D, Dove A, Kent T. Guidelines for the use and statistical analysis of the Home Office fingermark grading scheme for comparing fingermark development techniques. *Forensic Science International*. 2021;318:110604. doi:<u>https://doi.org/10.1016/j.forsciint.2020.110604</u>

227. Bicknell DE, Ramotowski RS. Use of an Optimized 1,2-Indanedione Process for the Development of Latent Prints. *Journal of Forensic Sciences*. 2008;53(5):1108-1116. doi:<u>https://doi.org/10.1111/j.1556-4029.2008.00826.x</u>

228. WilkinsonD. Spectroscopic study of 1,2-indandione. Forensic ScienceInternational.2000;114(3):123-132.0738(00)00294-2doi:https://doi.org/10.1016/S0379-

229. Chan J, Spindler X, Brust A, et al. Evaluation of DFO and 1,2-indanedione formulations under two different Australian conditions. *Science & Justice*. 2010;50(1):38-38. doi:<u>https://doi.org/10.1016/j.scijus.2009.11.047</u>

230. Fitzi T, Fischer R, Moret S, Bécue A. Fingermark detection on thermal papers: Proposition of an updated processing sequence. *Journal of forensic identification*. 2014;64(4):329-350. <u>https://www.proquest.com/docview/1547329512</u>

231. Stimac JT. The search for safe, non-running solvents: A brief history. Journalofforensicidentification.2000;50(5):455-461.https://www.proquest.com/docview/194828029

232. Petruncio AV. A comparative study for the evaluation of two solvents for use in ninhydrin processing of latent print evidence. *Journal of forensic identification*. 2000;50(5):462-469. <u>https://www.proquest.com/docview/194799534</u>

233. Olszowska I, Deacon P, Lindsay M, Leśniewski A, Niedziółka-Jönsson J, Farrugia K. An alternative carrier solvent for fingermark enhancement reagents. *Forensic Science International*. 2018;284:53-64. doi:<u>https://doi.org/10.1016/j.forsciint.2017.12.012</u>

234. Farrugia K. Alternative solvent to HFE7100 and Triton X100 for fingermark formulations. 2019,

235. Buck RC, Franklin J, Berger U, et al. Perfluoroalkyl and polyfluoroalkyl substances in the environment: Terminology, classification, and origins. *Integrated Environmental Assessment and Management*. 2011;7(4):513-541. doi:<u>https://doi.org/10.1002/ieam.258</u>

236. Dhore R, Murthy GS. Per/polyfluoroalkyl substances production, applications and environmental impacts. *Bioresource Technology*. 2021;341:125808. doi:<u>https://doi.org/10.1016/j.biortech.2021.125808</u>

237. Khuu A, Spindler X, Roux C. Detection of latent fingermarks and cells on paper.Forensicscienceinternational.2020;309:110185-110185.doi:https://doi.org/10.1016/j.forsciint.2020.110185

238. Zhao Y-B, Wang L-X, Li W-J, You W, Farrugia K. Effect of carrier solvent in 1,2indanedione formulation on the development of fingermarks on porous substrates. *Forensic Science International*. 2021;318:110589. doi:https://doi.org/10.1016/j.forsciint.2020.110589

239. Honeywell International Inc. Solstice® Performance Fluid Technical Bulletin.December2017.https://sustainability.honeywell.com/content/dam/sustainability/en/documents/d

ocument-lists/technical/SolsticePf-Specifications-TechBulletin.pdf

240.Serrano E, Sturelle V. Modification and Evaluation of a 1,2-Indanedione-ZincChloride Formula using Petroleum Ether as a Carrier Solvent.Canadian Society ofForensicScienceJournal.2010;43(3):108-116.2010;43(3):108-116.doi:https://doi.org/10.1080/00085030.2010.10757622

241. Sigma-Aldrich. Petroleum ether, bp 40-60 °C. February 2020. https://www.sigmaaldrich.com/SC/en/product/vetec/v800295

242. Hilgert M. Comparison of different Ind/Zn recipes as a pre-stage for the evaluation of this method in Germany. 8th Biennial Meeting of the International Fingerprint Research Group; 2011; Linkoping, Sweden.

243. Honeywell Technical Service. Solstice PF Hansen solubility parameters, Personal Communication. 2022.

244. Hansen Solubility Parameters. 2022, <u>https://www.hansen-solubility.com/</u>

245. Durkee J. *Cleaning with solvents: science and technology*. William Andrew; 2013.

246. Hansen CM. Hansen solubility parameters: a user's handbook. CRC press; 2007.

247. info@reddychemtech.com. Shelf life of 1,2-indanedione - Personal communication. Reddy Chemtech, Inc.; 2022.

248. Nicolasora N, Downham R, Dyer R-M, Hussey L, Luscombe A, Sears V. A validation study of the 1,2-indanedione reagent for operational use in the UK: Part 2 - Optimization of the processing conditions. *Forensic Science International*. 2018;288:266-277.

249. Mayse K. *The development of latent fingerprints on problematic porous surfaces*. Loughborough University; 2011.

250. Sanchez-Salvador JL, Balea A, Monte MC, et al. Comparison Of Mechanical And Chemical Nanocellulose As Additives To Reinforce Recycled Cardboard. *Scientific Reports*. 2020;10(1):3778. doi:10.1038/s41598-020-60507-3

251. Aguilar-Rivera N. Emerging technology for sustainable production of bleached pulp from recovered cardboard. *Clean Technologies and Environmental Policy*. 2021;23(9):2575-2588. doi:10.1007/s10098-021-02171-3

252. Bouzin JT, Horrocks AJ, Sauzier G, Bleay SM, Lewis SW. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry*. 2022;30doi:<u>http://doi.org/10.1016/j.forc.2022.100439</u>

253. Britton C, Matthews N. Investigation into alternatives to HFE7100 for fingerprint development chemicals, Final presentation WSU interns Online 03 Nov. New South Wales Police Force; 2022:

254. Almog J, Azoury M, Elmaliah Y, Berenstein L, Zaban A. Fingerprints' third dimension: The depth and shape of fingerprints penetration into paper—cross section examination by fluorescence microscopy. *Journal of Forensic Sciences*. 2004;49(5):1-5.

255. Spindler X, Shimmon R, Roux C, Lennard C. The effect of zinc chloride, humidity and the substrate on the reaction of 1,2-indanedione–zinc with amino acids in latent fingermark secretions. *Forensic Science International*. 2011;212(1-3):150-157.

256.Zhao YB, Wang LX, Li WJ, You W, Farrugia K. Effect of carrier solvent in 1,2-indanedione formulation on the development of fingermarks on porous substrates.ForensicScienceInternational.2021;318:110589.doi:https://doi.org/10.1016/j.forsciint.2020.110589

257. Zhao Y, Wang L, Li W, You W, Farrugia K. Effect of carrier solvent in 1,2indanedione formulation on the development of fingermarks on porous substrates. *Forensic Science International*. 2021;318:110589.

258. Boudreault A, Beaudoin A. Pseudo-operational study of the efficacy of various fingermark development techniques during the aging process *Journal of Forensic Identification*. 2017;67(1):85-117.

259. Luscombe A, Sears V. A validation study of the 1,2-indanedione reagent for operational use in the UK: Part 3 - Laboratory comparison and pseudo-operational trials on porous items. *Forensic Science International*. 2018;292:254-261.

260. Siegel JA. Ink Analysis. In: Siegel JA, Saukko PJ, Houck MM, eds. *Encyclopedia of Forensic Sciences*. 2nd ed. Academic Press; 2013:375-379.

261. Kunjappu JT. Ink Chemistry. *Chemistry in Britain*. 2003;39(3):22-25.

262. Farrugia K. Personal communication. 2021.

263. Nicolasora N, Downham R, Hussey L, Luscombe A, Mayse K, Sears V. A validation study of the 1,2-indanedione reagent for operational use in the UK: Part 1 - Formulation optimization. *Forensic Science International*. 2018;292:242-253.

264. Trejos T, Almirall JR. Forensic Characterization and Comparisons of Inks. In: Saferstein R, Hall AB, eds. *Forensic Science Handbook*. CRC Press; 2020:425-463:chap 7.

265. Sworn G. Chapter 27 - Xanthan gum. In: Phillips GO, Williams PA, eds. *Handbook of Hydrocolloids*. Woodhead Publishing; 2021:833-853.

266. Köppen Climate Classification System. National Geographic Society. 2019. Updated Oct. 24, 2019. Accessed 01/02/2022, <u>https://www.nationalgeographic.org/encyclopedia/koppen-climate-classification-</u> <u>system/</u>

267.Wargacki SP, Lewis LA, Dadmun MD. Enhancing the Quality of Aged LatentFingerprints Developed by Superglue Fuming: Loss and Replenishment of Initiator.JournalofForensicSciences.2008;53(5):1138-1144.doi:https://doi.org/10.1111/j.1556-4029.2008.00822.x

268. De Paoli G, Lewis Sr. SA, Schuette EL, Lewis LA, Connatser RM, Farkas T. Photoand Thermal-Degradation Studies of Select Eccrine Fingerprint Constituents. *Journal of Forensic Sciences*. 2010;55(4):962-969. doi:<u>https://doi.org/10.1111/j.1556-4029.2010.01420.x</u>

269. Hansen DB, Joullie MM. The development of novel ninhydrin analogues.*chemical*SocietyReviews.2005;34(5):408-417.doi:https://doi.org/10.1039/b315496n

270. Wiesner S, Springer E, Sasson Y, Almog J. Chemical development of latent fingerprints: 1,2-indanedione has come of age. *Journal of Forensic Sciences*. 2001;46(5):1082-1084.

271. Bouzin JT, Meredino J, Bleay SM, Sauzier G, Lewis SW. New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc. *Forensic Science International: Reports*. 2020;2:100145.

272. Australian Government Bureau of Meterology. Climate statistics for Australian locations. 2022. http://www.bom.gov.au/climate/averages/tables/cw 009225.shtml

273. Pfingstag G. Colorants in inks for writing, drawing and marking. *Journal of the Society of Dyers and Colourists*. 1993;109(5-6):188-192. doi:<u>https://doi.org/10.1111/j.1478-4408.1993.tb01556.x</u>

274. Nedjma S, Djidjelli H, Boukerrou A, et al. Effect of chemical treatment on newspaper fibers reinforced polymer poly(vinyl chloride) composites. *Journal of vinyl* & additive technology. 2016;22(3):173-181. doi:<u>https://doi.org/10.1002/vnl.21425</u>

275. de Ronde A, van Aken M, de Poot CJ, de Puit M. A study into evaluation the location of fingemarks on letters given activity level propositions. *Forensic Science International*. 2020;315:110443.

276. Łojewski T, Miśkowiec P, Molenda M, Lubańska A, Łojewska J. Artificial versus natural ageing of paper. Water role in degradation mechanisms. *Applied Physics A*. 2010;100(3):625-633. doi:<u>https://doi.org/10.1007/s00339-010-5645-9</u>

277. Almirall J, Trejos T. Interpol review of paint, tape, and glass evidence 2019-2022. *Forensic Science International Synergy*. 2023;6:100306. doi:<u>https://doi.org/10.1016/j.fsisyn.2022.100306</u>

278. Zech WD, Malik N, Thali M. Applicability of DNA analysis on adhesive tape in forensic casework. *Journal of Forensic Science*. 2012;57(4):1036-1041. doi:<u>https://doi.org/10.1111/j.1556-4029.2012.02105</u>

279. Sanders N. Recovery of Fingerprint Evidence from Post-Blast Device Materials. *Journal of Forensic Identification*. 2011;61(3):281-295.

280. Feine I, Shpitzen M, Geller B, et al. Acetone facilitated DNA sampling from electrical tapes improves DNA recovery and enables latent fingerprints development. *Forensic Science International.* 2017;276:107-110. doi:<u>https://doi.org/10.1016/j.forsciint.2017.04.023</u>

281. Vanderheyden N, Verhoeven E, Vermeulen S, Bekaert B. Survival of forensic trace evidence on improvised explosive devices: perspectives on individualisation. *Science Reports*. 2020;10(1):12813. doi:<u>https://doi.org/10.1038/s41598-020-69385-1</u>

282. Everaerts AV, Clemens LM. Pressure sensitive adhesives. In: Chandhury M, Pocius AV, eds. Adhesion Science and Engineering : Surfaces, Chemistry and Applications. 2 ed. Elsevier Science & Technology; 2002:chap 11. http://ebookcentral.proguest.com/lib/curtin/detail.action?docID=317137

283. Martin-Martinez JM. Rubber base adhesives. In: Chandhury M, Pocius AV, eds. *Adhesion Science and Engineering : Surfaces, Chemistry and Applications*. Elsevier Science & Technology; 2002.

284. Goodpaster JV, Sturdevant AB, Andrews KL, Brun-Conti L. Identification and comparison of electrical tapes using instrumental and statistical techniques: I. Microscopic surface texture and elemental composition. *Journal of forensic sciences*. 2007;52(3):610-629.

285. National Center for Biotechnology Information. PubChem Compound Summary for CID 590836, Diisononyl phthalate. Accessed March, 2023. https://pubchem.ncbi.nlm.nih.gov/compound/Diisononyl-phthalate

286. Morrison ID, Ross S. Aqueous Solutions of Surface-active Solutes, Colloidal Dispersions: Suspensions, Emulsions, and Foams. 2002:251-253. https://www.scopus.com/inward/record.uri?eid=2-s2.0-85054762534&partnerID=40&md5=9168779fdc083d9c0c011d545cbc9492

287. Bacon SR, Ojeda JJ, Downham R, Sears VG, Jones BJ. The effects of polymer pigmentation on fingermark development techniques. *J Forensic Sci.* 2013;58(6):1486-1494. doi:<u>https://doi.org/10.1111/1556-4029.12206</u>

288. Claveria S. Personal communication. October 2022.

289. Schramm LL, Stasiuk EN, Marangoni DG. 2 Surfactants and their applications. *Annual Reports Section "C" (Physical Chemistry)*. 2003;99(0):3-48. doi:<u>https://doi.org/10.1039/B208499F</u>

290.Kralova I, Sjöblom J. Surfactants Used in Food Industry: A Review. Journal of
DispersionScienceandTechnology.2009;30(9):1363-1383.
doi:
https://doi.org/10.1080/01932690902735561

291. Qiu S, Punzalan ME, Abbaspourrad A, Padilla-Zakour OI. High water content, maltose and sodium dodecyl sulfate were effective in preventing the long-term retrogradation of glutinous rice grains - A comparative study. *Food Hydrocolloids*. 2020;98:105247. doi:<u>https://doi.org/10.1016/j.foodhyd.2019.105247</u>

292. Zhou Z, Zuber S, Cantergiani F, et al. Inactivation of viruses and bacteria on strawberries using a levulinic acid plus sodium dodecyl sulfate based sanitizer, taking sensorial and chemical food safety aspects into account. *International Journal of Food Microbiology*. 2017;257:176-182. doi:https://doi.org/10.1016/j.ijfoodmicro.2017.06.023

293. Sigma-Aldrich. Sodium dodecyl sulfate (CAS-No. 151-21-3) Safety Data Sheet. Revised 18 April 2023. <u>https://www.sigmaaldrich.com/AU/en/sds/sial/l4509</u>

294. ChemSupply. Sodium Lauryl sulfate Safety Data Sheet. Revised November 2020. <u>https://shop.chemsupply.com.au/documents/SL0781CHHR.pdf</u>

295. Kwiecinski W, Segers T, van der Werf S, et al. Evaporation of Dilute Sodium Dodecyl Sulfate Droplets on a Hydrophobic Substrate. *Langmuir*. 2019;35(32):10453-10460. doi:<u>https://doi.org/10.1021/acs.langmuir.9b00824</u>

296. Montgomery ME, Wirth MJ. Spectroscopic Investigation of Sodium Dodecyl Sulfate Adsorption on a Hydrocarbon Monolayer. *Langmuir*. 1994;10(3):861-869. doi:<u>https://doi.org/10.1021/la00015a042</u>

297. Duan M, Wang H, Fang S, Liang Y. Real-time monitoring the adsorption of sodium dodecyl sulfate on a hydrophobic surface using dual polarization interferometry. *J Colloid Interface Sci.* 2014;417:285-292. doi:<u>https://doi.org/10.1016/j.jcis.2013.11.052</u>

298. Rodríguez A, Graciani MdM, Moyá ML. Effects of Addition of Polar Organic Solvents on Micellization. *Langmuir*. 2008;24(22):12785-12792. doi:<u>https://doi.org/10.1021/la802320s</u>

299. Gudelj M, Surina P, Jurko L, Prkic A, Boskovic P. The Additive Influence of Propane-1,2-Diol on SDS Micellar Structure and Properties. *Molecules*. 2021;26(12):3773. doi:<u>https://doi.org/10.3390/molecules26123773</u>

300. Zana R. Aqueous surfactant-alcohol systems: A review. *Advances in Colloid and Interface Science*. 1995;57:1-64. doi:<u>https://doi.org/10.1016/0001-8686(95)00235-I</u>

301. Heydari N, Asgari M, Shojai Kaveh N, Fakhroueian Z. Novel application of PEG/SDS interaction as a wettability modifier of hydrophobic carbonate surfaces. *Petroleum Science*. 2019;16(2):318-327. doi:<u>https://doi.org/10.1007/s12182-018-0260-z</u>

302. Llamas R, Talamantes MC. An analytical study of polypropylene as a support for paint layers. From concept to material in contemporary art. *Journal of Cultural Heritage*. 2014;15(2):136-143. doi:<u>https://doi.org/10.1016/j.culher.2013.04.006</u>

303. Zieba-Palus J. The usefulness of infrared spectroscopy in examinations of adhesive tapes for forensic purposes. *Forensic Sci Criminol*. 2017;2(2):1-9.

304. Nimi C, Chophi R, Singh R. Application of ATR-FTIR spectroscopy and chemometrics to the forensic examination of duct tapes. *Forensic Chemistry*. 2023;32:100461.

305.Smith B. The Infrared Spectra of Polymers II: Polyethylene. Spectroscopy.2021;36(9):24–29.https://www.spectroscopyonline.com/view/the-infrared-spectra-of-polymers-ii-polyethylene

306. Gorassini A, Adami G, Calvini P, Giacomello A. ATR-FTIR characterization of old pressure sensitive adhesive tapes in historic papers. *Journal of Cultural Heritage*. 2016;21:775-785.

307. Mirabile A, Chelazzi D, Ferrari P, et al. Innovative methods for the removal, and occasionally care, of pressure sensitive adhesive tapes from contemporary drawings. *Heritage Science*. 2020;8(1):1-16.

308. Gómez-Sánchez E, Kunz S, Simon S. ATR/FT-IR spectroscopy for the characterisation of magnetic tape materials. *Spectroscopy Europe*. 2012;24(1):6.

309. Keto R. Forensic characterization of black polyvinyl chloride electrical tape. *Crime Lab Digest*. 1984;11(4):71-74.

310. Smith B. The Infrared Spectra of Polymers IV: Rubbers. *Spectroscopy*. 2022;37(1):8–12. <u>https://www.spectroscopyonline.com/view/the-infrared-spectra-of-polymers-iv-rubbers</u>

311. Nishikida K, Coates J. Infrared and Raman analysis of polymers. *PLASTICS ENGINEERING-NEW YORK*-. 2003;68:201-340.

312. John Wiley & Sons, Inc. SpectraBase. Diisononyl phthalate Absorbance Infrared (IR) Spectrum. <u>https://spectrabase.com/spectrum/GZsAgrCyJsG</u>

313.Fernández-Berridi MJ, González N, Mugica A, Bernicot C. Pyrolysis-FTIR and
TGA techniques as tools in the characterization of blends of natural rubber and SBR.
*Thermochimica*Acta.2006;444(1):65-70.
doi:https://doi.org/10.1016/j.tca.2006.02.027

314. Gran Velada. Sodium dodecyl sulphate, Safety Data Sheet. Revised 16 June 2020,

315. ChemSupply. Sodium Lauryl sulfate Safety Data Sheet,. Revised November 2020.

https://shop.chemsupply.com.au/documents/SL0781CHHR.pdfhttps://shop.chems upply.com.au/documents/SL0781CHHR.pdf

316. Lunkenheimer K, Czichocki G. On the Stability of Aqueous Sodium Dodecyl Sulfate Solutions. *Journal of Colloid and Interface Science*. 1993;160(2):509-510. doi:<u>https://doi.org/10.1006/jcis.1993.1429</u>

317. Frade VHJ, Gonçalves MST, Coutinho PJG, Moura JCVP. Synthesis and spectral
properties of long-wavelength fluorescent dyes. Journal of Photochemistry and
Photobiology A: Chemistry. 2007;185(2):220-230.
doi:https://doi.org/10.1016/j.jphotochem.2006.06.013

318. Fowler SD, Greenspan P. Application of Nile red, a fluorescent hydrophobic probe, for the detection of neutral lipid deposits in tissue sections: comparison with oil red O. *Journal of Histochemistry & Cytochemistry*. 1985;33(8):833-836. doi:<u>https://doi.org/10.1177/33.8.4020099</u>

319. Jose J, Burgess K. Benzophenoxazine-based fluorescent dyes for labeling
biomolecules.2006;62(48):11021-11037.doi:https://doi.org/10.1016/j.tet.2006.08.0562006;62(48):11021-11037.

320. Fritz P, Frick A, van Bronswijk W, et al. Investigations into the Influence of Donor Traits on the Performance of Fingermark Development Reagents. Part 1: 1,2-Indanedione-Zinc Chloride. 2017,

321. Fritz P. In-situ Studies of the Chemical Composition of Latent Fingermark Residues. *Curtin University, WA*. 2010,

322. Alsolmy E, Abdelwahab WM, Martinez V, Henary M, Patonay G. Investigation of benzophenoxazine derivatives for the detection of latent fingerprints on porous surfaces. *Journal of Photochemistry and Photobiology A: Chemistry*. 2020;392:112416. doi:<u>https://doi.org/10.1016/j.jphotochem.2020.112416</u>

323. Giridhar G, Manepalli RRKN, Apparao G. Chapter 7 - Confocal Raman Spectroscopy. In: Thomas S, Thomas R, Zachariah AK, Mishra RK, eds. *Spectroscopic*

Methods for Nanomaterials Characterization. Elsevier; 2017:141-161. doi:<u>https://doi.org/10.1016/B978-0-323-46140-5.00007-8</u>

324. 3M. Announcement Letter: Plans to exit per- and polyfluoroalkyl substance (PFAS), David N. Schneider, VP Chemicals and Semiconductor, 3M Company, Electronics Materials Solution Division. December 2022,

325. Doyle S. A review of the current quality standards framework supporting forensic science: Risks and opportunities. *WIREs Forensic Science*. 2020;2(3):e1365. doi:<u>https://doi.org/10.1002/wfs2.1365</u>

326. Spigarelli C. Understanding the hierarchy of controls through a pandemic. *Professional Safety*. 2020;65(5):20-21.

327. Liberati EG, Peerally MF, Dixon-Woods M. Learning from high risk industries may not be straightforward: a qualitative study of the hierarchy of risk controls approach in healthcare. *International Journal for Quality in Health Care*. 2018;30(1):39-43.

328. Roux C, Bucht R, Crispino F, et al. The Sydney declaration - Revisiting the essence of forensic science through its fundamental principles. *Forensic Sci Int*. 2022;332:111182. doi:<u>http://doi.org/10.1016/j.forsciint.2022.111182</u>

Every reasonable effort has been made to acknowledge the owners of copyright material. I would be pleased to hear from any copyright owner who has been omitted or incorrectly acknowledged.

Appendices

Appendix A-1: Statistical analysis test assumptions.

Mann-Whitney U Test

Assumptions

- # Data points should be independent
- # Only two categorical dependent variables (Treatment A and B)
- # Ordinal independent variable (CAST grades)

Hypothesis

- # Null hypothesis: The mean ranks of the groups are the same (Ho)
- # Alternate hypothesis: The mean ranks of the groups are not the same (HA)

Kruskal-Wallis Test

Assumptions

- # Data points should be independent
- # Two or more categorical dependent variables (Treatment A, B, C)
- # Ordinal independent variable (CAST grades)

Hypothesis

- # Null hypothesis: The mean ranks of the groups are the same (Ho)
- # Alternate hypothesis: The mean ranks of the groups are not the same (HA)

Chi-squared Test

Assumptions

Data points should be independent

At least ten values recorded in each cell

Hypothesis

Null hypothesis: The mean ranks of the groups are the same (Ho)

Alternate hypothesis: The mean ranks of the groups are not the same (HA)

Difference of Proportions – Treatment A vs. Treatment B

Assumptions

Data points should be independent

At least ten values recorded in each cell

Hypothesis

Null hypothesis: The proportions are the same (Ho)

Alternate hypothesis: The proportions are not the same (HA)

Appendix A-2: Statistical analysis - Chapter 3 R codes and results.

HeadtoHead_IND/Zn_Statistical test

Grade frequency table

. ,

HeadtoHead_IND/Zn (Kruskal-Wallis Test)

R Code

tA = c(rep(0,10), rep(1,28), rep(2,40), rep(3,50), rep(4,22))

tB = c(rep(0,5), rep(1,32), rep(2,42), rep(3,50), rep(4,21))

tC = c(rep(0,8), rep(1,38), rep(2,34), rep(3,44), rep(4,26))

Nsamp = 450

data_matrix = as.data.frame(matrix(0, Nsamp, 2))

colnames(data_matrix) = c("Treatment", "Grade")

data_matrix[1:150,1] = "tA"

data_matrix[151:300,1] = "tB"

```
data_matrix[301:450,1] = "tC"
```

```
data_matrix[,1] = as.factor(data_matrix[,1])
```

data_matrix[1:150,2] = tA

data_matrix[151:300,2] = tB

```
data_matrix[301:450,2] = tC
```

kruskal.test(Grade~Treatment, data = data_matrix)

Result

Kruskal-Wallis chi-squared = 0.12693, df = 2, p-value = 0.9385

Conclusion

At a 95% level of confidence there not enough evidence to support that a randomly drawn value from one treatment is more likely to be larger or smaller on average than a randomly drawn value from a different treatment.

HeadtoHead_IND/Zn (chi-square)

R Code

tA = c(10,28,40,50,22)

tB = c(5,32,42,50,21)

tC = c(8,38,34,44,26)

Ntreatment = 3

data_matrix = matrix(0, length(tA), Ntreatment)

data_matrix[,1] = tA

```
data_matrix[,2] = tB
```

data_matrix[,3] = tC

chisq.test(data_matrix)

Result

X-squared = 5.2084, df = 8, p-value = 0.7351

Conclusion

At a 95% level of confidence there not enough evidence to support that there is some dependency between the grades and the treatments.

Appendix A-3: Statistical analysis - Chapter 4 R codes and results.

LabTrial_IND/Zn_Statistical test

Grade frequency table

	SPF (tA)	NCFS (tB)
Grade 0	136	130
Grade 1	67	69
Grade 2	93	94
Grade 3	202	192
Grade 4	102	115

LabTrial_IND/Zn (Difference in proportion)

R Code

uA = 304

uB = 307

nA = 600

nB = 600

prop.test(c(uA,uB), c(nA,nB), alternative="two.sided", conf.level = 0.95, correct =
FALSE)

Result

X-squared = 0.03001, df = 1, p-value = 0.8625

95 percent confidence interval:

```
-0.06156907 0.05156907
```

sample estimates:

prop 1 prop 2

0.5066667 0.5116667

Conclusion

At a 95% level of confidence there not enough evidence to support that the proportion of Grade 3&4 under Treatment A is not equal to the proportion of Grade 3&4 under Treatment B.

LabTrial_IND/Zn (Mann-Whitney U)

R Code

tA = c(rep(0,136), rep(1,67), rep(2,93), rep(3,202), rep(4,102))

tB = c(rep(0,130), rep(1,69), rep(2,94), rep(3,192), rep(4,115))

wilcox.test(tA, tB, paired = FALSE, alternative = "two.sided")

Result

W = 176653, p-value = 0.5657

Conclusion

At a 95% level of confidence there not enough evidence to support that a randomly drawn value from one treatment is more likely to be larger or smaller on average than a randomly drawn value from a different treatment.

LabTrial_IND/Zn (Chi-squared Test)

R Code

tA = c(136,67,93,202,102)

tB = c(130,69,94,192,115)

Ntreatment = 2

data_matrix = matrix(0, length(tA), Ntreatment)

data_matrix[,1] = tA

data_matrix[,2] = tB

chisq.test(data_matrix)

Result

X-squared = 1.2027, df = 4, p-value = 0.8777

Conclusion

At a 95% level of confidence there not enough evidence to support that there is some dependency between the grades and the treatments.

IncidentalTrial_IND/Zn_Statistical test

Number of (3&4) fingermarks frequency table

		NCFS
	SPF (tA)	(tB)
Substrate 1	34	26
Substrate 2	39	37

Substrate 3	26	24
Substrate 4	8	14
Substrate 5	46	48

IncidentalTrial_IND/Zn (Chi-squared Test)

R Code

tA = c(136,67,93,202,102)

tB = c(130,69,94,192,115)

Ntreatment = 2

data_matrix = matrix(0, length(tA), Ntreatment)

data_matrix[,1] = tA

data_matrix[,2] = tB

chisq.test(data_matrix)

Result

Kruskal-Wallis chi-squared = 0.12693, df = 2, p-value = 0.9385

Conclusion

At a 95% level of confidence there not enough evidence to support that a randomly drawn value from one treatment is more likely to be larger or smaller on average than a randomly drawn value from a different treatment.

Appendix A-4: Statistical analysis - Chapter 5 R codes and results.

LabTrial_CPS_Statistical test

Grades frequency table_LabTrial_CPS

	Wet SPF	Wetwop
Grade 0	16	29
Grade 1	26	19
Grade 2	10	15
Grade 3	14	21
Grade 4	94	76

Grades frequency table_Sensitivity_CPS

	Wet SPF	Wetwop
Grade 0	1	10
Grade 1	7	21
Grade 2	12	36
Grade 3	32	35
Grade 4	98	48

LabTrial_CPS (Mann-Whitney U Test)

R Code

tA = c(rep(0,16), rep(1,26), rep(2,10), rep(3,14), rep(4,94))

tB = c(rep(0,29), rep(1,19), rep(2,15), rep(3,21), rep(4,76))

wilcox.test(tA, tB, paired = FALSE, alternative = "two.sided")

Result

W = 14285, p-value = 0.05069

Conclusion

At a 95% level of confidence there not enough evidence to support that a randomly drawn value from one treatment is more likely to be larger or smaller on average than a randomly drawn value from a different treatment.

Sensitivity_CPS (Mann-Whitney U Test)

R Code

tA = c(rep(0,1), rep(1,7), rep(2,12), rep(3,32), rep(4,98))

tB = c(rep(0,10), rep(1,21), rep(2,26), rep(3,35), rep(4,48))

wilcox.test(tA, tB, paired = FALSE, alternative = "two.sided")

Result

W = 14428, p-value = 2.742e-09

Conclusion

At a 95% level of confidence there is enough evidence to support that a randomly drawn value from one treatment is more likely to be larger or smaller on average than a randomly drawn value from a different treatment.

LabTrial_CPS (Chi-squared Test)

tA = c(16,26,10,14,94)

tB = c(29,19,15,21,76)

Ntreatment = 2

data_matrix = matrix(0, length(tA), Ntreatment)

data_matrix[,1] = tA

data_matrix[,2] = tB

chisq.test(data_matrix)

Result

X-squared = 9.1503, df = 4, p-value = 0.05745

Conclusion

At a 95% level of confidence there not enough evidence to support that there is some dependency between the grades and the treatments.

Sensitivity_CPS (Chi-squared Test)

tA = c(1,7,12,32,98)

tB = c(10,21,36,35,48)

Ntreatment = 2

data_matrix = matrix(0, length(tA), Ntreatment)

data_matrix[,1] = tA

data_matrix[,2] = tB

chisq.test(data_matrix)

Result

X-squared = 43.621, df = 4, p-value = 7.69e-09

Conclusion

At a 95% level of confidence there is enough evidence to support that there is some dependency between the grades and the treatments.

LabTrial_CPS (Difference in proportion)

R Code

uA = 108

uB = 97

nA = 160

nB = 160

```
prop.test(c(uA,uB), c(nA,nB), alternative="two.sided", conf.level = 0.95, correct =
FALSE)
```

Result

X-squared = 1.6424, df = 1, p-value = 0.2

Conclusion

At a 95% level of confidence there not enough evidence to support that the proportion of Grade 3&4 under Treatment A is not equal to the proportion of Grade 3&4 under Treatment B.

Sensitivity_CPS (Difference in proportion)

R Code

uA = 130 uB = 83 nA = 150 nB = 150 prop.test(c(uA,uB), c(nA,nB), alternative="two.sided", conf.level = 0.95, correct = FALSE)

Result

X-squared = 35.762, df = 1, p-value = 2.23e-09

Conclusion

At a 95% level of confidence, there is enough evidence to support that the proportion of Grade 3&4 under Treatment A is not equal to the proportion of Grade 3&4 under Treatment.



Appendix B-1: Air-condition office datalogger temperature and relative humidity records.

Temperature Relative H

Relative Humidity



Appendix B-2: Non-air-condition Level 5 space datalogger temperature and relative humidity records.

Temperature Relative H

Relative Humidity

Appendix C-1: 3M Announcement letter (PFAS Discontinuation).

3M Electronics Materials Solutions Division

3M Center, Building 223-3S-32 St. Paul, MN, USA 55144-1000



December 20, 2022

Dear Valued Customer,

3M today announced it plans to exit per- and polyfluoroalkyl substance (PFAS) manufacturing and work to discontinue use of PFAS in its product portfolio by the end of 2025. This portfolio decision is based on careful consideration and a thorough evaluation of the external landscape, including multiple factors such as accelerating regulatory trends focused on reducing or eliminating the presence of PFAS in the environment and changing stakeholder expectations.

This impacts products/product families you purchase from 3M listed in Appendix 1. Unfortunately, supply of most of these products continues to be impacted by global supply challenges and strong industry demand exceeding current capacity. Some of the products are also impacted by ongoing force majeure events. We will continue to take all reasonable steps to avoid supply disruptions, but we anticipate that our ability to supply will continue to be constrained.

3M will continue to fulfill contractual obligations during the transition period. Please note that regardless of any order acknowledgement, 3M may not be able to supply your orders due to continuing force majeure events and supply constraints, and we strongly urge customers to find alternative sources of supply.

3M will manage customer last time buy requests based on availability of supply, and we will work with customers to determine specific needs and put in place an action plan to help ensure an orderly transition. 3M does not currently anticipate stockpiling these products.

We understand you may have questions concerning this announcement. Please contact your 3M sales representative, who will work with you to the best of their ability to help support.

Sincerely,

David N. Schneider Vice President, Chemicals and Semiconductor 3M Company, Electronics Materials Solutions Division

Appendix D-1: Human ethics participant information form (English).

Investigation into improved protocols for latent fingermark **Curtin University** detection techniques for jurisdictions with limited resources and tropical climates

HREC Project Number:	HRE2020-0162	
Project Title:	Investigation into improved protocols for latent fingermark detection techniques for jurisdictions with limited resources and tropical climates	
Chief Investigator:	Professor Simon Lewis	
Student researcher:	Jemmy Bouzin	
Version Number:	01	
Version Date:	06/Apr/2020	

PARTICIPANT INFORMATION STATEMENT

1. What is the Project About?

Fingermarks are a valuable tool in criminal investigations. This research project will investigate improved protocols for fingermark detection techniques and facilitate their use in operational laboratories, particularly in countries with limited resources and/or tropical climates. Increased scientific capabilities in fingermark detection techniques have the potential to increase recovery rate of fingermark evidence and improve their effectiveness in criminal investigations.

2. Who is doing the Research?

The project is being conducted by Jemmy Bouzin under the supervision of Professor Simon Lewis and is partially funded by the Government of Seychelles. The results of this research project will be used by Jemmy Bouzin to obtain a Doctor of Philosophy at Curtin University.

3. Why am I being asked to take part and what will I have to do?

This research depends upon collecting a number of sample latent fingermarks to analyse. We are looking for volunteers to allow the research team to collect a sample of your fingermarks for this research. For this research, we are asking you to manually handle and/or intentionally deposit latent fingermarks on a variety of objects. These latent fingermarks will be developed using chemical detection methods and digitally recorded using photography.

4. Are there any benefits' to being in the research project?

There is no direct benefit to you from participating in this research. However, by participating you will be assisting us to increase the recovery of fingermark evidence and improve their effectiveness in criminal investigations to benefit the community in general.

5. Are there any risks, side-effects, discomforts or inconveniences from participating?

Apart from giving up a small portion of your time, we do not expect that there will be any risks or inconveniences associated with taking part in this study. You will handle different objects to deposit fingermarks, however there are no foreseeable physical risks.

Page 1 CRICOS Provider Code 00301J Investigation into improved protocols for latent fingermark **Curtin University** detection techniques for jurisdictions with limited resources and tropical climates

6. Who will have access to my information?

The information collected in this research will be re-identifiable (coded). This means that we will collect data that can identify you, but will then remove identifying information on any data or sample and replace it with a code when we analyse the data. The data will not be used for automated verification or biometric identification. Only the research team have access to the code to match with your name, if it is necessary to do so. Any information we collect will be treated as confidential and used only in this project unless otherwise specified. The following people will have access to the information we collect in this research: the research team and, in the event of an audit or investigation, staff from the Curtin University Office of Research and Development. The information we collect in this study will be kept under secure conditions at Curtin University for 7 years after the research is published, before a decision is made as to whether it should be destroyed. The results of this research may be presented at conferences or published in professional journals. You will not be identified in any results that are published or presented. The research data may be used in future related projects.

7. Will you tell me the results of the research?

If you are interested in obtaining a summary of the results, please contact the researchers after February 2024. Results will not be individual but based on all the information we collect and review as part of the research.

8. Do I have to take part in the research project?

Taking part in this research project is entirely voluntary and you do not have to agree if you do not want to. You also have the right to withdraw from the project at any time without it affecting your relationship with the University, the funding body or any staff member. If you choose to withdraw from the project, any data collected from you will be destroyed. You will be informed in a timely manner if any information becomes available that may affect your consent to continue participation.

9. What happens next and who can I contact about the research?

If you decide to take part in this research, we will ask you to sign a consent form. By signing the consent form, you are telling us that you understand what you have read and what has been discussed. Signing the consent form indicates that you agree to participate in the research project. Please take your time and ask any questions you have before you decide what to do. You will be given a copy of this information and the consent form to keep.

10. Further Information:

Curtin University Human Research Ethics Committee (HREC) has approved this study (HREC number 2020-0162). Should you wish to discuss the study with someone not directly involved; in particular, any matters concerning the conduct of the study or your rights as a participant, or you wish to make a confidential complaint, you may contact the Ethics Officer on (08) 9266 9223 or the Manager, Research Integrity on (08) 9266 7093 or email <u>hrec@curtin.edu.au</u>.

For further information about the study, please contact Professor Simon Lewis on (08) 92662484 or email <u>s.lewis@curtin.edu.au</u> or Jemmy Bouzin at <u>j.bouzin@postgrad.curtin.edu.au</u>

Thank you very much for your involvement in this research.

Your participation is greatly appreciated.

Participant Information Form Version 01, 06/Apr/2020

Page 2 CRICOS Provider Code 00301J

Appendix D-2: Human ethics consent form (English).

💡 Curtin University

Investigation into improved protocols for latent fingermark detection techniques for jurisdictions with limited resources and tropical climates

CONSENT FORM

HREC Project Number:	HRE2020-0162
Project Title:	Investigation into improved protocols for latent fingermark detection techniques for jurisdictions with limited resources and tropical climates
Chief Investigator:	Professor Simon Lewis
Student researcher:	Jemmy Bouzin
Version Number:	01
Version Date:	06/April/2020

1. I have read, the information statement version listed above and I understand its contents.

- 2. I believe I understand the purpose, extent and possible risks of my involvement in this project.
- 3. I voluntarily consent to take part in this research project.
- 4. I understand that I have the right to withdraw my participation in this research project at any time without it affecting my relationship with Curtin University or any staff member.
- I understand that my data collected will not be used for automated verification or biometric identification.
- 6. I have had an opportunity to ask questions and I am satisfied with the answers I have received.
- 7. I give permission for my contribution to this research to be included in peer and non-peer reviewed journals and conferences, provided that I am not identified in any way.
- 8. I understand that my contribution to this research may be used in future related projects.
- 9. I understand that this project has been approved by Curtin University Human Research Ethics Committee and will be carried out in line with the National Statement on Ethical Conduct in Human Research (2007).
- 10. I understand I will receive a copy of this Information Statement and Consent Form.

Participant Name	
Participant Signature	
Date	

Participant Consent Form Version 01, 06/Apr/2020

	Page	1
RICOS Provid	er Code	00301J

C

Appendix D-3: Human ethics participant information form (Creole).

Lenvestigation dan amelyorasyon protokol teknik pou deteksyon lanprent-ledrwa pou bann zirediksyon avek resours limite e klima tropik

DEKLARATION LENFORMASYON POU PARTISIPAN

HREC Nimero Proze:	HRE2020-0162
Tit Proze:	Lenvestigation dan amelyorasyon protocol teknik pou deteksyon lanprent-ledrwa pou ban zirediksyon avek resous limite e klima topik
Sef Resers:	Profeser Simon Lewis
Etidyan ki per fer resers:	Jemmy Bouzin
Nimero Verzion	01
Dat Verzion	06/Apr/2020

1. Lo kwa sa resers i baze?

Lanprent-ledrwa i en zouti enportan dan bann lenvestigation kriminel. Sa proze pou fer resers lo lenvestigation dan amelyorasyon bann protocol teknik pou deteksyon lanprent-ledrwa e fasilit zot litilizasyon dan laboratwar, sirtou pou bann pei avek resours limite ek klima tropik. Amelyorasyon dan teknik deteksyon lanprent-ledrwa i ogmant posibilite pou dekouver plis levidans lanprent-ledrwa e ogmant son lefikasite dan bann ka kriminel.

2. Lekel ki pe fer sa Resers?

Sa proze resers pe ganny fer par Jemmy Bouzin anba sipervizyon Professer Simon Lewis e ganny parsyelman finanse par Gouvernman Sesel. Rezilta sa proze resers pou ganny servi par Jemmy Bouzin pou obtenir en degree Doktora kot Liniversite Curtin.

3. Akoz mon pe ganny demande pou pran par e ki mon pou bezwen fer?

Sa resers i depann lo koleksyon en serten kantite lanprent-ledrwa pou analize. Nou pe rod volonter pou donn zot lanprent-ledrwa pou sa resers. Nou pou demann ou pou tous bann lobze pou depoz ou lanprent-ledrwa entonsyonelman ou non-entonsyonelman. Sa bann lanprent-ledrwa pou ganny devlope avek bann metod kemikal e apre ganny rikorde par fotografi.

4. Eski i annan okenn benefis pou pran par dan sa resers?

Napa oken benefis direk dan pran par dan sa resers. Me si ou partisipe, ou pou ede amelyor teknik deteksyon e lefikasite levidans lanpret-ledrwa dan bann ka kriminel, e sa i anmenn gran benefis pou lasosyte anzeneral.

5. Eski I annan oken risk ou lenkonvenyans dan partisip dan sa resers?

Otreman ki donn ou letan, nou pa vwar okenn risk ou lenkonvensyan dan pran par dan sa resers. Ou pou bezwen tous bann lobze pou depoz ou lanpret-ledrwa, me nou pa antisip okenn risk fizik.

6. Lekel ki pou annan akse avek mon lenformasyon?

Bann lenformasyon ki ganny kolekte dan sa resers pou ganny re-idantifye. Sa I vedir ki nou pou anmas lenformasyon ki kapab idantifye ou, me nou pou ranplas sa lenformasyon avek en kod ler

Participant Information Form Version 01, 06/Apr/2020

Page 1 CRICOS Provider Code 00301J Lenvestigation dan amelyorasyon protokol teknik pou deteksyon lanprent-ledrwa pou bann zirediksyon avek resours limite e klima tropik



nou pe analize. Ou lenformasyon pa pou ganny servi pou verifikasyon otomatik oubyen pou idantifikasyon. Zis bann endividi ki dan sa group resers ki pou konnen e annan akse avek ou lenformasyon ki korespond avek sa kod, e sa pou ganny akse zis dan ka kot i neseser. Tou lenformasyon ki nou kolekte pou konfidansyel e ganny servi zis dan sa proze, sof, si i ganny endike otreman. Sa bann dimoun swivan pou annan akse avek bann lenformasyon ki nou kolekte dan sa resers: tim resers e personel sorti kot lofis Resers e Delopman kot Liniversite Curtin dan ka en odit ou lenvestigation. Lenformasyon ki nou kolekte dan sa proze pou ganny garde an sekirite kot Liniversite Curtin pou 7 an apre ki sa resers i ganny piblirye, e en desizyon pou ganny pran apre sa si sa bann lenformasyon pou ganny detri. Rezilta sa resers i kapab ganny prezante kot bann konferans ou piblirye dan ban zournal profesyonel. Ou pa pou ganny idantifye dan okenn sa bann prezantasyon ou konferans. Sa bann lenformasyon resers i kapab ganny servi dan bann lezot proze relevan dan le fitir.

7. Eski mon pou ganny enformen rezilta sa resers?

Si ou enterese pou konn rezilta sa resers, silvouple kontakte sa tim resers apre Fevriye 2024. Rezilta pa pou baze lo en endividi me lo tou lenformasyon kolektiv kin ganny anmase dan sa resers.

8. Eski mon bezwen pran par dan sa resers?

Pran par dan sa resers i volanter e ou pa bezwen agree pou partisipe si ou pa anvi. Ou annan drwa pou ou retir ou participation dan sa proze a okenn moman ki ou anvi san afekte ou relasyon avek Gouvernman Sesel, Liniversite Curtin ou okenn personnel sa bann lenstitisyon. Si ou retir ou partisipasyon, tou ou lenformasyon ki nou kolekte pou ganny detri. Ou pou ganny enformen o pli vit posib si okenn nouvo lenformasyon i afekte ou konsantman pou kontiyen partisip dan sa resers.

9. Ki arrive apre e lekel ki mon kontakte pou lenformasyon lo sa resers?

Si ou deside partisip dan sa resers nou pou demann ou pou sir en form konsantman. Par sir sa form konsantman i vedir ou konpran sa ki oun lir e as kin ganny diskite e ou endike ki ou agree pou partisip dan sa resers. Silvouple pran letan pou demann kestyon ki ou annan avan deside ki ou pou fer. Ou pou ganny donnen en kopi sa bann lenformasyon e form konsantman.

10. Lezot Lenformasyon:

Sa resers in ganny aprouve par Komite Etik pou Resers Imen kot Liniversite Curtin (HREC number 2020-0162). Si ou anvi diskit sa proze avek en endividi ki pa direkteman enplike dan sa resers; partikilverman okenn konsern ki annan pou fer avek la fason ki sa resers pe ganny fer , oubven ou drwa koman en partisipan, oubyen si ou anvi fer en konplent konfidansyel, ou kapab kontakte en Zofisye Etik lo (+61 8 9266 9223) oubyen Manager, Entegrite Resers lo (+61 8 9266 7093) ou email hrec@curtin.edu.au. Pou plis lenformasyon lo sa resers, silvouple kontakte Professer Simon Lewis lo (+61 8 9266 2484) ou email s.lewis@curtin.edu.au oubyen Jemmy Bouzin j.bouzin@postgrad.curtin.edu.au

Mersi bokou pou ou lentere dan sa resers.

Ou partisipasyon i ganny vreman apresye.

Participant Information Form Version 01, 06/Apr/2020

Page 2 CRICOS Pro

Appendix D-4: Human ethics consent form (Creole).

Curtin University

Lenvestigation dan amelyorasyon protokol teknik pou deteksyon lanprent-ledrwa pou bann zirediksyon avek resours limite e klima tropik

FORM KONSANTMAN

HREC Nimero Proze:	HRE2020-0162
Tit Proze:	Lenvestigation dan amelyorasyon protokol teknik pou deteksyon lanprent-ledrwa pou bann zirediksyon avek resours limite e klima tropik
Sef Resers:	Profeser Simon Lewis
Etidyan ki pe fer Resers:	Jemmy Bouzin
Nimero Verzion:	01
Dat Verzion:	06/Apr/2020

1. Monn lir verzion deklaration lenformasyon parey i ganny liste anler e mon konpran son konteni.

- 2. Mon kwar ki mon konpran rezondet e posiblite bann risk dan pran par dan sa proze.
- 3. Mon volanterman donn mon konsantman pou pran par dan sa resers.
- Mon konpran ki a okenn moman, mon annan drwa pou retir mon partisipasyon dan sa resers san afekte mon relationsyon avek Liniversite Curtin ou okenn manm.
- Mon konpran ki mon lenformasyon ki ganny kolekte pa pou ganny servi pou fer verifikasyon otomatik ou identifikasyon.
- 6. Monn ganny loportinite pou demann kestyon, e mon satisfe avek larepons ki monn ganyen.
- 7. Mon donn permisyon pou mon kontribisyon dan sa resers ganny enkli dan bann piblikasyon avek kondisyon ki mon pa ganny identifye dan okenn fason.
- 8. Mon konpran ki mon kontribisyon dan sa resers i kapab ganny servi dan bann lezot proze relevan dan le fitir.
- 9. Mon konpran ki sa resers inn ganny aprouve par komite Resers lo Etik Imen kot Liniversite Curtin, e pou ganny fer an lir avek Declaration National lo "Conduite" Etik pou Resers Imen (2007).
- 10. Mon konpran ki mon pou resevwar en kopi Deklaration Lenformasyon e Form Konsantman.

Participant Consent Form Version 01, 06/Apr/2020

Page 1 CRICOS Provider Code 00301J

Appendix E-1: Contribution statement - New light on old fingermarks paper.

To whom it may concern,

I, Jemmy T. Bouzin, contributed conceptualisation, investigation, methodology, visualization, writing-original draft, writing-review & editing to the following paper:

Jemmy T. Bouzin, Jason Merendino, Stephen M. Bleay, Georgina Sauzier, and Simon W. Lewis. New light on old fingermarks: The detection of historic latent fingermarks on old paper documents using 1,2-indanedione/zinc. *Forensic Science International: Reports 2020, 2, 100145*.

I as a Co-Author, endorse that this level of contribution by the candidate indicated above is appropriate.

Jason Merendino

Stephen M. Bleay

Georgina Sauzier

Simon W. Lewis
Appendix E-2: Contribution statement – Forensic science in Seychelles paper.

To whom it may concern,

I, Jemmy T. Bouzin, contributed conceptualisation, investigation, visualization, writing-original draft, writing-review & editing to the following paper:

Jemmy T. Bouzin, Georgina Sauzier, and Simon W. Lewis. Forensic science in Seychelles: An example of a micro-jurisdiction forensic delivery system. *Forensic Science International: Synergy 2021, 3, 100139.*

I as a Co-Author, endorse that this level of contribution by the candidate indicated above is appropriate.

Georgina Sauzier

Simon W. Lewis

Appendix E-3: Contribution statement – IND/Zn Comparative assessment paper.

To whom it may concern,

I, Jemmy T. Bouzin, contributed conceptualisation, investigation, methodology visualization, writing-original draft, writing-review & editing to the following paper:

Jemmy T. Bouzin, Aaron J. Horrocks, Georgina Sauzier, Stephen M. Bleay, and Simon W. Lewis. Comparison of three active 1,2-indanedione-zinc formulations for fingermark detection in the context of limited resources and supply chain risks in Seychelles. *Forensic Chemistry 2022, 30, 100439*.

I as a Co-Author, endorse that this level of contribution by the candidate indicated above is appropriate.

Aaron J. Horrocks

Georgina Sauzier

Stephen M. Bleay

Simon W. Lewis

Appendix E-4: Contribution statement – Solstice® PF paper.

To whom it may concern,

I, Jemmy T. Bouzin, contributed conceptualisation, investigation, methodology visualization, writing-original draft, writing-review & editing to the following paper:

Jemmy T. Bouzin, Amanda A. Frick, Georgina Sauzier, and Simon W. Lewis. Preliminary evaluation of Solstice[®] PF as a replacement carrier solvent for Australian fingermark detection. *Forensic Science International 2022, 340, 111465*.

I as a Co-Author, endorse that this level of contribution by the candidate indicated above is appropriate.

Amanda A. Frick

Georgina Sauzier

Simon W. Lewis

Appendix E-5: Contribution statement – Mind the gap paper.

To whom it may concern,

I, Jemmy T. Bouzin, contributed conceptualisation, investigation, visualization, writing-original draft, writing-review & editing to the following paper:

Jemmy T. Bouzin, Thais Lopes, Anna L. Heavey, Jessie. Parrish, Georgina Sauzier, and Simon W. Lewis. Mind the gap: The challenges of sustainable forensic science service provision. *Forensic Science International Synergy 2023, 6, 100318*.

I as a Co-Author, endorse that this level of contribution by the candidate indicated above is appropriate.

Thais Lópes

Anna Heavey

Jessie Parrish

Georgina Sauzier

Simon Lewis